Highly Scaled Silicon Field Emitter Arrays with Integrated Silicon Nanowire Current Limiters

by

Stephen Angelo Guerrera

B.S., Electrical and Computer Engineering, Northeastern University (2008),
S.M., Electrical Engineering and Computer Science, Massachusetts Institute of Technology (2011)



Submitted to the Department of Electrical Engineering and Computer Science in partial fulfillment of the requirements for the degree of

Doctor of Philosophy in Electrical Engineering

at the

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

February 2016

© Massachusetts Institute of Technology 2016. All rights reserved.

Signature redacted

Signature redacted October 16, 2015

Akintunde I. (Tayo) Akinwande

Professor

Thesis Supervisor

Signature redacted

Accepted by . .

Certified by .

UC Leslie A. Kolodziejski

Chairman, Department Committee on Graduate Theses

То ту Рорру.

•

[Quantum mechanics] describes nature as absurd from the point of view of common sense. And yet it fully agrees with experiment. So I hope you can accept nature as She is – absurd.

.

- Richard P. Feynman

.

Highly Scaled Silicon Field Emitter Arrays with Integrated Silicon Nanowire Current Limiters

by

Stephen Angelo Guerrera

Submitted to the Department of Electrical Engineering and Computer Science on October 16, 2015, in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Electrical Engineering

Abstract

Field emitter arrays (FEAs) are a promising class of cold electron sources with applications in RF amplifiers, terahertz sources, lithography, imaging, and displays. FEAs are yet to achieve widely implemented because of serious challenges which have limited their viability in systems that require advanced electron sources. We identified four major challenges that posed significant barriers to the application of field emitter arrays in systems. These challenges are (1) charge injection and breakdown of the insulator between the emitter and the extraction gate, (2) thermal runaway due to Joule heating or micro-plasma discharge, (3) back-ion bombardment resulting in emitter tip damage (4) large capacitance between the gate and the substate that limits switching performance.

In this thesis, we address these challenges with a new device architecture that consists of a sharp silicon emitter atop a silicon nanowire embedded in a dielectric matrix of SiO_2 and SiN_x . The 10- μ m tall, 200-nm diameter silicon nanowire limits current and improves reliability through velocity saturation and the pinch-off of majority carriers. The 2- μ m thick SiO_2 insulator between the gate and the substrate and the conformal dielectric matrix that embeds the nanowire current limiters prevents charge injection and minimizes the capacitance between the gate and the substrate. Since the nanowire current limiter is fabricated directly underneath each field emitter, we maintain an emitter density of 10^8 emitters/cm², enabling high current density. The design of the anode prevents tip erosion from back-streaming ions.

These arrays demonstrate consistent current scaling of array sizes from a single emitter to 25,000 emitters, low voltage ($V_{GE} < 60V$), high current density ($J > 100 \text{ A/cm}^2$), and long lifetime (t > 100 hours at 100 A/cm², > 100 hours at 10 A/cm², and > 300 hours at 100 mA/cm²). The current density enabled by our device structure is an improvement of > 10× over state-of-the art ($\approx 1 - 10 \text{ A/cm}^2$) for Si field emission cathodes operated in a direct current mode. Our devices demonstrated a turn-on voltage as low as 8.5 V. This low-voltage enabled operation in a 500 Torr He ambient with an anode-emitter voltage below the first ionization potential of He (~ 19 V). These high current, high current density, long lifetime cold cathodes could enable new approaches to x-ray imagers, RF amplifiers, THz sources, and deep UV sources.

Thesis Supervisor: Akintunde I. (Tayo) Akinwande Title: Professor

Acknowledgments

So many people have supported and helped me over the years that it is likely that I'm going to omit someone. Please don't take it personally.

First, I want to sincerely thank my thesis advisor, Tayo Akinwande, for his never-ending support and enthusiasm for my project. I learned so much from him, and continue to be awed by the breadth and depth of his technical knowledge, his compassion, and his ability to know every single person on the face of this Earth. I have learned so much working with him over the years. I'm honored to consider him a friend and so very thankful that he finally decided to grace this piece of work with his "John Hancock" and hope for many more trips to the "Library" in the future.

I also wish to thank the rest of my committee, Provost Marty Schmidt and Prof. Karl Berggren for many interesting conversations, ideas, and directions to take my research. The guidance has been extremely valuable, and Prof. Berggren's nanofabrication class was a huge boon in understanding my fabrication process to the depth needed to attain my results.

I also wish to acknowledge Dr. Luis Velasquez-Garcia for many interesting conversations and providing a solid bedrock on which this work is heavily based. His fabrication insight and his daring to make 100:1 aspect ratio silicon pillars allowed me to succeed in this project.

I wish to thank our collaborator, Matt Cole, at the University of Cambridge for useful discussion and transferring graphene to our samples.

I have been extremely fortunate to worth with a wonderful group of people in the A-team over the years. The Akinwande group is full of amazing and wonderful people that made research so much fun. I particularly wish to thank Melissa, Annie, Michael, Arash, Alex, Andy, Neptune.

The rest of the folks on the 6th floor and MIT at large made the cleanroom and late nights in the office a surprising amount of fun, and for the uncountable useful discussions and times you helped me. I especially wish to thank Jamie, Winston, Nicole, Alon, Omair, Eva, Laura, Radhika, Allen, Daniel, Cheri, and Liz.

The MTL staff have done an amazing job of keeping the lights on and the tools running over the years and have been an huge source of knowledge, insight, and fun. I particularly wish to thank Vicky, Bernard, Bob, Eric, Paul, and Dennis.

My friends outside of MIT and particularly Forest Hills Runners have been largely responsible for keeping me sane, and I wish to thank you for that. I want to shout out to Dave, Manis, Glen, J-Rod, Kyle, Zach, Heather in particular.

Thanks to John Kymissis for all of the insight and support, and for probably getting me into MIT in the first place.

None of this work would have been possible without the love of my family.

To Lauren, thank you for your immense and continuing love, support, compassion, and understanding. I'm extremely lucky to have you as a partner, and I'm excited to take this next adventure with you.

This work was supported by the Defense Advanced Research Projects Agency/SPAWAR under Grant N66001-12-1-4212 and Grant N66001-15-1-4022.

Also, because there is precedent for this, here is a picture of my kitten, Franklin:



Contents

1	Intro	oduction	29
	1.1	Problem Statement	29
	1.2	Objectives of This Work	36
	1.3	Structure of Thesis	36
2	Field	l Emission Theory and Technology	39
	2.1	Objectives	39
	2.2	Fowler-Nordheim Model	39
	2.3	The Field Factor, eta	43
	2.4	State of the Art Field Emitter Arrays	46
		2.4.1 Spindt approach	48
		2.4.2 Self-aligned gate apertures through etching	50
		2.4.3 Controlling the Electron Supply	53
	2.5	Technical Approach	55
3	The	Vertical Silicon Nanowire Current Limiter	61
	3.1	Introduction	61
	3.2	Modeling the Si Nanowire Current Limiter — An Intuitive Model	64
	3.3	Numerical Modeling	67
	3.4	I-V Characteristics	70
	3.5	Array Effects	77
	3.6	Thermal Modeling and Analysis	84
	3.7	Summary	86

4	Ung	ated Silio	con Field Emitter Arrays with Nanowire Current Limiters	87
	4.1	Introdu	action	87
	4.2	Design		87
	4.3	Fabrica	tion	90
	4.4	Curren	t-Voltage Characterization	94
	4.5	Discuss	sion	95
	4.6	Summa	ary	98
5	Gate	ed Silicor	n Field Emitter Arrays	99
	5.1	Introdu	action	99
	5.2	Numer	ical Modeling	99
	5.3	Process	Development and Proof-of-Concept Devices	102
		5.3.1	Process Development	102
		5.3.2	Dielectric Fill Process	109
		5.3.3	I-V Characterization	115
	5.4	High P	Performance Silicon Field Emitter Arrays	118
		5.4.1	Design	118
		5.4.2	Fabrication	120
		5.4.3	I-V Characterization	129
		5.4.4	New technique for electrical estimation of emitter tip statistics	141
		5.4.5	Emission Angle Measurements	147
		5.4.6	Failure Analysis	152
	5.5	Summa	ary	154
6	Life	time Cha	aracterization of Silicon Field Emitter Arrays	155
	6.1	Introdu	uction	155
	6.2	The S-1	K Chart	156
	6.3	0.1 A/c	m^2 Device	156
	6.4	10 A/cı	m ² Device	158
	6.5	100 A/	cm ² Devices	163
	6.6	Discus	sion	167

	6.7	Summary	167
7	Pote	ntial New Applications of FEAs	169
	7.1	Operation in Helium Ambient	169
	7.2	Electron-transparent Graphene-Gated FEAs	. 177
	7.3	Summary	182
8	Sum	mary and Future Directions	185
	8.1	Summary	185
	8.2	Significant Contributions of This Work	186
	8.3	Future Directions	188
A	Tabl	e of Fundamental Physical Constants	191
B	Tabl	e of Important Material Parameters of Si and GaAs at 300 K	193
С	Tabl	e of Important Material Parameters of SiO $_2$ and Si $_3\mathrm{N}_4$ at 300 K	195
D	Tuni	neling Through a Surface Barrier	197
E	Don	or Ionization Efficiency in Silicon Nanowires	203
F	Mod	eling of Surface States in Si Nanowire Current Limiters	207
G	Solu	tion of Poisson's equation in cylindrical coordinates for a partially depleted S	ìi
	nano	owire	219
н	Proc	ess Flow for ungated field emitter arrays with nanowire current limiters	221
I	Proc	ess Flow for gated field emitter arrays with nanowire current limiters	223

List of Figures

1-1	Applications for Gated Electron Field Emitters	30
1-2	0.85 THz Vaccum amplifer, reprinted with permission from [8]. ©2012 IEEE	32
2-1	Two methods of emission of electrons from a metal. (a) Thermionic or photo	
	emission. (b) Field emission	40
2-2	Comparison of the barrier with image charge correction (solid line) to the	
	uncorrected barrier (dashed line)	41
2-3	Comparison of simulations of the Ball-in-sphere model and a conical emitter	
	tip with tip radii, $r = 10$ nm and gate aperture radius $d = 175$ nm. The top	
	figures depict equipotential contours of a 2-D cross-section of the 3-D models	
	used. The bottom figures are 2-D cross-sections at the apex of the emitter with	
	equipotential lines drawn. The ball-in-sphere model tends to slightly overesti-	
	mate the field at the apex of the emitter	45
2-4	Tip radius exhibits a log-normal distribution [103]	46
2-5	The effect of tip radius distribution on emission current. The large, central plot	
	show emission currents for different emitters of a typical field emitter device,	
	and the effect of that the modeled statistical variation in tip radius has on	
	emission current. For this typical emitter tip size distribution, it is possible for	
	the sharpest emitters to burn out before the dullest emitters even turn on, thus	
	limiting the overall performance and utilization of the array.	47
2-6	Schematic diagram of key processing steps in the Spindt approach for the fab-	
	rication of field emitter tips with self-aligned gates uses the deposition and	
	lift-off of refractory metals, traditionally Molybdenum	49

2-7	Schematic diagram of key processing steps in two different materials for us-	
	ing chemical-mechanical polishing (CMP) to form emitters with self-aligned	
	gates, (Left) Silicon and (Right) Carbon Nanotubes. In both cases, a con-	
	formal oxide deposition forms a "bump" that later becomes the gate aperture	
	when excess material is removed during the polishing step	52
2-8	Block diagram of the factors involved with field emission. A field emitter may	
	be limited by either the transmission at the surface, or the flux to the surface	53
2-9	Qualitative load lines of different types of current sources connected in series	
	with each individual field emitter, demonstrating the variation of emission cur-	
	rent for different tip radii. The dashed blue line shows that even if the output	
	resistance of the FET equaled the that of the resistor, the FET would provide	
	more emission current	54
2-10	The effect of tip burnout on current performance. By limiting the current to	
	each individual emitter in an array, the overall performance of the array can	
	be improved. Note the hysteresis of the simulation of the array with burnout.	
	During the initial sweep, as the voltage is increased, the anode current decreases	
	due to the burnout of more and more emitter tips. On the down sweep and on	
	subsequent sweeps, the burnt out emitters do not participate in field emission.	56
2-11	Schematic cross-section diagram of the ungated FET current limiter by Baek.	
	Reprinted with permission from [7]. ©1985 IEEE	57
2-12	Measured and simulated I-V characteristics from Baek et al. Reprinted with	
	permission from [7]. ©1985 IEEE	57
2-13	(a) Array of field emitters with 10 μ m pitch, 1 μ m diameter, and 100 μ m tall Si	
	pillar current limiters. (b) Characterization setup (c)Fowler-Nordheim plot of	
	a field emitter array with individual silicon pillar current limiters that demon-	
	strates current control by the current limiter. Due to the dependencies of the	
	FN equation, if $\ln(\frac{I}{V^2})$ is plotted against V^{-1} , the graph should be a straight	
	line if the emission mechanism is field emission. This device shows a clear de-	
	viation from linear behavior at large voltages. Reprinted with permission from	
	[125]. ©2011 IEEE	59

3-1	(a) Schematic diagram of a single field emitter in series with a current lim-	
	iter. (b) Circuit diagram of the current limiter-FEA structure. (c) SEM of	
	the completed current limiter-FEA structure. (d) Cross-sectional SEM of a Si	
	nanowire current limiter without a field emitter with oxide removed to show	
	the pillar. This chapter focuses on modeling and characterization of the Si	
	Nanowire current limiter	63
3-2	Schematic transfer characteristic for the vertical current limiter	65
3-3	Schematic device cross-section showing the evolution of the equipotential lines	
	and depletion width	68
3-4	Representative simulation of a silicon nanowire current limiter with 200 nm	
	diameter, length of 8 μ m, and $N_D = 5 \times 10^{15} \text{ cm}^{-3}$.	69
3-5	(a) Effect channel doping has on linear and output conductances with a chan-	
	nel length of 10 μ m. (b) g_{lin}/g_o vs. doping. (c) Effect channel doping has on	
	linear and output conductances with a channel doping of 5×10^{14} cm ⁻³ . (d)	
	g_{lin}/g_o vs. channel length.	71
3-6	(top) Effect channel doping has on saturation current with a channel length	
	of 10 μ m. (bottom) Effect channel length has on saturation current with a	
	channel doping of $5 \times 10^{14} \text{ cm}^{-3}$	72
3-7	I-V Characteristics of single current limiters (left) and an array of 4×10^6	
	current limiters (right).	73
3-8	(A) Current-voltage characteristics of a single SiNW at a variety of different	
	temperatures. (B) Arrhenius plot of linear conductance, g_{lin} shows a nearly	
	linear relationship, with an activation energy of ~ $E_g/2$ (C) Arrhenius plot of	
	saturation current, I_{sat} shows linear relationship, with an activation energy of	
	$\sim E_g/2.$	75
3-9	Comparison of the n_i extracted from the saturation current to a model for	
	intrinsic carrier concentration, assuming a nanowire radius of 50 nm	76
3-10	Estimation of surface recombination velocity from the I-V characteristics	77

3-11	Schematic drawing of a silicon nanowire surrounded by oxide and a neutral	
	substrate. the substrate/oxide/nanowire structure forms a parasitic, unwanted	
	MOS structure	78
3-12	The effect of doping (left) and oxide thickness (right) on the voltage at which	
	the channel pinches off due to the parasitic MOS structure formed by the	
	substrate surrounding a single nanowire	80
3-13	Cross-section of a simulation of a 5x5 array at $V_{DS} = 10$ V. This cross-section	
	shows the electron concentration in the nanowires, and the depletion edge is	
	highlighted in white, defined as electron concentrations less than 10% of the	
	value in thermal equilibrium. This simulation demonstrates the effect of the	
	surrounding substrate on the I-V characteristics of the nanowires	81
3-14	Comparison of I-V Characteristics across a 10x10 Array, demonstrating the	
	effect of the substrate on the current limiter. The current limiters along the	
	periphery saturate at lower currents than those in the center	82
3-15	Comparison of output conductance (a), linear conductance (b), and saturation	
	current (c) for each position in the 10x10 array.	83
3-16	Schematic drawing of a silicon nanowire surrounded by oxide and a neutral	
	substrate. the substrate/oxide/nanowire structure forms a parasitic, unwanted	
	MOS structure.	85
3-17	Simulation of heating and avalanche breakdown in a silicon nanowire using	
	the hydrodynamic models	85
4-1	(A) Simulation structure for the exploration of the screening effects of adja-	
	cent emitters. The simulation shows the electric field as shaded colors, and	
	the equipotential lines. Units of all of the aves in this figure are in microns	
	(B) Detail of the emitter tip meshing used in this simulation (C) Detail of	
	the normalized notential (shaded) surrounding 2 emitter tips in a simulation	
	structure. The anode-emitter spacing is $x = 1$ //m and the emitter to emitter	
	structure. The anouc-children spacing is $x = 1 \ \mu \text{m}$ and the children-to-children	<u>8</u> 0
	$p_{1}(\mu) = p_{1}(\mu) + p_{2}(\mu) + p_{3}(\mu) $	07

90 91 92 92
90 91 92 92
90 91 92 92
91 92 92
91 92 92
92 92
92 92
92 92
92
92
93
94
96
98
101
102
103

5-5	Some of the challenges of DRIE 106
5-6	Exploration space for the design of experiments
5-7	(A) the response of vertical etch rate to SF6 flow rate and Passivation cycle time
	(B) response of lateral etch rate to the same (C) Selectivity demonstrating the
	shift from the original etch [denoted by black square] to the optimized etch
	[denoted by white circle] (D) Confirmation of etch selectivity 108
5-8	Process for filling in the voids between adjacent silicon nanowires with a di-
	electric matrix
5-9	Void-free dielectric matrix with vertical silicon nanowires embedded in them
	after CMP planarization
5-10	Schematic of a CMP process
5-11	Wafer thickness maps for three different times to extract the etch rate and
	uniformity of the polishing. 49 points across the wafer were measured using
	spectroreflectometry 113
5-12	Extraction of nitride polishing rate from thickness measurements. The error
	bars show the standard deviation of the measurments made on a single wafer,
	showing the spatial variation in film thickness
5-13	Cross-section SEM of a device after nitride polishing. There is about 200 nm
	of nitride remaining on top of the oxide bumps
5-14	SEM image of the oxide domes after nitride etchback in 160 C phosphoric acid. 116
5-15	Testing setup of the devices
5-16	I-V characteristics and Fowler-Nordheim plot of the proof of concept gated
	field emitter array. The device turns on at 20 V demonstrates current saturation
	at approximately 35 V
5-17	A 3-D rendering of the proposed device structure
5-18	The full process for FEAs with integrated SiNW Current Limiters and self-
	aligned gates
5-19	Etching the mesa that will form the active device region
5-20	
	filling the trenches surrounding the active device region and planarization 122

5-22	Silicon nanowire etch
5-23	Void-free dielectric matrix with vertical silicon nanowires embedded in them 124
5-24	Oxide domes that form the gate aperture
5-25	Cartoon and SEM of the gate polysilicon deposition
5-26	Characterization of the polysilicon polishing rate
5-27	The gate apertures after CMP 126
5-28	SEM cross-section of the completed Device structure
5-29	SEM of the completed Device structure
5-30	SEM of the completed Device structure with oxide removed to show detail of
	the Si nanowire. Inset: Detail of an emitter tip 129
5-31	(A) I-V Characteristics (current per tip on a semilog scale), (B) IV Charac-
	teristics (current per tip on a linear scale), (C) F-N Plot per tip of a variety of
	different array sizes, demonstrating that current scales with the number of tips.
	In each of these devices except the 50 \times 50 arrays at V_{GE} > 70 V, the total gate
	interception was less than 5 % of the emission current (anode efficiency > 95%.) 130
5-32	The effect of changing the anode voltage on a 500×500 array that exhibits
	current saturation
5-33	Output Characteristics for a 500x500 Array
5-34	Transfer characteristics for the same 500x500 array shown in Figure 5-33 133
5-35	Band diagram from the emitter tip to the anode at $V_{AE} = 0$ V
5-36	Band diagram from the emitter tip to the anode at $V_{AE} = \phi_A \dots \dots \dots 135$
5-37	Collection probability of the anode as a function of anode-emitter voltage for
	$\phi_A = 4.5 \text{ eV.}$
5-38	Band diagram from the emitter tip to the anode at V_{AE} < V_{GE} and V_{AE} > V_{GE} V136
5-39	Gate Transconductance of the transfer characteristics shown in Figure 5-33 137
5-40	The Ranadive gate current model 138
5-41	Output and transfer characteristics for the single device
5-42	Comparison of the transfer characteristics of 5 different devices scaled by the
	number of emitters

5-43	Fowler-Nordheim plot of the several different field emission arrays shown in
	the previous graph
5-44	Comparison of simulated I-V characteristics (left) and FN chart (right) to ex-
	perimental data for a single emitter (black dots)
5-45	Finite element electrostatics simulation to determine the relationship between
	tip radius (r) and field factor (eta) for the silicon nanowire with self-aligned
	gated tip. The nanowire is 200 nm in diameter and 10 microns tall, with a tip
	half-cone angle of 30 degrees. (A)-(C) Detail of the electric field around the
	apex of the tip for tip radii of 10 nm (A), 5 nm (B), and 1 nm (C)
5-46	Fit of field factors with the model $\beta = k/r^n$ from the finite element electro-
	statics model shows good experimental agreement for this structure for $r = 0.5$
5-47	Box chart of the distribution of average effective tip radii extracted from I-V
	characteristics and fitted to tip radius using the extrapolation from the finite
	element model. The square in the center of each box indicates the mean value
	of tip radius of the distribution. The horizontal line inside of each box is the
	median value of tip radius. The upper and lower bounds of the box show the
	first (Q_1) and third quartiles (Q_3) of the tip radius distribution, the height of
	the box is the inter-quartile range (IQR). The terminal points of the vertical
	lines extending from the boxes show approximately $\pm 3\sigma$ (i.e. $Q_1 - 1.5 \times IQR$
	and $Q_3 + 1.5 \times IQR$). The starred points are outliers
5-48	Probability density functions of the average effective tip radii of different array
	sizes based on the statistics of the data summarized in Figure 5-47 superim-
	posed a histogram of 200 tip radii measured on a representative sample 146
5-49	Diagram of the measurement Setup 148
5-50	I-V characteristics (left) and Fowler-Nordheim characteristics (right) of the
	device used for emission angle experiments 149

5-51 Dependence of the spot size on the height of the phosphor screen. Spots
produced on the phosphor anode with $V_G = 60$ V, $V_E = 0$ V, $V_A = 900$ V,
$I_A \approx 3.5 \ \mu$ A. The gate leakage is about 50 nA at 60 V, and appears to follow
Fowler-Nordheim behavior. The spot size measurement was taken on the short
axis of the spot. There was a linear relationship between spot size and the height. 150
5-52 Extraction of emission angle from spot size measurements
5-53 Transverse velocity (left) and energy (right) for the different anode-emitter sep-
artions measured
5-54 I-V and FN plots of a device that demonstrated sudden catastrophic failure at
$V_{GE} = 70 \text{ V} \dots $
5-55 Comparison of the finite element simulation of the ridge found at the edge of
the array with a cross-section SEM of the ridge. There is a field enhancement
of approximately 5 at the apex of the ridge
6-1 Example S-K chart From [49] reproduced with permission @2001 AIP 157
6.2 S K Chart of the 0.1 A/cm2 device
6.3 Lifetime summary $1mA = 0.1 A/cm^2 = 10$ Days after initial 70 hour experiment
310+ hours total
6.4 The operating voltage of the 100×100 array increased linearly with time 160
6.5 I.V. share staring the receiver of the 100x100 device
6-5 1-V characteristics showing the recovery of the 100x100 device 160
6-6 S-K chart of the FN characteristics showing the recovery of the 100x100 device. 161
6-7 Lifetime summary - 100x100 array 1mA, 10 A/cm ² , 100 hours total 162
6-8 S-K chart of the 50x50 Array
6-9 S-K Chart of the single emitter shows random walk over time 164
6-10 Lifetime summary - 50x50 array, 2.5 mA, 100 A/cm ² , 5 hours. $\dots \dots \dots 165$
6-11 Lifetime summary - Single Emitter, 1 μ A, 100 A/cm ² , 100 hours total 166
7-1 Comparision of the helium channel transistor to a planar MOSFET 170
7-2 Scattering cross-section for low energy electrons in helium, after [16] 172
7-3 Paschen's curve for helium at 1 atmosphere
7-4 Mean free path of electrons in helium as a function of pressure

7-5	I-V and FN Characteristics of a 1000x1000 array that turns on at 8.5 V \ldots 175
7-6	I-V (top) and FN (bottom) Characteristics of a 1000 × 1000 array operated in
	100 mTorr of helium with $V_A = 19$ V $\dots \dots $
7-7	I-V (A) and FN (B) Characteristics of a 1000×1000 array operated in 500
	Torr of helium with $V_A = 19$ V
7-8	Process for transferring graphene from Cu catalyst foil to a device substrate
	using PMMA. Figure from [78] 179
7-9	Structure of a field emitter device with an electron-transparent graphene gate 180
7-10	I-V (left) and FN Characteristics (right) of a 100×100 array with a graphene,
	electron transparent gate electrode
7-11	Three different approaches for using graphene to separate field emitter tips
	from there operating envrionment
D-1	Tunneling probability through a triangular barrier shows an exponential de- pendence on electric field. The electron has energy normal to the surface $E_x = E_F$. The dashed line shows where significant tunneling occurs, at 2 V/nm. 199
E-1	(A) Ionization Efficiency and Ionized donor concentration vs. Net Doping
	for a 100-nm diameter silicon nanowire in oxide (B) Ionization efficiency for
	several different doping densities for vs. nanowire diameter
F-1	Schematic depiction of the state of the Si/SiO ₂ interface. Dangling bonds give
	rise to trap sites, which can accept electrons and give rise to interface charge.
	An anneal in hydrogen can passivate some of the surface traps sites by allowing
	hydrogen to bond at the dangling sites, reducing the interface trap density 208
F-2	Schematic diagram of a hypothetical silicon nanowire current limiter with in-
	terface charge density, Q_{IT} . The interface charge density has the effect of de-
	pleting the outer part of the silicon nanowire, even with no voltage applied.
	If Q_{IT} is too high relative to the doping density and the tip radius, the entire
	nanowire can be depleted with no potential applied

,

¥

F-3	Band diagram of an Si nanowire with no potential applied, showing the outer
	edge depleted due to interface states. If D_{it} is high enough and the doping
	low enough, the entire nanowire may be depleted
F-4	Schematic diagram of a hypothetical silicon nanowire current limiter with in-
	terface charge density, Q_{it} . The interface charge density has the effect of de-
	pleting the outer part of the silicon nanowire, even with no voltage applied.
	If Q_{it} is too high relative to the doping density and the tip radius, the entire
	nanowire can be depleted with no potential applied
F-5	The nanowire in cross section showing the depletion (shaded) and neutral re-
	gions due to interface states
F-6	Simulations of effective electron concentration (n_{eff}) for various nanowire
	geometries and electrical parameters. (A) n_{eff} vs. N_D for various values of
	D_{it} . The NW radius was fixed at 50 nm. (B) n_{eff} vs R for various values
	of D_{it} . The doping was fixed at 5×10^{15} cm ⁻³ . (C) n_{eff} vs R for various
	doping concentrations. D_{it} was fixed at 10 ¹¹ cm ⁻³ . (D) semilog (left) and
	linear (right) plots of n/N_D as a function of position along the channel for the
	various doping densities in (A) for $D_{it} = 10^{11} \text{ cm}^{-2} \text{ eV}^{-1}$
F-7	(A) Midgap interface trap density vs dry oxidation temperature for p and n-
	type silicon, (100) and (111) orientations. Data are for samples with no hydro-
	gen anneal. Figure reprinted with Permission from [108]. (B) Structure and
	location of traps (Pb centers) on oxidized silicon wafers of the three primary
	orientations. Figure reprinted with permission from [104] 216
F-8	Comparison of extracted linear conductance, g_{lin} with a numerical model of
	linear conductance for a SiNW current limiter

List of Tables

1.1	Properties related to power performance at high frequencies, based on [89] 31
3.1	MEASURED CURRENT-VOLTAGE CHARACTERIZATION DATA 74
5.1	Extracted parameters from the device simulation in ATLAS 102
5.2	Etch Parameters
5.3	Summary of etching conditions used in the circumscribed central composite
	design of experiments 107
5.4	Summary of device performance for the characteristics
5.5	Summary of device performance for the characteristics
5.6	Summary of device performance for the characteristics
6.1	Devices Characterized for Lifetime

Chapter 1

Introduction

1.1 Problem Statement

Semiconductor electronic devices have become ubiquitous since the integrated circuits revolution in the 1960s, displacing vacuum electronics in all but a small number of niche applications. As the frequency and power performance of solid state transistors increased and their cost to manufacture plummeted exponentially according to Moore's Law [90], vacuum electronics could not compete with integrated circuits in a vast number of applications in terms of size, weight, power, or cost.

In some applications, however, creating electronic devices with increasing power output, frequency performance or smaller size becomes much more challenging using semiconductor devices. This may be due to environmental constraints, such as very high temperatures that cause the failure of metal-semiconductor contacts or the intrinsic carriers to dominate the extrinsic doping and thus breaking p-n junctions. Another area where solid-state electronics struggle are in high radiation environments (*i.e.* space), where cosmic rays can generate carriers that change the behavior of logic and memory circuits, leading to errors. Cosmic rays are photons that can have energy > 1 GeV, thus shielding electronics from them is an extreme challenge.

Building high frequency and high power electronic devices is a particular challenge due to carrier transit delays and breakdown of the device. These carrier transit delays set a fundamental limit to the switching frequency of a device and arise from effects of transport in a



Figure 1-1: Applications for Gated Electron Field Emitters

semiconductor crystal and include electron-phonon interactions and ionized impurity scattering [85]. To combat the carrier transit problem, the channel length is reduced, however, by reducing the channel length, the maximum voltage that can be applied between the drain and the source (V_{DS}) is also reduced. A measure of the high frequency and high power performance of a material system or device architecture is given by the Johnson Figure of Merit (*JFOM*) [71]

$$JFOM = \frac{E_{br}v_{sat}}{2\pi} \tag{1.1}$$

where E_{br} is the breakdown field of the channel material, and v_{sat} is the saturation velocity of the electrons in the channel.

Because there are no atoms for electrons to interact with in vacuum, they may be accelerated to velocities approaching the velocity of light, *c*, without scattering. The challenge in this case is designing high voltage systems.

Terahertz radiation is considered the sub-mm wave frequency band between 300 GHz and

	Si	GaAs	4H-SiC	GaN	Diamond	Vac.
E_g (eV)	1.1	1.42	3.26	3.39	5.45	∞
$n_i ({\rm cm}^{-3})$	1.5×10^{10}	1.5×10^{6}	8.2×10^{-9}	1.9×10^{-10}	1.6×10^{-27}	0
μ_n (cm ² /Vs)	1350	8500	700	1200 (bulk) 2000 (2DEG)	1900	≃∞
v_{sat} (10 ⁷ cm/s)	1.0	1.0	2.0	2.5	2.7	<i>≃</i> 300
E_{br} (MV/cm)	0.3	0.4	3.0	3.3	5.6	> 100 [130]
Θ (W/cm K)	1.5	0.43	3.3-4.5	1.3	20	≈ 0
JFOM	1	2.7	20	27.5	50	> 4000

Table 1.1: Properties related to power performance at high frequencies, based on [89]

30.0 THz. The energy of a 1 THz photon is 4 meV, which is near the energy gap between molecular bands [2]. For this reason, THz is particularly exciting for applications investigating the structure and composition of materials, including radar medical imaging (tomography), biological and chemical sensing, and spectroscopy. In addition, while the atmosphere absorbs radiation in the THz regime quite readily, the bandwidth available is enormous, making it a promising band for high-bandwidth satellite-satellite or satellite-aircraft communications. Room-temperature photonics also struggle in the "THz gap". Photonic devices rely on population inversion, and at room temperature, kT is the energy of a 6 THz photon. Because of that, it is a serious challenge for operation below that frequency. Vacuum electronics are an exciting technology because it does not suffer from either of these problems. For a relativistic beam, slow wave interaction structures for 1 THz are approximately 300 μ m: not a particular fabrication challenge using state-of-the-art MEMS technologies.

For an example of a THz system that could benefit from high current, high current density cathodes, Basten et al. recently demonstrated the operation of a 0.85 THz power amplifier using a folded wave guide slow-wave interaction structure Shown in Fig. 1-2[8]. They used a thermionic cathode that required a large permanent magnet for beam forming and compression to obtain 3.1 mA of circuit current and 311 A/cm² circuit current density. If this beam performance could be achieved without large permanent magnets, there could be a dramatic reduction in the size and weight of these systems.

With the possible exception of carbon nanotube cathodes for x-ray sources[45, 56], cold



Figure 1-2: 0.85 THz Vaccum amplifer, reprinted with permission from [8]. ©2012 IEEE

cathode electron sources have yet to fully deliver on their promise of a high-performance cathode adequate for integration into commercial systems. The performance requirements of these applications include beam quality (*low energy dispersion, high brightness, small spot size*), current (*high current, high current density*), uniformity (*spatial uniformity and stability*) and lifetime (*long lifetime*). While electron sources based on field emission have had promise, they have not met the performance goals required for systems.

Through careful examination of the literature, it has become clear that there are four main challenges that have thus far prevented afield emission source from achieving the required metrics:

- large capacitance between the gate and the emitter electrodes that has limited switching and high frequency performance [117, 99];
- 2. insulator breakdown due to injection of charge from either the silicon substrate into the dielectric spacer between the gate and the emitter substrate or the gate electrode pad into the substrate, or flashover of the dielectric where the desorption of gas molecules adsorbed in the dielectric leads to a plasma on the dielectric surface and vaporization of the dielectric [5, 50]

- erosion of field emitter tips due to bombardment by back streaming ions emanating from impact ionization of neutral gas molecules desorbed form the anode or the gate that may include the formation of a plasma
- 4. emitter tip melting due to Joule heating and thermal runaway or cathodic arc. [13, 12].

Temple *et al.* reduced the capacitance between the gate and the substrate of field emitter arrays in order to increase the unity current gain cut-off frequency, f_T .

$$f_T = \frac{g_m}{2\pi C_{GE}} \tag{1.2}$$

where C_{GE} is the gate-emitter capacitance and g_m is the transconductance i.e. $\partial I_A / \partial V_{GE}$. They used silicon pillar structures with aspect ratios of 2.5:1 (column height: pillar diameter). The pillar diameter has an impact on the minimum gate aperture which in turn has an impact of the field factor, β , and consequently the slope of the Fowler-Nordheim plot, b_{FN} . To reduce C_{GE} , further, the aspect ratio of the field emitter arrays need to be increased further.

Holland *et al.* reduced charge injection into the oxide between the gate and the emitter by increasing the insulator thickness to 4 μ m while keeping the gate aperture at a diameter of 1 μ m. Using this structure; they reported a reduction in the electrostatic field across the insulator between the gate electrode and the substrate leading to a dramatic decrease in charge injection and device reliability. Furthermore, the increase in the insulator thickness while keeping gate aperture the same made the structure more amendable to field ionization at relatively low voltage, a physical process that requires more intense electrostatic fields at the tip [114, 40]. In this work, the aspect ratio is increased to >50:1 while the gate aperture is reduced to 350 nm and oxide thickness increased to 10 μ m dramatically reducing the electrostatic field across the gate insulator thereby reducing time dependent dielectric breakdown (TDDB) and hence increase lifetime [87].

Takemura *et al.* showed that a vertical current limiter connected in series to a small array of field emitters could improve reliability while not having any significant impact on the voltage drop across the field emitter and hence the field emission current. They showed that the vertical current limiter improved lifetime of cathodes in traveling wave tubes (TWTs) [120]. Browning et al. had earlier shown that cathodic arcs could be arrested or quenched if a high valued

resistor is placed in series with the emitter or gate of the field emitter with the central idea of limiting the current in the gate/emitter circuit [13, 12, 3]. Velasquez-Garcia *et al.* improved emission current uniformity [55, 124, 125] by incorporating high aspect ratio silicon vertical current limiters [tip-to-tip spacing 10 μ m, column diameter 1 μ m, column height 100 μ m] in series with each emitter tip. The inclusion of a silicon vertical current limiter with current source behavior in series with each emitter tip improved uniformity, severely reduced under-utilization of tips in the arrays and also improved lifetime/reliability.

While Velasquez-Garcia *et al.* were successful in improving emission current uniformity and reliability; their device required high operating gate-to-emitter voltage and only $\approx 1\%$ of the emitted electrons were collected by the anode. This is because they needed to use an external gate electrode that is not self-aligned to the emitter leading to significant interception of the emitted electrons but more importantly high gate-to-emitter distance leads to low field factor and hence high operating voltages. They could not fabricate proximate self-aligned gates because they could not embed their high aspect ratio silicon columns that form the basis for the current limiter in a dielectric matrix.

In this work, the tip density was increased by a factor of 100× compared to Velasquez-Garcia *et al.*, and also the operating voltage was reduced by a factor of 10× allowing the attainment of very high current per tip and current density of 100 A cm⁻² at low operating gate-to-emitter voltages ($V_{GE} < 75$ V) while also achieving long lifetimes (>100 hours @ 100 A cm⁻²). The presence of a current limiter in the emitter circuit improves both emission current uniformity and lifetime/reliability.

None of the previous work of integrating silicon pillars into field emitter arrays has been able to simultaneously achieve high current, high current density, and low voltage while maintaining long lifetime due to the lack of a fabrication process that could simultaneously build, dense, high aspect ratio silicon nanowires with sharp emitter tips, and integrated, self-aligned extraction gates.

This thesis addresses all of the challenges that have prevented field emitters from being viable sources in a holistic manner as follows:

• The silicon nanowire (diameter = 200 nm, column height 10 μ m) embedded in a dielectric matrix proposed here increases the aspect ratio to 50:1 (and potentially 100:1) and thus dramatically reduces C_{GE} and improves switching performance.

- Emission current uniformity is improved because the emission current from sharper tips which turn-on at lower voltages (due to of the higher field factor of sharper emitters) are regulated by the current limiter while the duller tips turn-on at higher voltages and emit lower currents.
- Reliability is improved because the current limiter prevents a tip from ever attaining a current level that is sufficient to result in thermal runaway and melting. The current limiters also help to prevent cathodic arcs or plasma formation by preventing the current runaway needed to sustain the arc or plasma.
- The lifetime is improved because the electrostatic field across the insulator is significantly reduced leading to reduction in time dependent dielectric breakdown (TDDB) and hence longer lifetime.
- The final piece of the puzzle is tip erosion from back ion bombardment or arc formation due to desorption of gas molecules from the anode. For our device characterization, we used a hollow anode structure that prevents desorption of gas molecules from the anode upon energetic electron impact and subsequent ionization of the desorbed molecules. The tip, which is often biased at the lowest potential, is bombarded by the ions generated from gas molecules desorbed from the andoe. The current limiters based on silicon nanowires connected in series with tips also prevent plasma formation and thus increase lifetime.

To address these failure mechanisms, in this thesis we report the design, fabrication process, and characterization of silicon field emitter arrays with integrated silicon nanowire current limiters and self-aligned gate apertures. The field emission cathode has tip radii less than 10 nm integrated with silicon nanowires (\approx 100 nm diameter), high aspect ratio (10 μ m tall), and dense (1 μ m pitch) silicon nanowire arrays. Our solution to the problem results in devices that span three length scales and seven orders of magnitude, creating a number of challenges both in fabrication and in modeling:

• Emitter tip radius — on the order of 1 nm

- Gate diameter and nanowire length on the order of 1 μ m.
- Emitter-Anode separation on the order of 1 cm.

These devices have demonstrated current > 10 mA, current density > 100 A/cm², $V_{GE,ON}$ < 60 V. These results demonstrate that these devices could become an enabling electron source for applications that require high current density cathodes.

1.2 Objectives of This Work

The objectives of this thesis can be summarized as follows:

- Develop a fabrication process for creating silicon nanowire current limiters with silicon field emitter arrays with integrated, self-aligned extraction gate apertures.
- Demonstrate that the integration of the silicon nanowire current limiters results in a cathode with significantly improved performance (current, current density, and reliability) over the state-of-the-art.
- We demonstrated current > 10 mA, current density > 100 A/cm², and lifetime > 100 hours at 100 A/cm².

1.3 Structure of Thesis

The outline of this dissertation is as follows:

Chapter 2 of this dissertation reviews the physics of electron emission, with particular emphasis on field emission. It describes field emission as a two step process, where first electrons are transported to the surface barrier, and then they tunnel through the deformed barrier into vacuum. This concept is central to the hypothesis that by controlling the supply of electrons, we can build more uniform and reliable electron sources.

Chapter 3 introduces the silicon pillar/nanowire current limiter as an attractive option to controlling the supply of electrons. These silicon pillars act as an ideal current source for field emitters, providing simultaneously high current and high dynamic resistance due to velocity
saturation and pinch off in the channel. Because they are located directly underneath each emitter tip in an array, the emitters may be packed densely. This chapter also describes thermal modeling and the second-order effects and non-idealities of silicon nanowire current limiters, such as surface states and 3-D effects.

Chapter 4 presents a proof-of-concept structure that demonstrates that a scaled silicon field emitter array with silicon nanowire current limiters is feasible. This structure is the first to demonstrate saturation effects of a scaled silicon field emitter array with silicon nanowire current limiters. Due to the distance between the gate and the emitter tip, emitter-to-emitter pitch was limited to 5 μ m. SEM imaging of the completed structure revealed that the fabrication process achieved a tip radius < 6 nm with a standard deviation of 1.2 nm.

Chapter 5 presents the gated field emitter array with silicon nanowire current limiters. First, it explores in detail a structure that demonstrates a successful method for simultaneously embedding the silicon nanowires in a high quality dielectric matrix while allowing for an integrated gate with a gate aperture that is self-aligned to the emitter tip. However, this structure suffered from several significant drawbacks including large gate leakage. To combat these problems, a new device design that addressed these problems and attained breakthrough device performance in regards to current density and device lifetime was developed.

In Chapter 6, the lifetime and reliability of the field emission arrays with integrated gates and silicon nanowire current limiters is investigated. Our device characterization shows that the Si nanowire current limiter enables long lifetime operation, with lifetime >10 days @ 100 mA/cm², and >100 hours @ 100A/cm². The I-V characteristics show that there is increase in gate-emitter voltage to achieve the same current with time, but that by allowing the device to relax in a ultra-high vacuum environment, the original I-V characteristics can be restored, indicative of charge trapping at the Si/SiO₂ interface.

Chapter 7 explores some novel device structures and applications that are enabled by the technology developed in this thesis. These include field emission operation in high pressure helium ambient environments and low-voltage transmission of electrons through a graphene window.

Chapter 8 presents a summary of this thesis, lists the contributions, and offers some suggestions for future work.

Chapter 2

Field Emission Theory and Technology

2.1 Objectives

This chapter provides an introduction to the theory of cold field emission, and some of the challenges of field emission. It also introduces the most-researched technologies for microfabricated emitters with self-aligned extractor gates. It covers the advantages and disadvantages of these different methods of building field emitter arrays with self-aligned gates, and it introduces the concept of using supply limitation to improve the reliability and uniformity of field emitter arrays.

Electrons in a metal or semiconductor can overcome the energy barrier at the surface (work function) and be ejected from the metal or semiconductor into vacuum by two different methods. In the case of thermionic emission or photoemission, the electrons are either imparted with thermal energy or photo-energy so they can overcome the potential barrier into vacuum [Fig. 2-1a]. In the case of field emission, however, the potential barrier is deformed by the application of an electric field to the point where electrons can tunnel *through* this barrier and leak into vacuum [Fig. 2-1b].

2.2 Fowler-Nordheim Model

The tunneling current density for electrons through a triangular barrier is given by (For a full derivation, please see Appendix D):



Figure 2-1: Two methods of emission of electrons from a metal. (a) Thermionic or photo emission. (b) Field emission.

$$J_{tnl} = q \frac{4}{3} \frac{16\pi m^* \sqrt{\phi/E_F}}{h^3 B^2 (\phi + E_F)} F^2 \exp\left[-B \frac{\phi^{3/2}}{qF}\right]$$
(2.1)

here q is the electronic charge, m^* is the effective mass, ϕ is the work function, E_F is the Fermi energy, F is the electric field, and B is a constant ($B = 4\sqrt{2m_e}/3\hbar q \approx 6.87 \times 10^7$). While the above equation gives a simple quantum mechanical model for the tunneling current for a triangular barrier, additional considerations must be taken into account [46] to account for image forces that lowers the energy barrier at the surface. Near the surface, the electrons see an image potential V_{im} due to their proximity to the conducting surface, given classically by $V_{im} = -q^2/4x$ [98]. This gives a total potential term of

$$V = -q \left[Fx + \frac{q}{4x} \right] \tag{2.2}$$

This new energy barrier at the surface, while rounded at the top, is still almost triangular, with a maximum height of $\sqrt{\phi - yF^{1/2}}$ and a tunneling width of ϕ/qF . The area under the curve $\sqrt{V(x) - E_x}$ from 0 to W is different from the uncorrected area by a factor of $\alpha = \sqrt{1-y}$, where $y = 3.79 \times 10^{-4} qF^{1/2}/\phi$

The corrected potential term is inserted into the tunneling probability expression to compute the integral in eq. D.12. The result is the Fowler-Nordheim equation.

$$J = \frac{AF^2}{\phi t^2(y)} \exp\left[-B\frac{\phi^{3/2}}{F}\nu(y)\right]$$
(2.3)



Figure 2-2: Comparison of the barrier with image charge correction (solid line) to the uncorrected barrier (dashed line)

Where $t^2(y)$ and v(y) are elliptic functions that take into account the image charge barrier rounding effects, $A = q^3/8\pi h \approx 1.54 \times 10^{-6}$, $B = 4\sqrt{2m_e}/3\hbar q \approx 6.87 \times 10^{-7}$ and $y = \sqrt{q^3 F/\phi} \approx 3.79 \times 10^{-4} F^{1/2}/\phi$.

The Fowler-Nordheim model is not without its limits. First, the above section assumes that the charge cloud terminates abruptly at the surface of the emitter. Instead, the electron cloud extends part way into vacuum. Because of this, the potential step is not as abrupt as depicted.

In the case of a semiconductor, there is also the issue that the electric field penetrates into the material some distance, resulting in band-bending and a potential at the surface, ϕ_s , which makes the surface barrier different from the work function of the bulk. At high fields, the semiconductor will be driven into accumulation, and a lower-dimensionality electron gas will form at the surface.

Generally, the Fowler-Nordheim equation is only valid for planar, metallic surfaces at 0 K. In the case of a nano-fabricated semiconducting field emitter at room temperature, none of these approximations are exactly true. For example, depending on the geometry of the emitter and the material used to fabricate it, the assumption that the electrons are in a 3-dimensional electron gas (3DEG) may not be valid. For example, in a semiconductor, a sheet charge, a line charge, or a point charge may form at the field emitter, resulting in a 2-D, 1-D, or 0-D electron gas. There has been work from Patterson et al. [100] and Qin *et al.* [105] to extend the model to different geometries and different materials. While these solutions are more correct, applying

them to experimental data is much more challenging, and their analytical complexity tends to obfuscate the meaning of the results.

Despite some of these drawbacks, if we continue to use the Fowler-Nordheim model instead of one of the more advanced models, several approximations could be made to make the Fowler-Nordheim model analytically solvable and applicable to nanofabricated field emission triode structures. In order to apply the theory to experiments, the Fowler-Nordheim model must be translated from current density and surface electric field to the experimentally observable quantities of measured current and applied voltage. To translate the current density, J to current, I, we can approximate the emission current as constant over an effective area of emission, α , and make the substitution $J = I/\alpha$.

To translate the field, F to applied voltage, V, we can make the approximation that the surface electrostatic field is proportional with gate voltage, V_G through a constant β , such that $F = \beta V_G$. β is known as the field factor and depends on the geometry of the emitter structure. A large β implies that the effective electric field at the surface of the apex of the emitter tip will be much larger than the macroscopic electric field (i.e. $\beta V_G \gg V_G/(R-r)$). β is described in more detail in Section 2.3 below.

Because the elliptical functions are slowly varying, they can be approximated as $t^2(y) = 1.1$ and $v(y) = 0.95 - y^2$ [11]. With these simplifications, Equation 2.3 becomes:

$$I = \frac{\alpha A \beta^2}{1.1 \phi} \exp\left[\frac{B(1.44 \times 10^{-7})}{\phi^{1/2}}\right] V_G^2 \exp\left[-\frac{0.95 B \phi^{3/2}}{\beta V_G}\right]$$
(2.4)

This equation can be further simplified by the introduction of the FN coefficients commonly found in the literature [116], a_{FN} and b_{FN} :

$$a_{FN} = \frac{\alpha A \beta^2}{1.1 \phi} \exp\left[\frac{B \cdot 1.44 \times 10^{-7}}{\phi^{1/2}}\right]$$
(2.5)

$$b_{FN} = \frac{0.95B\phi^{3/2}}{\beta}$$
(2.6)

Making the appropriate substitutions results in a simplified version of the FN equation that could be used in interpreting experimental data:

$$I = a_{FN} V_G^2 \exp\left[-\frac{b_{FN}}{V_G}\right]$$
(2.7)

2.3 The Field Factor, β

The field factor β is an important approximation for microfabricated, non-planar emitter geometries because it allows us to apply the Fowler-Nordheim model to a wide variety of emitters, instead of just the planar geometry that Fowler and Nordheim first considered. The potential at every point in this system, Φ is a boundary value problem governed by the Laplace equation. A simple analytical solution is found through the approximation of the tip as a hard "ball" and the gate as a spherical shell. The Laplace equation (assuming no space charge in the vicinity of the emitter tip):

$$\nabla^2 \Phi = 0 \tag{2.8}$$

is trivial to solve in spherical coordinates to find the electric field at the surface of the "ball" [68] While the tip is not truly a spherical ball, nor is the gate not a spherical shell, it gives a quick approximation for the field factor:

$$\beta \approx \frac{1}{r} \frac{d}{(d-r)} \tag{2.9}$$

Where Φ is the potential at each point. Typically, r is several orders of magnitude smaller than d (~5 nm compared to ~200-300 nm for an integrated gate electrode. If not integrated, d may be many orders of magnitude larger), so the approximation that $d \gg r$ is valid, resulting in:

$$\beta \approx \frac{1}{r} \tag{2.10}$$

Hence to first order, β is inversely proportional to tip radius. To check the validity of the ball-in-sphere model, 3-D electrostatics simulations of both the ball-in-sphere model and a conical emitter tip with tip radii, r = 10 nm and gate aperture radius d = 175 nm were performed, as shown in Figure 2-3. The top figures depict equipotential surfaces of a quadrant

of the 3-D models used. The bottom figures are 2-D cross-sections at the apex of the emitter with equipotential lines drawn. The ball-in-sphere model tends to slightly overestimate the field at the apex of the emitter. With 50 V applied to the gate, the numerical ball in sphere model predicts an electric field at the apex of the emitter of 5.3×10^7 V/cm, the conical tip model predicts 3.9×10^7 V/cm, and the $\beta = 1/r$ approximation predicts 5×10^7 V/cm.

More careful modeling of the electrostatics have resulted in new analytical and semi-numerical models of field enhancment, including the "bowling pin model" and the "Saturn model," [37, 70, 32]. These models increase the validity of the approximations used and introduce an angular dependence of the field at the emitter tip, which is lost with the ball-in-sphere model. These models suggest that the field factor β can be modeled as varying with tip radius r in $\beta \approx k/r^n$, where k and n are geometry dependent. The exact calculation of these coefficients varies depending on the model used, however, typically for real devices with a gate aperture > 250 nm in diameter and tip radius, r between 1 nm and 10 nm, n is typically close to 0.7 and k is approximately 10⁶.

Examining the geometry of field emission tips fabricated out of silicon using transmission electron microscopy (TEM) imaging, as performed by Pflug [103] in Fig. 2-4, shows that the radius of the field emitters that were fabricated followed log-normal distribution, with radii ranging from 1.4-14nm. Nilsson et. al. have suggested that the field enhancement factor, β follows a Poisson distribution [97], and by implication, that a similar distribution can be deduced for tip radius.

Because of the limits of fabrication at these small scales, it is impossible for a FEA to be completely uniform even across a single die, and large spatial distributions may exist across a wafer. This distribution of tip radius will result in an even larger distribution of emission currents, due to the exponential dependence on tip radius that exists in the FN tunneling model. Thus, even while some tips do have enough electric field to turn on, other tips may have enough electric field are current such that the joule heating could lead to tip destruction.

Fig. 2-5 illustrates the effect of tip radius distribution on emission current. The large, central plot show emission currents for different emitters of a typical field emitter device, assuming an average tip radius $r_0 = 5$ nm and a standard deviation of ~ 1 nm. Within one standard deviation of tip radius, the emission current changes by over 2 orders of maginude at 30 V. For this typical



Figure 2-3: Comparison of simulations of the Ball-in-sphere model and a conical emitter tip with tip radii, r = 10 nm and gate aperture radius d = 175 nm. The top figures depict equipotential contours of a 2-D cross-section of the 3-D models used. The bottom figures are 2-D cross-sections at the apex of the emitter with equipotential lines drawn. The ball-in-sphere model tends to slightly overestimate the field at the apex of the emitter.



Figure 2-4: Tip radius exhibits a log-normal distribution [103]

emitter tip size distribution, it is possible for the sharpest emitters to burn out before the dullest emitters even turn on, thus limiting the overall performance and utilization of the array.

2.4 State of the Art Field Emitter Arrays

Field emitter arrays are very attractive as a replacement for thermionic cathodes, however they suffer from stability and uniformity problems arising from the exponential dependencies of tip emission current on surface electrostatic field as indicated by the Fowler-Nordheim model presented in Section 2.2. Because emission current has an exponential relationship to the emitter radius, it is difficult to make the emission current uniform over large area. In addition, when not operating in a perfect vacuum, gas molecules from the operation environment may absorb onto the surface of the emitter and change the local work function, resulting in large current fluctuations over time.

It is a significant microfabrication challenge to build a patterned gate in close proximity of field emitter tips. To keep the required voltages small, the gate must be as close as possible to the emitter tip. However, to prevent an electrical short between the gate and substrate, as well as dielectric breakdown of the insulator, thick dielectrics must be used. In addition, if the gate aperture is not precisely aligned with the emitter tip in the center of the aperture, the electric field around the tip will be distorted, resulting in an increased transverse force on the emitted



Figure 2-5: The effect of tip radius distribution on emission current. The large, central plot show emission currents for different emitters of a typical field emitter device, and the effect of that the modeled statistical variation in tip radius has on emission current. For this typical emitter tip size distribution, it is possible for the sharpest emitters to burn out before the dullest emitters even turn on, thus limiting the overall performance and utilization of the array.

electrons resulting in interception of the electrons by the gate. Hence, the gate aperture must be self-aligned to the emitter.

There are two main approaches for the microfabrication of field emitter arrays with integrated, self-aligned gates: the additive patterning approach developed by Spindt et al. [116] which uses metal deposition through an aperture and the substractive patterning approach that was developed by Gray et al. at the Naval Research Laboratory (NRL) [51].

2.4.1 Spindt approach

Spindt et al. first demonstrated a microfabricated field emitter using thin film Molybdenum cones in their seminal 1976 paper [116]. Their approach revolves around an angled incidence deposition of a sacrificial layer followed by a vertical deposition of a refractory metal, usually molybdenum, through a gate aperture to form the emitter tip. After the cone deposition, the sacrificial layer is etched, releasing the sharp cone underneath. There have been many refinements to the Spindt process, including 2-step depositions of different materials and different deposition conditions to make for higher aspect ratio emitter cones with thicker dielectrics, and the formation of double gated structures. The Spindt process is the most heavily researched form of field emitter array. The base process for forming Spindt emitters is as follows:

- 1. Oxidize low resistivity, n-type Si wafer to the desired dielectric thickness ($\approx 1 \ \mu m$)
- 2. Deposit molybdenum gate through e-beam evaporation
- 3. Coat with poly-methyl-methacrylate (PMMA) resist
- Use electron beam (e-beam) lithography to to expose PMMA to write gate apertures of approximately 1 μm
- 5. Develop PMMA in isopropyl alcohol and expose underlying molybdenum
- 6. Selectively etch the molybdenum down to the SiO_2 layer
- 7. Strip remaining PMMA, then etch through the SiO_2 in hydrofluoric acid (HF)



Figure 2-6: Schematic diagram of key processing steps in the Spindt approach for the fabrication of field emitter tips with self-aligned gates uses the deposition and lift-off of refractory metals, traditionally Molybdenum.

- Deposit aluminum sacrificial layer at a grazing incidence while rotating the sample. The sacrificial layer also serves to reduce the diameter of the holes that will form the emitter cones
- 9. deposit molybdenum through the partially closed via e-beam deposition using a small source with long throwing distance at normal incidence. The size of the hole continues to shrink during this process. A cone forms in the cavity as the Mo vapor condenses on a smaller area until it is completely closed, leading to the formation of the tips and making the tips relatively uniform
- 10. Etch aluminum and lift-off excess molybdenum
- 11. Clean sample and mount in vacuum system

The benefits of the Spindt approach are that all the depositions occur at room temperature, allowing for a variety of different substrate materials to be used. Because the metals are evaporated, many different emitter materials may be employed in addition to refractory metals, such as low-work function materials. Low work function materials are interesting for field emission because they have the potential to emit more current at lower electric fields, however, they are typically much more reactive and less stable than materials with higher work functions.

A disadvantage of the Spindt process is the emitter tip uniformity that can be achieved through deposition in large area devices.

2.4.2 Self-aligned gate apertures through etching

The other approach for fabricating extraction gates which are self-aligned to the emitter tips uses a combination of deposition etching, and chemical-mechanical polishing (CMP). The use of chemical-mechanical polishing in the fabrication of field emitter arrays was first demonstrated by Cathey and Browning at Micron [18, 14]. There have been further enhancements by groups since, including multiple gates [38, 21], electrostatic lenses[92], and extension to different materials[22].

In the CMP approach, shown schematically in Figure 2-7, emitter tips are either etched (in the case of silicon emitters) or grown (in the case of carbon nanotube emitters). A conformal

dielectric material is then deposited which buries the emitter tips, however, due to the conformal nature of the deposition process, "bumps" are formed where the emitter tips are located. A conducting gate is then comformally deposited. Using chemical mechanical polishing, both materials are removed isotropically, exposing the underlying dielectric where the emitters are. This dielectric then can be removed, exposing the emitter surface.

A simple process for forming sharp silicon emitter tips with is as follows:

- 1. Deposit photoresist on an n-type silicon wafer.
- 2. Expose photoresist "dots" which become the etching mask for silicon tips
- 3. Perform an isotropic etch of the silicon to form rough tips
- 4. Sharpen tips by thermally oxidizing the silicon using traditional semiconductor processing techniques
- 5. Perform CVD deposition of silicon dioxide for the insulator between the emitter and the gate
- 6. LPCVD deposition of in situ doped polysilicon gate material
- 7. Polish the device using a timed chemical-mechanical polishing step to form the gate apertures
- 8. etch the underlying dielectric in hydrofluoric acid to expose the emitter tips
- 9. Load the sample into vacuum chamber for operation

Advantages of the CMP approach are that using silicon emitters or carbon nanotubes, the emitter tip radius may be incredibly sharp[36] [32, 88], and the deposition and CMP process leverage years of technology development and experience in the semiconductor industry on these processes for good uniformity and repeatability.

The primary disadvantage of this approach is that there is added complexity and processing steps.



Figure 2-7: Schematic diagram of key processing steps in two different materials for using chemical-mechanical polishing (CMP) to form emitters with self-aligned gates, (Left) Silicon and (Right) Carbon Nanotubes. In both cases, a conformal oxide deposition forms a "bump" that later becomes the gate aperture when excess material is removed during the polishing step.

2.4.3 Controlling the Electron Supply

Field emitters typically operate in a regime where the current is controlled by the probability of electrons tunneling through the barrier from the metal or semiconductor into vacuum. This is because the number of available carriers at the surface is large, and there is a quasi-neutral bulk nearby with an electrical contact to replenish electrons that are transmitted, as is the case with a metallic or heavily-doped n-type semiconductor. This tunneling-probability regime is extremely sensitive to the tip radius and the work function of the emitter. The introduction of a current limitation element changes the system to make the transport process from the bulk to the emitter surface the dominant process in controlling the current and to reduce the sensitivity of the current to tip radius and work function.

Field emission can be treated as a two-step transport process shown in Figure 2-8. First, electrons are transported to the surface barrier. This is included in the Fowler-Nordheim derivation shown in Appendix D as the supply function N(E), but is often obfuscated in simplifications of the Fowler-Nordheim equation. This supply function gives a number of electrons that are impinging on the surface barrier. At the surface barrier, there is a probability that the electrons impinging on the barrier will be transmitted through the surface barrier instead of being reflected.

Most prior work on limiting the supply of electrons to the surface of field emitters focused on using high-resistivity substrates [3, 64] to approximate a current source. The effect of adding



Figure 2-8: Block diagram of the factors involved with field emission. A field emitter may be limited by either the transmission at the surface, or the flux to the surface

a large resistance in series with the emission results in the load line shown in fig 2-9 with respect to tip radius variation. Increasing emission current results in a voltage divider, with part of the voltage dropping across the resistor, reducing the gate-to-emitter voltage and resulting in lower emission current. Adding this resistance improves emitter reliability and reduces cathodic arcs by quenching the formation of a microplasma [13], but it is at the cost of operating voltage and efficiency. In order to reduce sensitivity, the resistance must be large, however, with a linear resistor, it is impossible to simultaneously achieve high current and low sensitivity to tip radius.

To address the non-idealities of the resistive element in series with the FEAs and and closer approximate an ideal current source, MOSFET structures have been integrated into FEAs to control individual [67] or small groups of emitters [62] [74]. The variation of emission current is much less when a MOSFET that is in saturation is used as a current control element compared to when limited by a resistor (provided that the FET output resistance is much less than the



Figure 2-9: Qualitative load lines of different types of current sources connected in series with each individual field emitter, demonstrating the variation of emission current for different tip radii. The dashed blue line shows that even if the output resistance of the FET equaled the that of the resistor, the FET would provide more emission current

linear resistance, *ie.* $g_{out} \ll 1/R$ [Fig. 2-9]. The sensitivity of the current S is given by:

$$S = \frac{I_0}{\Delta I} \tag{2.11}$$

where I_0 is the current, and ΔI is the derivative, i.e. dI/dt at $I = I_0$. Hence, a MOSFET with higher current, I_0 , will have a lower sensitivity than a linear resistor.

This combination of high conductance at low V_{DS} and high dynamic resistance in the saturation regime is precisely the behavior required to implement a good current source. The drawback of using a MOSFET to approximate current sources is that planar MOSFETs require significant area in the array, reducing the packing density and hence the attainable current density of the array. A possible trade-off is to use a MOSFET to drive a small number of emitter elements, such as was implemented by Hong et al. and Itoh et al. [62, 67]. In this scheme, however, emission non-uniformity will still occur in the smaller set of emitters controlled by the same transistor. To control each individual emitter with a conventional lateral MOSFET is not advantageous because the relatively large area of the MOSFET will result in a greatly reduced packing density.

Through careful design of the current limiting element, the performance of the array can be improved when compared to that of unlimited arrays. If the limited current is high enough while still preventing burnout, higher current may be obtained than without limitation. Figure 2-10 demonstrates by plotting the hypothetical current of an ideal array, the performance of an array whose emitters burnout at 10 μ A, and an array whose current is limited to slightly below burnout $I_{limit} = 5 \mu$ A. Note the hysteresis of the simulation of the array with burnout. During the initial sweep, as the voltage is increased, the anode current decreases due to the burnout of more and more emitter tips. On the down sweep, and on subsequent sweeps, the burnt out emitters do not participate in field emission, and the emission current is several orders of magnitude lower than on the initial sweep.

2.5 Technical Approach

A suitable current limiter has been recently demonstrated: the vertical silicon pillar current limiter. This two-terminal device utilizes velocity saturation of carriers in semiconductor ma-



Figure 2-10: The effect of tip burnout on current performance. By limiting the current to each individual emitter in an array, the overall performance of the array can be improved. Note the hysteresis of the simulation of the array with burnout. During the initial sweep, as the voltage is increased, the anode current decreases due to the burnout of more and more emitter tips. On the down sweep and on subsequent sweeps, the burnt out emitters do not participate in field emission.

terials at sufficiently high fields to create a large dynamic resistance. This device was motivated by early work in GaAs by Baek *et al.* where a voltage difference applied between two contacts separated by a distance L generated the required field for velocity saturation [7]. At low fields, the device behaves like a linear resistor. Above a critical voltage, the velocity of carriers in the channel saturates, resulting in pinch-off and current saturation. A schematic diagram of the the device in GaAs reported in [7] is shown in Figure 2-11, and their obtained current-voltage characteristics are shown in Figure 2-12.

In the proposed device, the contacts are spaced much further apart with a small crosssectional area, resulting in a high aspect ratio structure. The silicon pillars are embedded in an oxide, which passivates the surface and reduces the density of interface traps. When a voltage is applied to the drain end of the channel with respect to the source, the drain electric field creates a channel electric field. The drain potential also depletes the surface of the silicon column at the drain end, narrowing the channel width. The channel becomes narrower as the drain potential increases, eventually resulting in pinch-off. The current saturates due to a combination of the



Figure 2-11: Schematic cross-section diagram of the ungated FET current limiter by Baek. Reprinted with permission from [7]. ©1985 IEEE.



Fig. 8. Comparison of the theoretical results with the experimental data (solid line-experiment; dashed line-theory):

 $W = 20 \times 10^{-6} \text{ m}, \qquad A = 1.25 \times 10^{-7} \text{ m}, \\ N_d = 1.74 \times 10^{23} \text{ m}^{-3}, \qquad \mu_n = 0.35 \text{ m}^2/\text{V-s} \\ \text{(a) } V_{S\text{bi}} = 0.47 \text{ V}, \qquad v_s = 1.21 \times 10^5 \text{ m/s}, \qquad R_c = 0.74 \text{ }\Omega\text{-mm} \\ \text{(b) } V_{S\text{bi}} = 0.46 \text{ V}, \qquad v_s = 1.20 \times 10^5 \text{ m/s}, \qquad R_c = 0.88 \text{ }\Omega\text{-mm}. \end{cases}$

Figure 2-12: Measured and simulated I-V characteristics from Baek et al. Reprinted with permission from [7]. ©1985 IEEE.

pinch-off and the saturation of electron velocity under large electric fields. If a lower aspect ratio were used, a higher drain-source voltage is required to pinch off the channel.

Velásquez-García *et al.* demonstrated a 1 cm² array with 10 μ m pitch (*i.e.* emitter density of 10⁶ cm⁻² for an array of 1000 × 1000 emitters). Each field emitter is atop a vertical silicon pillar that is 1 μ m in diameter and × 100 μ m tall [125]. The tip radius was measured to be

 \simeq 30 nm. These arrays lacked an integrated gate electrode, and thus the voltages required for field emission were > 200 V. They demonstrated current saturation at gate-emitter voltage of 1600 V.

These field emitter arrays showed very good stability and uniformity and prevent the destructive heating of the sharpest tips, however, the 1 μ m pitch limits the current density that may be obtained from this device. The highest current and current density reported from these device are 0.6 A and 0.6 A/cm². The gate was not integrated with the emitter substrate and aligned by hand. Consequently the alignment of the emitters to the gate apertures was poor resulting in an anode efficiency (I_A/I_E) that was less than 1 %.

Numerical simulations, presented in detail in Chapter 3 indicate that the pillar dimensions can be scaled from 100 μ m tall, and 1 μ diameter to 10 μ m tall and 100 nm diameter, and that by increasing the doping density, the current per tip can remain constant (analogous to Dennard scaling rules for CMOS [27]).



Figure 2-13: (a) Array of field emitters with 10 μ m pitch, 1 μ m diameter, and 100 μ m tall Si pillar current limiters. (b) Characterization setup (c)Fowler-Nordheim plot of a field emitter array with individual silicon pillar current limiters that demonstrates current control by the current limiter. Due to the dependencies of the FN equation, if $\ln(\frac{1}{V^2})$ is plotted against V^{-1} , the graph should be a straight line if the emission mechanism is field emission. This device shows a clear deviation from linear behavior at large voltages. Reprinted with permission from [125]. ©2011 IEEE.

Chapter 3

The Vertical Silicon Nanowire Current Limiter

3.1 Introduction

The objective of this chapter is:

- 1. Introduce the Silicon nanowire current limiter as a means to regulate the supply of current to individual field emitters
- 2. Build an intuitive model of the operation of silicon nanowire current limiters
- 3. Measure the current-voltage characteristics of Si nanowire current limiters, and compare experimental data with models
- 4. Develop device physics models for surface traps of silicon nanowires as a means to explain the current obtained from silicon nanowires and develop methods to mitigate them
- 5. Model the interactions between Si nanowire current limiters when operated in parallel in arays, as in FEAs

In Figure 3-1, a schematic drawing (Figure 3-1a) and equivalent circuit diagram (Figure 3-1b) of the FEA and current limiter are shown. The nanowire current limiter has a current source-like I-V characteristic when biased at voltages larger than its saturation voltage [7, 9]. Due to the geometry, the drain region pinches off and the electron velocity saturates, resulting in current saturation.

When vertical current limiters are connected in series with individual field emitters, they limit the current from each field emitter in the array, allowing for uniform emission without thermal runaway or burnout, provided their saturation current is below the burnout limit as demonstrated by Velásquez-Garcá *et al.* [125]. This current limitation is consistent with the operation of the device in the electron supply controlled regime instead of the electron transmission controlled regime as observed by Ding et al. for Si field emitter arrays [62, 31] allowing for reliable operation of FEAs at high currents.

The combined current limiter-FEAs reported by Velásquez-García et al. [125] had Si pillars with a diameter of 1 μ m, height of 100 μ m, and 10 μ m pitch, resulting in a density of 10⁶ tips/cm². The FEAs are capable of high and uniform current emission (0.5 A and 0.5 A/cm²). As mentioned in Section 2.5, the arrays reported by Velásquez-García et al. demonstrated stable and uniform emission, and regulation of the emission current by the silicon pillar. Their arrays had pulsed current and current density up to 0.5 A and 0.5 A/cm², and the current that they were able to obtain was limited by interception of electrons by the gate electrode, resulting in heating of the gate leading to reflow and failure of the polymer spacer used to separate the gate from the emitter tips. Because the gate was not integrated with the emitters, and aligned by hand, the anode efficiency ($I_A/I_E \times 100\%$) was less than 1 %. With a gateemitter voltage of > 1000 V and current of 0.5 A, the gate must — in ultra-high-vacuum dissipate a power of > 500 W (and > 500W/cm²), > 5× higher than the power dissipated by current microprocessors. Hence, to limit the power dissipated, the characterization at gate voltages higher than 1000 V the arrays were pulsed with a duty cycle of 10⁻⁵, lowering the average power dissipation to 50 mW.

By spacing the emitters closer together and the making the tips sharper, the operating voltage can be reduced and the current density increased. However, without including an integrated gate, the electric field at the surface of the emitter is reduced due to electrostatic screening of adjacent emitters (For modeling of this effect, please see Section 4.2). Hence, if the distance between emitter tips is to be reduced, a gate must be much closer to the emitter tip (*i.e.* integrated and self-aligned to the emitters). In order to form a self-aligned gate, the gaps between adjacent pillars must be filled in with a dielectric, and the filling of voids that are 100 μ m deep is an unsolved problem. In addition, if the emitter-to-emitter spacing were to be reduced, deep reactive ion etching (DRIE) limits how closely the emitters may be spaced. DRIE of holes is limited to aspect ratios of $\approx 20 : 1[129]$. Finally, forming emitter tips that are less than 10 nm in radius with this structure uniformly due to etching and oxidation rate variations presents a challenge due to the size of the etching mask used in the fabrication process.

For these reasons, we decided to scale the current limiter demonstrated by Velásquez-García et al. by an order of magnitude, replacing the 1 μ m diameter, 100 μ tall Si pillar with a 100 nm diameter, 10 μ m tall Si nanowire. Reducing all of the dimensions of the current limiter-FEA structure provides several benefits. By shrinking the pitch of the emitters, the tip density is increased. Because the pillar cross-sectional area is smaller, the saturation current for a given doping density is lower; however, by increasing the doping in the pillar, the current density increases, allowing the same current per emitter to be obtained from narrower pillars. In addition, the gate aperture scales with the pillar diameter. Thus, by decreasing the pillar diameter — and



Figure 3-1: (a) Schematic diagram of a single field emitter in series with a current limiter. (b) Circuit diagram of the current limiter-FEA structure. (c) SEM of the completed current limiter-FEA structure. (d) Cross-sectional SEM of a Si nanowire current limiter without a field emitter with oxide removed to show the pillar. This chapter focuses on modeling and characterization of the Si Nanowire current limiter.

by implication the gate aperture — the field factor, β (cm⁻¹), which relates the gate voltage to the electrostatic field at the tip surface increases, resulting in lower turn-on and operating voltages. In practice, Pflug et al. demonstrated gate apertures as small as 70 nm, resulting in a turn-on voltage of 8.5 V [102].

A possible consequence of using current limiters to regulate FEAs is the potentially larger energy spread of the emitted electrons when compared to un-regulated FEAs. In an unregulated FEA, each emitter in the array has the same gate-emitter voltage bias. However, when a current limiter controls the current through each emitter, the gate voltage, V_G , is divided between the gate-emitter voltage, $V_G E$, and the drain-source voltage, V_{DS} , of the vertical current limiter, i.e. $V_G = V_{GE} + V_{DS}$. Because each emitter is biased at a different gate-emitter voltage, electrons from each emitter are accelerated by different voltages, resulting in a wider energy distribution. The energy spread of the FEA current could potentially be reduced by shrinking the gate aperture, and thus increasing β , while keeping its uniformity constant. Using an FEM simulation platform, we estimated that by increasing β by a factor of 5 while keeping the same tip radius statistics, the energy spread across the FEA decreased from 5 eV to 2 eV.

3.2 Modeling the Si Nanowire Current Limiter — An Intuitive Model

Figure 3-2 shows a typical I-V transfer characteristic for a vertical current limiter, with different regions of operation highlighted. In the following section, we will explore the physics that give rise to the current saturation shown in this graph.

The carrier flow in electronics is most generally described by the Boltzmann Transport Equation (BTE). However, for the purpose of developing a simple analytical model for the description of the operation of the current limiter, beginning with drift-diffusion equation will suffice (although, it is important to note that the drift-diffusion equation can be derived from the BTE). For this discussion, the hole current will be neglected, as the substrate used to fabricate the Si nanowire current limiters is n-type, and the device is operated in the dark, so that there are no photo-generated minority carriers. The substrate is uniformly doped so that there are neither junctions nor large electron concentration gradients, so the diffusion term may also



Figure 3-2: Schematic transfer characteristic for the vertical current limiter.

be safely neglected. The problem may be even further reduced by noting that all of the carrier flow will be along the axial direction of the pillar, resulting in a one-dimensional problem.

$$J \approx -q \, n v_e^{d \, rift}(\mathcal{E}) \tag{3.1}$$

At low fields, the current density is proportional to the electric field through the electron mobility, μ_e .

$$J = -q n \mu_e \mathcal{E} \tag{3.2}$$

Where q is the fundamental charge, n is the carrier concentration, v_e^{drift} is the drift velocity of the electrons, and \mathscr{E} is the electric field. Assuming that the pillar has a cross-sectional area, A, and a length, L, the total current supplied to the field-emitter may be obtained. At low voltages, the voltage drop is expected to be linear along the length of the pillar, that is $\mathscr{E} = -\nabla V = -dV(x)/dx$ is a constant, yielding the following expressions for the drain current, I_D and the linear resistance, G_{LIN} [S]:

$$I_D = qAn\mu_e \frac{dV(x)}{dx} \approx \frac{qAn\mu_e V_{DS}}{L}$$
(3.3)

$$g_{lin} = \frac{qAn\mu_e}{L} \tag{3.4}$$

However, at higher electric fields, the velocity of electrons begins to saturate. In silicon, the saturation velocity, v_{sat} is $\approx 1 \times 10^7$ cm/s. To describe the saturation effect, no longer can a linear relationship between current density and electric field be assumed. The drift velocity must now be replaced by the following simple analytic expression:

$$v_e^{drift} = \frac{\mu_e \mathscr{E}}{\sqrt{1 + \left(\frac{\mu_e \mathscr{E}}{v_{,at}}\right)^2}}$$
(3.5)

In addition to the field-dependent mobility, the source depletion layer increases from the source to the drain end of the nanowire, resulting in a cross-sectional area, A(x), that decreases along the length of the channel. The full drain current expression is:

$$I_D = \frac{qA(x)n\mu_e}{\sqrt{1 + \left(\frac{\mu_e}{v_{sat}}\right)^2 \left(\frac{dV(x)}{dx}\right)^2}} \frac{dV(x)}{dx}$$
(3.6)

Above a certain V_{DS} , the velocity of the electrons reaches v_{sat} , and the electron concentration in the drain end of the channel drops off substantially. Defined as V_{DSS} , it is at this voltage that the channel is pinched off and the current reaches its saturation value, I_{DSS} . Increasing V_{DS} beyond this value causes the overdrive voltage $\Delta V_{DS} = V_{DS} - V_{DSS}$ to be dropped across the depletion region. As ΔV_{DS} is further increased, the depletion region widens, effectively shortening the length of the channel by an amount, ΔL . This effect, known as channel length modulation, can be modeled as a linear increase in the drain current for $V_{DS} > V_{DSS}$. Figure 3-3 illustrates the effect of increasing V_{DS} beyond V_{DSS} on the equipotential lines in the device. For the illustration, it is assumed that the electric field at the edges of the illustration normal to that surface is 0. The current in the saturation regime is thus:

$$I_D \cong I_{DSS}[1 + \lambda V_{DS}] = I_{DSS} + g_{out} \Delta V_{DS}$$
(3.7)

where λ [V⁻¹] is the channel length modulation parameter and $g_{out} \approx \lambda I_{DSS}$ [s] is the resulting output conductance. While the behavior of channel length modulation in Si nanowire current limiters is largely analogous to its behavior in standard planar MOSFETs, the channel pinch-off in the current limiter is a 3-D effect, as the extension of the depletion region varies in both orthogonal directions perpendicular to the axial direction.

Another consideration is that the dependence of I_D on V_{DS} beyond V_{DSS} could be explained by drain induced barrier lowering (DIBL). DIBL is a short channel effect in MOSFETs where increasing the drain voltage affects the barrier at the source end of the channel, increasing the injection of charge into the channel. For our devices, it seems unlikely that DIBL plays a major role. The silicon nanowire is long ($\approx 10 \ \mu$ m, and the voltages applied are moderate, so that the electric field at the source end of the channel remains low. Finally, because there is no junction in our device, there is no barrier to the injection of carriers into the silicon nanowire.

3.3 Numerical Modeling

To estimate the performance of the vertical current limiters and serve as a guide to their fabrication, process and device simulations were performed using the SILVACO toolset (Silvaco International, Santa Clara, CA). For these simulations, the Si pillar cross-sectional area was fixed at 100 nm × 100 nm. The area surrounding the pillar was filled with SiO₂, and both the top drain contact and substrate source contact were assumed to be perfect ohmic contacts. No interface states or fixed charge at the Si/SiO₂ interface were included in these simulations.

Once the structure was obtained, ATLAS was used to simulate the current-voltage characteristics of a single current limiter. These simulations solved Poisson's equation self-consistently with the carrier flow and continuity equations. To explore the functional dependencies on channel length and doping, the doping density, N_D , was varied between 10^{13} cm⁻³ and 10^{16} cm⁻³, and the channel length, *L*, was varied between 1 μ m and 10 μ m. For the simulations where N_D was varied, the channel length was fixed at 10 μ m, and for the simulations where *L* was varied, N_D was fixed at 5×10^{14} cm⁻³. Figure 3-4 shows a representative I-V characteristic



Figure 3-3: Schematic device cross-section showing the evolution of the equipotential lines and depletion width.

from the simulation framework.

The linear conductance (g_{lin}) , the output conductance, (g_o) , and the drain saturation current per pillar (I_{DSS}) were extracted from the simulations. The dependence of g_o , and g_{lin} on the doping density is plotted in Figure 3-5(a). In Figure 3-5(b), the ratio of g_{lin}/g_{out} is plotted. This metric is an indicator of how well a device with given doping density saturates. In general, nanowires with doping density < 10^{15} can have a linear conductance that is > $1000 \times$ larger than their dynamic conductance. In Figure 3-5(c), the dependence of g_o and g_{lin} on the channel length are shown. An aspect ratio greater than 50:1 is required to obtain a sufficiently large dynamic resistance while simultaneously providing the large current per tip required for high performance field emitter arrays. The saturation current is shown in Figure 3-6.

We followed the sensitivity analysis approach originally proposed by Hong et al. [60] and adopted by Velásquez-García et al. [125] to determine the dynamic resistance ($r_o = 10^{11} \Omega$) required for uniform emission current. Increasing the aspect ratio much beyond 100:1, however, gives diminishing returns, with the output conductance, linear conductance, and saturation current approaching asymptotic values. In addition, current limiters with aspect ratios larger



Figure 3-4: Representative simulation of a silicon nanowire current limiter with 200 nm diameter, length of 8 μ m, and $N_D = 5 \times 10^{15}$ cm⁻³.

than 100:1 further complicate fabrication. Based on these simulations, wafers with doping density of 2×10^{14} cm⁻³ and a target pillar height of 10 μ m (corresponding to a pillar aspect ratio of 100:1) were chosen to theoretically obtain devices with a saturation current of 1 nA/pillar and an output resistance > $10^{11} \Omega$.

Our simulation results predicted the saturation current, output conductance, and linear conductance that were about a factor of 30 larger than what was extracted from I-V characterization of the fabricated devices. There are several important phenomena in the fabricated Si nanowire devices that these simulation models did not capture. The first is that doping tends to follow a Poission distribution. With a doping density of 2×10^{14} cm⁻³, assuming a Poisson distribution of dopant atoms, there are approximately 35 donor atoms in the entire 10 μ m long channel. Because of the small number of dopant atoms, we expect random dopant fluctuations (RDF) to result in large variations from nanowire to nanowire. Additionally, due to the electrostatics of the nanowire, the effective doping can be even lower due to incomplete donor ionization [29, 28].

Interface traps at the Si/SiO₂ interface, D_{it} have a substantial impact on the silicon nanowire current limiters. For a complete treatment of the effect of interface traps on the Si nanowire, see Section F. While RDF and interface traps typically manifest themselves as threshold voltage shifts in MOS devices [59, 76, 112]; in the nanowire, both RDF and interface traps directly affect the channel conductivity, and in large part explain the variation of saturation currents and linear conductivities measured in the actual devices. With low doping and moderate interface trap density, the models suggest that it is possible to fully deplete the nanowire without any voltage applied.

3.4 I-V Characteristics

I-V characterization took place in a probe station using an Agilent 4156C semiconductor parameter analyzer. During the measurement, the substrate (and the source end of the channel) was held at 0 V, and a positive voltage, V_{DS} , was applied to the drain contact. Table 3.1 is a summary of experimental I-V measurements performed. In general, the devices demonstrate excellent current saturation, with a saturation current (I_{DSS}) of approximately 15 pA/pillar and



Figure 3-5: (a) Effect channel doping has on linear and output conductances with a channel length of 10 μ m. (b) g_{lin}/g_o vs. doping. (c) Effect channel doping has on linear and output conductances with a channel doping of 5×10^{14} cm⁻³. (d) g_{lin}/g_o vs. channel length.



Figure 3-6: (top) Effect channel doping has on saturation current with a channel length of 10 μ m. (bottom) Effect channel length has on saturation current with a channel doping of 5×10^{14} cm⁻³.

an output conductance (g_{lin}) less than 1.8×10^{-13} S/pillar. The current saturates at a drain-tosource saturation voltage (V_{DSS}) under 0.2 V. A linear conductance (g_lin) of up to 2.6×10^{-10} S/pillar was measured.

To ensure that the resistance was not a direct result of contact resistance, transfer length measurements (TLM) were performed. From the TLM structures, a specific contact resistance was $3.2 \times 10^{-3} \Omega/cm^2$ was obtained, resulting in an estimated contact resistance of $3.80 \times 10^7 \Omega$
to the pillar, several of orders of magnitude less than the linear resistance. The above extraction of contact resistance likely underestimates the contact resistance to the pillars, as the contact openings in the TLM test structures were 300 μ m × 10 μ m, whereas the pillar cross-section is 100 nm × 100 nm.



Figure 3-7: I-V Characteristics of single current limiters (left) and an array of 4×10^6 current limiters (right).

Figure 3-7 shows the I-V characteristics for several single current limiters in a single array and an array of 4M current limiters, and Table 3.1 summarizes the I-V characterization for many different array sizes. These I-V characteristics demonstrate the current limiting capabilities of the scaled current limiter structure.

The measured current/nanowire is much lower than the both the analytical model and the numerical simulations predict. Two non-idealities that could impact the electrical characteristics and ultimate performance of silicon nanowires are the effect of the Si/SiO₂ interface along the surface of the nanowire and the incomplete donor ionization that results in silicon nanowires. For a discussion of incomplete donor ionization, please see Appendix E. The effect of interface states is covered in detail in Appendix F. Briefly, as the diameter of the nanowire is reduced, the surface-to-volume ratio increases and surface effects can become a limiting factor to performance. Incomplete donor ionization does not appear to be a factor in the current limiters, but for the doping density and device geometry chosen, the interface charge Q_{it} that arises from a moderate interface trap density could fully deplete the Si nanowires.

Size of Array	IDSS/pillar [pA]	VDSS [V]	glin/pillar [pS]	gout/pillar [pS]
1	8.5	0.06	150	0.33
4	0.50	0.20	6.7	0.18
5	1.3	0.20	12	0.061
8	2.2	0.80	28	0.096
62.5k	2.6	0.01	23	-
125k	1.8	0.08	22	-
250k	1.9	0.06	33	-
500k	2.1	0.06	38	0.0060
1M	15	0.15	250	0.090
2M	10	0.08	42	0.022
4M	4.7	0.05	99	0.0065

a. Indicates negative resistance in the saturation regime Table 3.1: MEASURED CURRENT-VOLTAGE CHARACTERIZATION DATA

To explore the effect of temperature on current-voltage relationship of silicon nanowire current limiters and explore the trap occupancy/density, we collected several $I_D - V_{DS}$ sweeps of a silicon nanowire current limiter at different temperatures, ranging from 274 K to 403 K. The current-voltage characteristics are shown in Figure 3-8. The linear conductance, g_{lin} and saturation current I_{sat} were extracted from the I-V characteristics, and plotted on a semilog scale. The activation energy of both the saturation current and the linear conductance varies with $\simeq E_g/2$.

In a semiconductor that is extrinsically doped, the majority carrier concentration and thus the current-voltage characteristics of majority carrier devices devices should not shift very much over this temperature range. This strong temperature dependence of the I-V characteristics with activation energy $\simeq E_g$ suggests that perhaps the nanowire is fully depleted, consistent with the modeling of interface states. The temperature dependence of the intrinsic carrier concentration n_i can be modeled near room temperature as:

$$n_{i} = \sqrt{n_{0} \cdot p_{0}} = \sqrt{N_{c} N_{v}} \left(\frac{T}{300}\right)^{3/2} \exp\left(-\frac{E_{g}(T)}{2kT}\right)$$
(3.8)

Where n_0 and p_0 are the equilibrium electron and hole concentrations, respectively, N_c and N_v are the effective density of states in the conduction and valence bands at 300 K, respectively, and E_g is the band gap of silicon. Around 300 K, the temperature dependence of E_g is roughly



Figure 3-8: (A) Current-voltage characteristics of a single SiNW at a variety of different temperatures. (B) Arrhenius plot of linear conductance, g_{lin} shows a nearly linear relationship, with an activation energy of ~ $E_g/2$ (C) Arrhenius plot of saturation current, I_{sat} shows linear relationship, with an activation energy of ~ $E_g/2$.

linear, and can be modeled as $E_g(T) = E_{go} + CT$ where $E_{g0} = 1.206$ and $C = 2.73 \times 10^{-4}$ [26]. If we assume that the nanowire is fully depleted due to interface traps, the saturation current is:

$$I_{sat} = q A_{NW} n_i v_{sat} \tag{3.9}$$



Figure 3-9: Comparison of the n_i extracted from the saturation current to a model for intrinsic carrier concentration, assuming a nanowire radius of 50 nm.

Where n_i is the intrinsic carrier concentration and v_{sat} is the saturation velocity of electrons. A comparison of the n_i extracted from the saturation current to the model in Equation 3.8 is shown in Figure 3-9. Using the current to extract the intrinsic carrier concentration tends to underestimate the temperature dependence at T > 300 K. A possible explanation for this difference is that a fraction of the carriers injected into the depleted Si nanowire are recombining at the Si/SiO₂ interface. The interface traps have a surface recombination velocity (SRV) S associated with them which depends on the interface trap density D_{it} and the capture crosssection σ . The recombination velocity gives rise to a recombination current:

$$I_{rec} = A_{surf} q S n_i \tag{3.10}$$

where A_{surf} is the surface area of the nanowire ($A_{surf} = 2\pi r L$). It has been suggested in the literature that the surface recombination velocity, *S*, has a temperature dependence[44]. If the difference in the intrinsic carrier concentration extracted from the I-V characteristics and the model is due to surface recombination, we can estimate *S* from the difference between the two.



Figure 3-10: Estimation of surface recombination velocity from the I-V characteristics

Figure 3-10 shows the estimated surface recombination velocity as a function of temperature. The estimated surface recombination velocity ($\sim 10^4 - 10^5$ cm/s) is higher than expected for a well-passivated Si/SiO₂ interface.

These results suggest that the lower than expected drain current of the nanowire current limiters is likely due to a combination of depletion by the interface trap density and recombination at the silicon nanowire surface.

3.5 Array Effects

Another phenomenon not taken into account by previous simulations is the effect of neighboring nanowires when they are arranged in an array. In the case of the simulations of the single nanowire, the area surrounding the nanowire is etched and a region of quasi-neutral silicon spaced about half a micron away from the nanowire remains. When a positive voltage is applied to the top of the nanowire, using the gradual channel approximation, the drain end of the nanowire will be at V_{DS} . The substrate, some distance away, will remain approximately ground. This substrate acts like the gate in a MOSFET and will form a depletion region in the nanowire. Depending on the doping and the geometry of the nanowire, this effect may cause channel pinchoff earlier than velocity saturation.

To explore this effect, we will analytically examine the electrostatics of two situations:



Figure 3-11: Schematic drawing of a silicon nanowire surrounded by oxide and a neutral substrate. the substrate/oxide/nanowire structure forms a parasitic, unwanted MOS structure.

- 1. A cylindrical nanowire surrounded by the substrate (Figure 3-11)
- 2. a cylindrical nanowire buried in oxide with the substate very far away

Following the analytical treatment of this problem, we will present numerical, 3-D device simulations. To treat this problem analytically, we will make arguments similar to those used when treating 2-D planar MOSFETs, but solving Poisson's equation in cylindrical coordinates as was performed in the treatment of traps at the nanowire/SiO₂ interface. We consider a nanowire that is uniformly doped and cylindrical, with radius *R*. The uniform doping in the semiconductor is equal to the substrate doping. The oxide is uniform, and has thickness $t_{ox} = a - R$. The nanowire is *L* microns long. We make (1) the gradual channel approximation, where we assume that the vertical electric field along the channel varies much slower than the radial electric field i.e. $\partial E_x/\partial x \ll \partial E_r/\partial r$ (2) The full depletion approximation, in that space charge $\rho(x, y) = 0$ when the semiconductor is quasineutral, and $\rho(x, y) = qN_D$ where depleted. (3) Neglect any interface states or fixed charge at the Si/SiO₂ interface.

Consider this structure with a voltage V_{DS} applied between the drain of the nanowire and the substrate. At the drain end of the nanowire, the channel potential is thus:

$$V(x) = V(0) = V_{DS}$$
 (3.11)

With the surrounding silicon substrate essentially at ground, this V_{DS} will give rise to a depletion region at the surface of the nanowire. We will denote the radius of the edge of the depletion region as r_d . From the charge neutrality condition, a corresponding accumulation layer will from in the substrate surrounding the nanowire. The charge in this accumulation layer must be equal to the space charge inside of the nanowire. These charges are denoted Q_{Acc} and Q_{NW} , respectively. Thus:

$$Q_{NW} = -Q_{Acc} \tag{3.12}$$

where:

$$Q_{NW} = q \,\pi N_D (R^2 - r_d^2) \tag{3.13}$$

Let us consider the case where the nanowire is on the onset of being fully depleted, i.e. $r_d = 0$. In this case, that we will define as "pinch-off", the potential at the very center of the channel is $V_{DS,sat}$, and the surface potential, ϕ_s at pinch-off is found through the solution of Poisson's equation:

$$\phi_{s}(x=0) = V_{DS} - \frac{qN_{D}R^{2}}{4\epsilon_{Si}}$$
(3.14)

The voltage drop across the oxide can be calculated with the capacitance per unit length and the charge per unit length in the semiconductor through $V = Q_s/C_{ox}$. The capacitance of the oxide is the capacitance of a coaxial structure with inner radius R and outer radius $a = t_{ox} + R$:

$$C_{ox} = \frac{2\pi\epsilon_{ox}}{\ln\left(\frac{a}{R}\right)}$$
(3.15)

Thus, the voltage at which pinch-off occurs is:

$$V_{DS,sat} = \frac{qN_DR^2}{2} \left(\frac{1}{2\epsilon_{si}} + \frac{\ln\left(\frac{t_{ox}+R}{R}\right)}{\epsilon_{ox}} \right)$$
(3.16)

Figure 3-12 plots $V_{DS,sat}$ against doping and t_{ox} to examine the trend of when the channel pinches off. We see that for moderately doped silicon nanowires, even with an oxide thickness > 1 μ m, the channel can pinch-off at $V_{DS} < 1V$ due to the logarithmic dependence of $V_{ds,sat}$ on t_ox . This suggests that even without surface states, for a pillar that is surrounded by neutral silicon at ground, the current may be less than what is expected from the intuitive model.

Based on this modeling of individual nanowires, when the silicon nanowire current limiters are in 2-D arrays surrounded by a neutral Si substrate, the current in nanowires at the corner or edge of the array will saturate at a lower drain-source voltage than those in the center of an array. The devices on the edge are at least partially surrounded by Si virtually at ground, whereas devices inside of the array will be surrounded by nanowires that have approximately the same



Figure 3-12: The effect of doping (left) and oxide thickness (right) on the voltage at which the channel pinches off due to the parasitic MOS structure formed by the substrate surrounding a single nanowire.



Figure 3-13: Cross-section of a simulation of a 5x5 array at $V_{DS} = 10$ V. This cross-section shows the electron concentration in the nanowires, and the depletion edge is highlighted in white, defined as electron concentrations less than 10% of the value in thermal equilibrium. This simulation demonstrates the effect of the surrounding substrate on the I-V characteristics of the nanowires.

potential distribution along their length, reducing the transverse electric field that might serve to deplete the nanowire.

To model this effect, we performed fully 3-D device simulations in Sentaurus, a Poisson equation and continuity equation solver, similar to SILVACO, but with better handling of 3-D structures and simulation domains with many nodes. We generated 5x5 and 10x10 arrays of Si nanowires, and swept the drain-source voltage, measuring the current in each individual nanowire. The electron concentrations in the various pillars across a cross-section cut through the center of the 5x5 array at $V_{DS} = 10$ V is shown in Figure 3-13. As expected based on the analytical calculations of individual nanowires above, the pillars at the edge of the array saturate at a lower voltage and at lower currents due to the gate effect of the surrounding neutral semiconductor. The depletion region forms on the side closest to the edge of the array, and extends towards the center of the array as the drain-source voltage in increased.



Figure 3-14: Comparison of I-V Characteristics across a 10x10 Array, demonstrating the effect of the substrate on the current limiter. The current limiters along the periphery saturate at lower currents than those in the center.



Figure 3-15: Comparison of output conductance (a), linear conductance (b), and saturation current (c) for each position in the 10x10 array.

Figure 3-14 shows the individual I-V characteristics of current limiters in a 10x10 array as simulated in Sentaurus. The emitters are all connected in parallel, and their currents measured individually as the drain-to-source voltage is ramped from 0-10 V. While linear conductance does not appear to be sensitive to position in the array, between the edge and the center there can be as much as a factor of 5 variation in saturation current I_{DSS} from this effect, and a factor of 6 difference in output conductance, g_{out} . These results are summarized in Figure 3-15.

It may be possible to mitigate this effect by varying the diameter of the nanowire based on its position in the array, making the nanowires closer to the edge larger than the nanowires in the center, however a remaining challenge is to keep the emitter tip radius uniform while changing the nanowire diameter. To enable uniform emitters with varying nanowire, it may be possible and necessary to decouple the emitter formation from the nanowire formation.

3.6 Thermal Modeling and Analysis

To explore the ultimate performance and failure mode of the Si nanowire current limitters, an analysis of the thermal breakdown of the silicon pillars was performed. Experiments were conducted to verify the maximum possible current sourced before failure of the current limiter. Several single nanowire current limiters were characterized under conditions where current was varied and voltage was measured until thermal runaway and device failure occurred. Figure 3-16 shows a representative voltage-current characteristic, which exemplifies the failure that occurred. I-V characteristics of the device were taken before and after stressing. Before stressing, the device had a saturation current of 8 pA. At a bias current of 11 pA, impact ionization in the high-field region of the channel was observed, resulting in a large change in current with small increase in voltage. At a current level of 200 nA, the device failed and the measured voltage hit compliance, indicating that an open circuit has formed. The inset of Figure 3-16 shows an optical micrograph of the metal contact pad after stressing, showing physical damage resulting from the destructive testing: a large crater in the metallization centered on the location of the single pillar contact.

Following the analysis of the thermal limits of field emitters presented by Utsumi in [122], the maximum current density at burn-out due to Joule heating approximated by:

$$i_{max} = \frac{\sqrt{2T_m \tilde{\sigma} k}}{h} \tag{3.17}$$

where *h* is the height of the structure, T_m is the melting temperature of silicon, $\tilde{\sigma}$ is the average value of electrical conductivity, taken between room temperature and T_m , and *k* is the thermal conductivity of silicon. Using the thermal conductivity and electrical conductivity values for n-type silicon with a donor concentration of 2×10^{14} cm⁻³, a value of 100 nA for I_{max} ($I_{max} = i_{max} \times$ cross-sectional area) is obtained for a 100 nm diameter column, agreeing with the experimental value of 200 nA to within a factor of 2.

To perform more rigorous thermal modeling, we switched to finite element analysis. To model the breakdown and heating in Si nanowires, a hydrodynamic model [118] implemented



Figure 3-16: Schematic drawing of a silicon nanowire surrounded by oxide and a neutral substrate. the substrate/oxide/nanowire structure forms a parasitic, unwanted MOS structure.



Figure 3-17: Simulation of heating and avalanche breakdown in a silicon nanowire using the hydrodynamic models.

in Sentaurus was used. The hydrodynamic model is a set of three energy balance equations that are solved self-consistently with the Poisson equation and the carrier continuity equations in the semiconductor to give impact ionization rates and heating of the semiconductor lattice. The boundary conditions were set so that that the backside of the wafer was a perfect thermal conductor, and the top of the nanowire was a perfect thermal insulator. The sides of the simulation domain were set to be reflecting boundaries (*i.e.* $\nabla T_{\perp} = 0$). The results of the simulation are shown in Figure 3-17. A nanowire diameter of 100 nm, a length of 10 μ m, and doping of 2×10^{15} cm⁻³ were used.

The simulation accurately predicts that breakdown of the device occurs at approximately 2×10^{-7} A, and the model predicts a breakdown voltage of ~ 23 V, compared to the actual breakdown voltage of 19 V. The model has its limitations, however. It does not accurately predict the saturation current of the nanowire. This is expected, as surface states were not included in the simulation model.

3.7 Summary

This chapter presented analytical models, numerical simulations, and measured I-V characteristics of silicon nanowire current limiters. Analytical results and numerical modeling gives insights into the effect of the silicon nanowire surface on current limiter performance and shows that the surface depletion layer that arises from interface traps could have a large effect on device performance. 3-D simulations show that the saturation current depends on the position of the nanowire in the array, and that current limiters towards the edges of the array saturate at lower voltages and lower currents than devices in the center of the array from a gate-effect of the surrounding substrate. Through understanding these effects, methods of minimizing rheir impact on the performance of the field emitter array can be developed.

Chapter 4

Ungated Silicon Field Emitter Arrays with Nanowire Current Limiters

4.1 Introduction

The purpose of this chapter is to present a proof-of-concept structure that demonstrates that a scaled silicon field emitter array with silicon nanowire current limiters is feasible. This device is the first to demonstrate current saturation in scaled silicon field emitter arrays with silicon nanowire current limiters. Due to the distance between the gate electrode and the emitter tip, electrostatic screening limited the emitter-to-emitter to 5 μ m. SEM imaging of the completed structure demonstrated a tip radius < 6 nm with a standard deviation of 1.2 nm. The result is shown to be consistent with both analytical and finite element models.

4.2 Design

During the design process, we performed numerical simulations to guide the design of emitter and extractor gate geometry. DC steady-state simulations were performed using the COMSOL Multiphysics simulation package. In the simulations, a 100-nm diameter, $10-\mu$ tall pillar that is topped with a cone with 15° half-angle. Emitter tip radius was fixed at 5 nm. A diode configuration was employed, where an equipotential surface was placed a distance x away from the emitters with tip-to-tip spacing of y. The emitter-anode separation was varied from 0.1 $\mu m < x < 20 \ \mu m$, and the tip-to-tip separation was varied from 1 $\mu m < y < 20 \ \mu m$. The simulation is a 2-D simulation, thus it is expected that the field factor extracted from this simulation will be slightly lower than would be the case in a full 3-D simulation; however, the simulation still gives valuable insight into the expected screening effects of the ungated array. Figure 4-1 shows the details of the simulation structure.

At the simulation boundary on the right side, a symmetry boundary condition was enforced. This boundary condition sets the normal electric field equal to zero (i.e. $d\Phi/dy = \mathscr{E}_{\perp} = 0$) and the charge at the interface equal to zero (i.e. $\rho_s = 0$). This boundary condition allows for the simulation to cover half the size, and still include appropriate interactions. 5 emitters are simulated, and by measuring the maximum electric field at the emitter furthest from the edge, perimeter effects are much reduced approximating what the field at the center of the array is expected to be. Because the simulation was a two-dimensional (2-D) simulation, the field enhancement at the tip may be somewhat lower than in the 3-D case; however, valuable insight may still be gleaned from these results.

During the simulation, the potential at the top of the simulation area was set to 1 V, and then the highest electric field at the tip of the furthest emitter from the edge of the array. This electric field is the field factor, β . From β , the Fowler-Nordheim slope b_{FN} is extracted using

$$b_{FN} = \frac{0.95 \cdot 6.87 \times 10^7 \cdot \chi_{Si}^{3/2}}{\beta}$$
(4.1)

Where χ_{Si} is the electron affinity of silicon (4.05 eV). b_{FN} has important implications for the performance of field emitters, and a lower b_{FN} implies lower voltage operation. $b_{FN} \approx 5000$ V implies a turn-on voltage, $V_{ON} \sim 200$ V. Figure 4-2 shows the effect of varying the emitter pitch and the tip-anode spacing on b_{FN} .

Examining the field lines shown in Figure 4-1A, due to the prominence of the emitters in the periphery of the array, the curvature of the potential is higher around these emitters resulting in a higher surface electric field.



Figure 4-1: (A) Simulation structure for the exploration of the screening effects of adjacent emitters. The simulation shows the electric field as shaded colors, and the equipotential lines. Units of all of the axes in this figure are in microns. (B) Detail of the emitter tip meshing used in this simulation. (C) Detail of the normalized potential (shaded) surrounding 2 emitter tips in a simulation structure. The anode-emitter spacing is $x = 1 \ \mu m$ and the emitter-to-emitter pitch is $y = 16 \ \mu m$.



Figure 4-2: screening effect of adjacent tips on the Fowler-Nordheim slope, b_{FN} as a function of different tip-anode separations (x in figure 4-1A) and emitter-to-emitter pitches (y in figure 4-1A). A higher b_{FN} implies a smaller field factor and hence a larger effective tip radius and a higher turn-on voltage.

4.3 Fabrication

Based on the design criteria that the device should turn on at a gate-emitter voltage under 200 V, a structure with 5 μ m emitter-to-emitter pitch and a substrate-to-gate separation of ~ 25 μ m (emitter-gate separation of ~ 15 μ m) was chosen. These design choices were a trade-off that allowed for microfabrication of the field emitter array, and hand-assembly of the extraction gate structure using commercially available polymer spacers as standoffs for the extraction gates.

The combined silicon nanowire current limiter-FEA structure was fabricated with 5 μ m pitch and hexagonal packing, and the device was expected to turn on at a gate-emitter voltage of approximately 200 V. Thermal oxide was grown, and photolithography was performed to define "dots" that would define the emitter locations. After the patterning of the oxide with reactive ion etching, the initial formation of the emitter cones is performed using a plasma etch with SF₆/O₂ chemistry. The partial pressure of oxygen changes the lateral etch rate, while barely changing vertical etch rate, allowing for control over the etch profile.

The undercutting step needed to be precisely controlled to obtain a sharp tip after the oxidation to reduce the pillar diameter. After the tip formation, the pillar is etched using the



Figure 4-3: Schematic cross-section of the fabrication steps for building ungated field emitter arrays with integrated nanowire current limiters.

same DRIE process as in the vertical current limiter process. Following the DRIE step, the structure was oxidized at 1000°C using dry O_2 to simultaneously form sharp emitter tips while reducing the pillar diameter. Finally, the thermally grown oxide is removed using 10:1 diluted HF. An SEM image of the completed structure is shown in Figure 4-4. The pillars are 10 μ m tall, with a diameter of 110 nm and a tip radius under 10 nm. Figure 4-6 shows representative SEM images of the emitter tips after field emission testing. Figure 4-5 is a plot of the radii of 209 tips and Figure 4-6 shows some representative SEM images of the emitter tips. The statistics follow a log-normal distribution with a mean of 5.6 nm and a standard deviation of 1.3 nm. The tip radius measurements were made using SEMs taken after the device characterization; hence they are representative of the actual device dimensions.

The full fabrication process flow for the fabrication used for these devices is included in Appendix H.



Figure 4-4: Scanning electron micrograph of the completed ungated field emitter array with integrated silicon nanowire current limiters. The device has 5 μ m pitch, and the current limiters are 100 nm diameter and 10 μ m tall.



Figure 4-5: Tip Radius statistics for 209 emitters measured with SEM. The distribution is log-normal with a mean of 5.6 nm and a standard deviation of 1.3 nm.



Figure 4-6: Representative SEM images of emitter tips used to generate the statistics shown in Figure 4-5



Figure 4-7: Schematic of the triode configuration used to test the ballasted FEAs. A nylon polymer gasket acts as a stand-off between the FEA and the extraction gate.

4.4 Current-Voltage Characterization

Arrays of 1.36 million emitters were fabricated following the process described in Section 4.3 using n-type Si wafers with a nominal donor concentration of $\simeq 10^{15}$ cm⁻³. The experimental setup is shown in Figure 4-7. A 25 ± 10 µm thick nylon spacer was used to insulate the emitters from an unaligned perforated extraction grid, resulting in a tip-grid distance of 5–25 µm. When electrons are emitted from the sample, a fraction of the electrons pass through the grid and are collected by a suspended ball-shaped anode biased at +1100 V, allowing us to determine whether the source of the current is leakage through the dielectric spacer ocesr electron field emission current. The linear relationship between the current collected at the grid and the suspended anode suggests that the origin was field emission. At higher voltages, this linearity does not hold. It is likely that the polymer spacer is beginning to break down at these voltages, resulting in significant leakage current in addition to the field-emission current.

I-V characterization was conducted in an ultra-high vacuum testing chamber at a pressure of 9×10^{-10} Torr using three commercially available high voltage source measure units (Keithley model 237 SMUs). Figure 4-8 shows the typical I-V characteristics of a 1.36-million tip FEA for electrons collected by both the extraction gate and the anode. The anode efficiency, *i.e.* the transparency of the extraction gate was about 1 %. The Fowler-Nordheim (FN) plot, also shown in Figure 4-8, is a common technique for plotting field emission data. Ideally, there is a

linear relationship between the natural log of the current over the square of the gate voltage and the inverse of the gate voltage [46]. At high extraction gate voltages, if the emission current is limited by the vertical current limiters, the FN plot should deviate from the linear relationship and the slope should become less negative at higher voltages. At a bias voltage of approximately 600 V, the FN characteristic in Figure 4-8 begins to deviate from its linear characteristics.

The tip radius may be estimated empirically from I-V characteristics by extracting the slope of the Fowler-Nordheim curve, b_{FN} and using

$$\beta \approx \frac{0.95 \cdot 6.87 \times 10^7 \cdot \phi^{3/2}}{b_{FN}}$$
(4.2)

and

$$r \approx \frac{1}{\beta} \tag{4.3}$$

where β is the field factor in cm⁻¹ ϕ is the work function barrier in eV, assumed to be 4.05 eV for n-type Si, and r is the emitter tip radius in cm. The slope extracted from the F-N gate current characteristic shown in Figure 4-8 was 2942, resulting in a β of 1.80×10^5 cm⁻¹, and an r of 55 nm. This result is not consistent with the experimental measured tip radii shown in Figure 4-5, and will be examined further below.

4.5 Discussion

The current-voltage characterization of the individually ballasted FEAs presented in the preceding section are consistent with the results obtained by Velásquez-Garcá et al. [125] and the characterization of the vertical current limiter pillars without field emission structures reported in Chapter 3. Using the data presented in Table 1 and assuming that the emission current is limited to 10 pA/emitter the expected array current is 13.6 μ A. From Figure 4-8, at a gate-tosource bias voltage of 550 V, we observe the transition of the F-N plot from the regime where the current is limited by the transmission of electrons through the barrier to a supply limited regime. The corresponding emission from the FEA is 11.1 μ A. This result is consistent with our suggestion that the vertical current limiter controls emission current.



Figure 4-8: (top)I-V Characteristics for an array of 1.36 million individually ballasted field emitters. At gate voltages over 550 V, the FEA enters a regime where the current is limited by the supply of electrons to the field emitters by the vertical current limiter, rather than the transmission through the barrier. (bottom) F-N Plot for an array of 1.36 million individually ballasted field emitters, demonstrating the different regions of operation.

We note that the surface of these current limiters are covered by thin oxides that are not carefully processed and hence may be effected by a higher interface trap density. This increased trap density will have an impact on the saturation current of the current limiters however, based on the experimentally obtained saturation characteristics of the FEAs reported above, we do not believe this had a major impact on the ability to use this structure for the current limitation of field emission arrays.

An explanation for this difference in extracted tip radius is that the models that describe the electrostatics of a microfabricated field emitter with a self-aligned gate that is very close to the tip [123, 37] do not translate well to structures where the gate is perforated and located about 25 microns away with tip-to-tip separation of 5 μ m and an aspect ratio of 100, as we have in this case. Screening of the tip electrostatic field by proximal tips has been suggested in the literature, especially when the tip-to-tip distance is far less than the tip-to-gate distance [96]. A 3-D finite element model of the emitter with the gate electrode positioned 12 μ m above the emitter tips, shows that the field factor, β is about an order of magnitude lower than predicted by the ball-in-sphere model for an emitter with the same radius. The model accounts for screening by nearest neighbor emitters on the electric field at the apex of the tip. A comparison of the field factors for the ball-in-sphere model and the simulation results are shown in Figure 4-9. Using the field factor measured in the I-V characteristics (1.8×10^5 cm⁻¹), from this model we estimate the tip radius to be ≈ 6 nm, agreeing very well with the tip radius statistics measured using SEM (mean = 5.6 nm).

Another possible explanation of this difference could be the presence of a thin native oxide could manifest itself as a lower field factor, which translates into an implied higher workfunction. However, assuming a work function increase from 4.05 eV to 4.5 eV changes the extracted tip radius by less than 20 %. Hence, this is not very likely.

The current saturates at $\simeq 10 \ \mu$ A, corresponding to $\simeq 7.36 \text{ pA/emitter}$. This number agrees very well with the measured saturation current of individual silicon nanowires presented in Chapter 3, 8.5 pA. Because there is no thermal SiO₂ to passivate the surface of these nanowires, we expect that the density of interface traps to be large, $D_{it} > 10^{11} \text{ cm}^{-2} \text{ eV}^{-1}$. Combinded with the small diameter and the low to moderate doping ($\simeq 10^{15} \text{ cm}^{-3}$), it is likely that the Si nanowire are fully depleted. Assuming $n_{eff} \approx n_i = 10^{10} \text{ cm}^{-3}$ and a diameter of 100 nm, we



Figure 4-9: 3-D finite element simulation of the ungated emitter structure taking into account screening of the electric field from nearest neighbors. emitter pitch is 5 μ m, the gate is 12 μ m above the emitters.

find that the saturation current is 1.7 pA/emitter. This discrepancy could be explained by either the pillar diameter being slightly larger than 100 nm, or that the effective carrier concentration is slightly higher.

4.6 Summary

This chapter demonstrated the design, fabrication, and characterization of ungated Si field emitter arrays with Si nanowire current limiters. We verified that the FEAs function as intended, and successfully limited the current, as evidenced by a "bending over" of the F-N plot. The measured saturation current was consistent with the measured I-V characteristics of the nanowire current limiters, and the tip radius statistics measured with SEM were consistent with the tip radius extracted from the Fowler-Nordheim plot of the data, using a finite element model for the field factor, β .

Chapter 5

Gated Silicon Field Emitter Arrays

5.1 Introduction

This chapter presents gated field emitter arrays with silicon nanowire current limiters. First, it explores in detail several key process steps method for forming silicon nanowires and filling the gap between nanowires with a conformal dielectric. This conformal dielectric is required to support an integrated gate with a gate aperture that is self-aligned to the emitter. However, this first structure suffered from several significant drawbacks including large gate leakage. To combat these problems, a new device architecture was developed that addressed the problems and the fundamental problems of FEAs. Through addressing these challenges, the new device design attained breakthrough device performance with regards to current density and device lifetime.

5.2 Numerical Modeling

A simple silicon FEA with vertical nanowire current limiter fabrication process was created and simulated in ATHENA, resulting in the structure shown in Figure 5-1. In this simulation the donor concentration was 2×10^{15} cm⁻³. To create a sharp tip ($r \approx 2$ nm) on top of the silicon pillar, a short isotropic etch to begin rough tip formation was performed continuing with the deep anisotropic etch to form the pillar. When performing the oxidation, the time was carefully controlled to produce a sharp emitter. After the oxidation, the gaps were filled in with LCPVD oxide, Si_3N_x , and polysilicon was deposited. To create a self-aligned gate, a polishing step was used to pattern the gate aperture, stopping just before the emitter tip was damaged. Finally, the oxide encasing the emitter was removed with a wet etch step.

The structure created in ATHENA was then imported into DevEdit, software which allows for arbitrary editing of structure files generated by SILVACO tools. The meshing around the tip was made much finer to be able to simulate the high gradient at the emitter surface required to model field emitters. The grid constraints in the rest of the simulation domain were relaxed to keep the number of grid points manageable and allow subsequent device simulation in ATLAS to converge.

ATLAS provides a framework for modeling Fowler-Nordheim tunneling through gate oxides that solves the tunneling current at the semiconductor-dielectric interface directly with the current continuity equations, setting a current boundary condition at the emitter tip surface. This model is typically used by CMOS designers for modeling tunneling into the floating gate in Flash memory. The model uses a version of the Fowler-Nordheim equation to calculate the current density at the tunneling interface:

$$J_{FN} = A_{FN} E^2 \exp\left(-\frac{B_{FN}}{E}\right)$$
(5.1)

E is the magnitude of the electric field in the dielectric. By default, A_{FN} and B_{FN} are Fowler-Nordheim coefficients for tunneling through a trapezoidal oxide barrier, but can be easily modified in the code to use appropriate F-N parameters for tunneling into vacuum that include image charge effects i.e. $A_{FN} = 4.752 \times 10^{-5}$ and $B_{FN} = 5.268 \times 10^{8}$.

The simulated I-V Characteristics and F-N plot are shown in Figure 5-2 and summarized in Table 5.1. The emitter reached a current of 1 pA at ≈ 28 V, and the onset of saturation is at a current of 1.5 μ A @ $V_{GE} \approx 70$ V. From the F-N plot in Figure 5-2, we find that the slope is 600, corresponding to a β of $\approx 8.8 \times 10^5$ and a tip radius of 11 nm using $\beta \approx 1/r$. The effective tip radius for this simulation was limited by the meshing at the apex of the emitter.



ATLAS Data from deckbY1NJSz



Figure 5-1: The structure of the numerical simulation

Parameter	Value
Doping Density, N_D (cm ⁻³)	2×10^{15}
NW Diameter (nm)	200
Turn-on voltage (V)	20 V
Gate-Emitter Saturation Voltage (V)	70 V
Saturation Current (A)	1.5 μA
Output conductance (S)	8×10^{-9}
Fowler-Nordheim Slope (b_{FN})	602
Fowler-Nordheim Intercept (a_{FN})	-11.24
β (cm ⁻¹)	8.79×10^{5}
Tip radius (nm)	11

Table 5.1: Extracted parameters from the device simulation in ATLAS



Figure 5-2: Simulated IV and FN Characteristics for a device with a 200-nm diameter, 8 μ m tall nanowire current limiter, and doping density $N_D = 2 \times 10^{15}$ cm⁻³.

5.3 Process Development and Proof-of-Concept Devices

5.3.1 Process Development

The general fabrication process for the gated structure is shown in Figure 5-3. The process begins identically to the ungated device presented in Chapter 4 with patterning an oxide hard mask followed by reactive ion etching to form the emitter tips and deep reactive ion etching to form the silicon nanowire current limiters. The tips are sharpened and the SiNW diameters is reduced to ~ 100 nm with thermal oxidation at 950 °C. After the oxidation, a multistep fill process is used to fill in the gaps between adjacent pillars with SiO₂ and Si₃N_x. The Si₃N_x is



Figure 5-3: Schematic of the process for gated FEAs.

then etched back to expose the oxide domes, which will form the self-aligned gate apertures. An *in situ*-doped poly-Si gate is then deposited, and thinned using CMP to form the gate apertures. Finally, a wet etch in buffered-oxide etchant (7:1 HF:NH₄F) is used to release the emitters.

To successfully build these FEAs, three key processing steps needed to be well-understood and optimized those processing steps were:

1. Deep reactive ion etching to form the Si nanowire current limiter



Figure 5-4: Schematic of the Bosch process used in DRIE.

- 2. Filling in the gaps between adjacent pillars
- 3. Chemical mechanical polishing to etch back the silicon nitride used to fill in the gaps between pillars.

The following sections will go into each of these steps in detail.

Deep-Reactive Ion Etching

Deep reactive ion etching (DRIE) is a process that was developed for the MEMS industry for the creation of high-aspect-ratio structures. It is also widely used in the semiconductor industry to fabricate deep trenches for capacitors used in dynamic memory (DRAM) [109], and though-silicon vias (TSVs) [35]. DRIE uses the Bosch process to achieve extremely anisotropic features and vertical sidewalls. The Bosch process is a time-multiplexed process where the etcher switches an inductively coupled plasma (ICP) between two modes to achieve vertical sidewalls:

- 1. An isotropic SF_6 plasma etching step in which ions are directed vertically towards the substrate, and free fluorine excited radicals react with the silicon on the surface to form volatile SiF_2 which is then removed
- 2. A passivation step, in which C_4F_8 which decomposes to CF_x reacts with the exposed silicon surface to form a polymer substance similar to Teflon.

Etch Parameter	Value
Bosch Process Gasses	SF6 and C4F8
Pressure (mTorr)	24
Etch Cycle Time (s)	5 (start on etch cycle)
Passivation Cycle Time (s)	5.8-7.5
SF6 Flow Rate (sccm)	80-110
C4F8 Flow Rate (sccm)	45
Etch/passivation Power (W)	800/600 W
Platen Power during Etch/Passivation (W)	120/60

Table 5.2: Etch Parameters

The passivation layer protects the sidewalls from further etching, but the polymer at the bottom of the feature being etched is sputtered away by energetic, directional ions. The process is shown schematically in Figure 5-4.

A challenge of using DRIE for high-aspect-ratio structures is that the etch profile is geometry dependent, and very sensitive to processing conditions. Hence, for challenging (i.e. high-aspect-ratio, high density structures) careful process development and optimization are absolutely critical to attain uniform and repeatable results. If a non-optimized process is used, small changes in the starting mask or processing conditions can result in poor etches with problems such as significant undercut, mask erosion, or black silicon formation. Examples of problems that resulted from non-optimized DRIE processes are shown in Figure 5-5.

To optimize the DRIE etch in order to improve the profile of the silicon nanowire current limiters, we conducted a 2-parameter design of experiments. A series of initial experiments to explore the parameter space and determine which parameters had the biggest impact on the etch were conducted. The parameters that were explored include RF power, platen power, pressure, and gas flow rates. After the exploration phase, the SF₆ flow rate and the ratio of etching time to passivation time were found to have the largest effect on on etch selectivity. Here, we define etch selectivity as the ratio of the vertical etch rate to the lateral etch rate. The parameters for the etch are summarized below in Table 5.2.

The SF_6 flow rate and passivation time were varied using a circumscribed central composite experimental design, which is a box circumscribed by "star points" and a center point that is measured multiple times to demonstrate repeatability and lend confidence to the values mea-



Figure 5-5: Some of the challenges of DRIE.



Figure 5-6: Exploration space for the design of experiments

Table 5.3: Summary of etching conditions used in the circumscribed central composite design of experiments

sured. The experimental design space is shown in Figure 5-6 and Table 5.3. A consequence of using a central composite design of experiments is that the response surface modeling is limited to linear and quadratic terms, and cannot model higher order interactions between the variables. To model these higher order terms, a full factorial design, with many more data points would need to be implemented. These interactions can be safely neglected for our purposes.

From the experiments, a response surface was fitted to the data in MATLAB using RSTOOL, an interactive response surface modeling tool included in the statistics and machine learning toolbox of MATLAB. This tool allows for the mapping of linear, quadratic and higher order response surfaces to an experimental design. In this case, a quadratic model with no interactions was used. The results are summarized in Figure 5-7.

In general, both the lateral etch rate and the vertical etch rate increase with increasing SF6 flow rate and decreasing passivation time. This makes intuitive sense. With a shorter passivation time, a higher porportion of the total time is spent etching, thus the etch rate is expected to be higher. In addition, a shorter passivation time implies that there will be less passivation polymer to etch through, and thus the etch rate will be faster.



Figure 5-7: (A) the response of vertical etch rate to SF6 flow rate and Passivation cycle time (B) response of lateral etch rate to the same (C) Selectivity demonstrating the shift from the original etch [denoted by black square] to the optimized etch [denoted by white circle] (D) Confirmation of etch selectivity.
A higher SF₆ flow rate means that the residual C_4F_8 will be flushed from the chamber faster, and there will be a higher concentration of SF₆ in the chamber. In addition, reacted products will have a shorter dwell time in the chamber because while the pressure remains constant, the flow into the vacuum chamber increases, which implies that the flow out of the vacuum chamber must increase by the same amount.

Because the lateral etch rate and the vertical etch rate do not have the same dependence on these parameters, a local selectivity maximum occurs at a flow rate of \approx 90 sccm and a passivation time of 6.6 s.

5.3.2 Dielectric Fill Process

Developing a process to fill in the gaps between the SiNWs was a significant fabrication and materials challenge. A variety of different techniques were employed before a feasible combination of techniques was developed.

The first attempts to fill in the gaps used a liquid spin-on-glass (SOG). SOG is hydrogen silsesquioxane (HSQ) or methylsiloxane-based material where the wafers are spun-coated with SOG, and then cured at $\approx 450^{\circ}$ C. During the curing process, solvent is removed from the film and the liquid becomes a glass. SOG is commonly used as an inter-level dielectric and to planarize topology, however, even low-shrinkage SOG materials were found to result in cracking in our structure.

Low temperature oxide (LTO) deposition was also used to fill the gaps. LTO deposition is an LPCVD reaction using SiH₄ and O₂ at $\approx 425^{\circ}$ C. After deposition, the oxide needs to be densified at a higher temperature (> 800°C to remove hydrogen from the film and to improve the dielectric qualities of the oxide. LTO deposition has several drawbacks that makes the process unsuitable for this application. The low-temperature of the deposition causes the deposition to be not very conformal, forming large"keyhole voids" where the top of the gaps close before the bottom of the gap is filled. In addition, the high temperature densification causes the oxide film to shrink 1 – 2 % and cracking results.

Undoped poly-Si deposition and oxidation was tried to fill the gaps between SiNWs. We found when using only poly-Si deposition and oxidation, the deposition was unable to be controlled enough. It was impossible to attain simultaneously filled gaps and complete poly-Si consumption during the oxidation process.

After this series of experiments, a process consisting of Si oxidation, poly-Si deposition, poly-Si oxidation, and Si_3N_x deposition was developed to create a conformal dielectric fill. The multi-stop process is shown in Figure 5-8.

This stack consists of silicon dioxide and silicon nitride formed through a combination of deposition of undoped poly-silicon films, oxidation of the poly-silicon films, and silicon nitride deposition. The undoped poly-silicon is deposited at 625 °C using low-pressure chemical vapor deposition (LPCVD). At this temperature, the polysilicon shows columnar growth and large grains [73] . In addition, due to the elevated temperature at which the deposition takes place, the deposition is reaction-rate limited [72] and the poly-silicon has high surface diffusion [72], allowing for conformal deposition into the deep voids between nanowires. The thickness of this deposition is controlled so that after subsequent oxidation, a gap of < 50 nm remains between adjacent nanowires, and preventing the oxidation fronts from meeting. If the oxidation fronts do meet, they form keyhole voids or leave unoxidized polysilicon.

The remaining void is filled by 1- μ m of low-stress, silicon-rich silicon nitride deposited in a vertical tube reactor via LPCVD at 800 °C. The deposition reaction to form SiNx is also ratelimited and has high surface diffusion so that the nitride fills in the remaining gaps between pillars. Subsequent cross-section shows that this process creates a reproducible, consistently



Figure 5-8: Process for filling in the voids between adjacent silicon nanowires with a dielectric matrix.

void-free dielectric matrix that completely surrounds the nanowires. A cross-section of the void free fill after planarization is shown in Figure 5-9.



Figure 5-9: Void-free dielectric matrix with vertical silicon nanowires embedded in them after CMP planarization.

Chemical-Mechanical Polishing (CMP)

?? Chemical mechanical polishing (CMP) is a process technology for creating smooth surfaces on semiconductor wafers though a combination of abrasive polishing and wet chemical etching. A typical CMP tool, such as the one shown schematically in Figure 5-10 uses a large, pliable polishing pad attached to a solid metal platen. The wafer is attached to a chuck, held in place with a retaining ring and vacuum, and is placed in intimate contact with the pad. The pad deforms to conform to the wafer surface, to account for wafer bow and long-range "waviness."



Figure 5-10: Schematic of a CMP process

A slurry, in this case a colloidal suspension of fine silica particles (mean size $\approx 0.1 - 0.2 \ \mu m$ [83]) in deionized (DI) water and potassium hydroxide (KOH) is used. The silica particles are the abrasive, and the KOH enhances the polishing rate by causing silica dissolution [33]. The solution is alkaline, with a pH ≈ 10 .

Chemical mechanical polishing is widely used by the semiconductor industry. Originally, the process was only used in back end-of-line (BEOL) processing for the planarization of metal layers and inter-level dielectrics (ILDs) [75], and the damascene process for Cu interconnects [128]. Now, CMP is widely used in shallow-trench isolation (STI) [94] and to pattern metal gates for transistors in advanced technology nodes.[63].

CMP is an important processing step for building field emitter arrays with self-aligned gates as explained in Chapter 2. The preferred method of forming self aligned gates in Si FEAs is to use an oxide deposition to define the aperture of the device, deposit the gate material, and then polish the structure, timing the polishing step so that it removes the gate material to expose the oxide that embeds the emitter tip but before the tip is exposed. This process takes advantage of the fact that "bumps" polish faster than planar surfaces, however if the polishing step goes too long, the tip may be damaged, resulting in blunt tips.

The process for fabricating self-aligned gated FEAs with vertical Si nanowire current limiters



Figure 5-11: Wafer thickness maps for three different times to extract the etch rate and uniformity of the polishing. 49 points across the wafer were measured using spectroreflectometry.

utilizes two CMP steps (three with the inclusion of the mesa structure). Polishing the silicon nitride is critical to both planarize the film and to thin it so that the nitride/oxide interface is within ≈ 200 nm of the surface. This step is required to achieve uniformity during the wet etchback to expose the oxide domes. A thick nitride is required to fill in the gaps, however, when wet etching thick films spatial non-unformities become worse. Silicon nitride is a very hard material (hardness ~ 30500 MPa) and etches extremely slow or not at all in most chemical etchants. This section will only go into depth into the process of CMP of silicon nitride. The CMP of polysilicon for the gate apertures is also an important step that was well characterized.

To characterize the polishing rate of the silicon nitride, a number of bare silicon wafers were coated with nominally 1- μ m of low-stress silicon-rich silicon nitride deposited using a lowpressure chemical vapor deposition (LPCVD) in a vertical tube reactor. These wafers were then polished for different times, and then 49 measurements of the film thickness across the wafer were taken using spectroreflectometry (DBS) using a UV1280 spectrophotometer/ellipsometer. DBS uses a spectroreflectometer that exposes a wafer to light (220-830 nm) and determines the film thickness or refractive index based on the reflected spectrum (reflected intensity as a function of wavelength) and a known standard (a clean bare silicon wafer) [1]. The wafer maps for several different polishing times are shown in Figure 5-11.

From the wafer maps of the film thickness, the thickness of the nitride has a clear spatial dependence, with the nitride on the outside edge of the wafer etching faster than the center of the wafer. This distribution can be explained by the fact that the wafer is spinning, and hence the edge of the wafer is moving faster across the surface of the pad than the center of the wafer. This non-uniformity is mitigated somewhat by the fact that the platen and the wafer chuck co-rotate and the wafer chuck oscillates back and forth on the platen, however it cannot be eliminated.

From the slope of the plot of film thickness vs. polishing time, we can extract the polishing rate of silicon nitride. The etch rate of the nitride is linear, as shown in Figure 5-12 and ≈ 24 nm/min. Compared to the etch rate of SiN_x in 165 °C Phosphoric acid – the typical method of etching silicon nitride films – 4 nm/min.

The plot of nitride thickness was created over multiple days starting with a bare silicon wafer coated with 1 μ m of nitride each time, so the CMP tool is exhibiting very good uniformity and



Figure 5-12: Extraction of nitride polishing rate from thickness measurements. The error bars show the standard deviation of the measurments made on a single wafer, showing the spatial variation in film thickness.

stability over the short term. The time point at 1800 s was repeated over 2 different days with good agreement to verify polishing stability. Over long periods of time, however, the polishing rate has drifted by as much as a factor of 5, so care must be taken to measure the actual polishing rate prior to processing important device wafers.

Using these results, we then polished wafers with silicon field emitter tips on top of nanowire current limiters. Due to the topology of the nitride film, the polishing rate was faster initially, but then slowed once a planar surface was formed. Figure 5-13 shows the emitter tips buried in SiO_2 and SiN_x , and Figure 5-14 shows the resulting oxide domes after the etchback in hot phosphoric acid.

5.3.3 I-V Characterization

Figure 5-15 shows the testing setup used to characterize these devices. Due to the lack of probe pads, contact needed to be made directly to the gate over the active area of the field emitter array. Two different methods of making contact were used, a sharp tungsten probe and a stainless steel ball electrode. The tungsten probe would short the emitter to the gate, either because it would punch through the thin polysilicon to the gates, or it would make



Figure 5-13: Cross-section SEM of a device after nitride polishing. There is about 200 nm of nitride remaining on top of the oxide bumps.



Figure 5-14: SEM image of the oxide domes after nitride etchback in 160 C phosphoric acid.

contact to the emitter tip by lodging itself into gate apertures. Thus, the ball electrode was the better method of making contact to the gate. While the large surface area of the ball prevented shorting between the gate and the emitter tips, a sizable percentage of the array was occluded by the ball and the ball deformed the electric field over the FEA, resulting in a high fraction of the current being intercepted by the ball itself. The I-V characteristics and Fowler-Nordheim plot of a representative device are shown in Figure 5-16.

The field emitter array turns-on at a gate-emitter voltage of 20 V, and exhibits a Fowler-Nordheim slope of < 300. If we approximate the field factor as $\beta \approx 1/r$, this implies that the tip radius is 5.2 nm. However, this approach typically overestimates the tip radius of sharp emitters (see Section 5.4.4), so the average tip radius is likely sharper than 5 nm.

At gate-emitter voltages of 35 V, the Folwer-Nordheim characteristic deviates from its linear regime, suggesting that the nanowire current limiters are controlling the supply of electrons to the emitter surface. The array consists of 10^6 emitters, and the emitter current, $I_E = I_G + I_A \approx 1 \ \mu$ A, implying a current/tip of approximately 1pA, consistent with earlier results for the saturation current of 100-nm diameter silicon nanowires and the ungated devices.



Figure 5-15: Testing setup of the devices



Figure 5-16: I-V characteristics and Fowler-Nordheim plot of the proof of concept gated field emitter array. The device turns on at 20 V demonstrates current saturation at approximately 35 V.

5.4 High Performance Silicon Field Emitter Arrays

5.4.1 Design

The devices presented in the previous section demonstrated a method to fabricate silicon field emitter arrays with silicon nanowire current limiters and self-aligned gate apertures. However, the high gate leakage current of these devices means that they are far from optimized. To prevent the high gate current, a better method of isolating the gate from the substrate needed to be developed to prevent leakage. Furthermore, this isolation needs to be able to withstand a high voltage without breakdown. High quality SiO₂ has a dielectric strength (the electric field at which breakdown of the dielectric material occurs) of approximately 10 MV/cm, however, when the dielectric is biased near its dielectric strength over time, charge injection into the dielectric can cause failure. This type of dielectric failure is called time-dependent dielectric failure (TDDB) [87]. In order to prevent TDDB and ensure long lifetime of the dielectric, the dielectric must be de-rated, and the electric field must be limited to a fraction of its dielectric strength. Usually, limiting the field applied to about 1/10 of the dielectric strength. Thus to support ≈ 200 V on the gate, the dielectric spacer between the gate and the substrate needs to be at least 2 μ m.

Based on previous results, and in order to attain high current per tip, the Si nanowire must



Figure 5-17: A 3-D rendering of the proposed device structure.

not be fully depleted. This implies that the nanowire must have a larger diameter, high quality dielectric passivation, and higher doping than the field emission devices previously discussed. We chose to increase the diameter to 200 nm, and use <100> 150 mm n-type phosphorus doped wafers with resistivity $\approx 3 \ \Omega \cdot cm$ (Doping density $N_D = 2 \times 10^{15} \ cm^{-3}$). A 3-D rendering of the proposed device structure is shown in Figure 5-17.

Assuming that the pillar diameter and doping are high enough that surface states and perimeter effects can be neglected, our simple analytical model predicts that the maximum current/pillar to be limited to $I \approx q A_{eff} N_D v_{e,sat} = 4 \ \mu A/tip$.



Figure 5-18: The full process for FEAs with integrated SiNW Current Limiters and self-aligned gates.

5.4.2 Fabrication

Arrays sizes ranged from a single emitter to 1000×1000 emitters. Each array has 1- μ m emitterto-emitter spacing. The emitters have tip radius < 10 nm and are on top of a silicon nanowire that is < 200 nm in diameter and 8 microns tall. A doped polysilicon gate electrode with 350nm diameter surrounds the emitter for applying the electric field to the field emitter tip. Figure 5-18 shows an outline of the fabrication steps.

First, a mesa region on which the array of silicon nanowires capped with field emitter tips will be later be fabricated is defined (Figure 5-19) by etching trenches around the array region. The trench is at least 200 μ m-wide and ~ 2 μ m-deep.





Active Device Regon

Figure 5-19: Etching the mesa that will form the active device region.

 $5-\mu$ m of plasma enhanced chemical vapor deposited (PECVD) oxide is deposited on both the front and the back of the wafer to fill in the trenches that were formed with dielectric. In order to manage stress and prevent bowing, the deposition was carried out in two steps, with an anneal step in between. Between depositions and after the second deposition, the wafers were annealed at 900 °C in O₂ ambient. Chemical Mechanical Polishing (CMP) is then used to planarize the top surface and expose the bare silicon in the mesa region (Figure 5-20).

A corresponding amount of the oxide film on the backside of the wafer was removed to balance the stress in the oxide films and remove the wafer bow that was resulted after CMP. After planarization, approximately $2-\mu m$ of SiO₂ remains only in the trenches around the active regions, and it isolates the gate electrode pad from the substrate and prevents dielectric breakdown of the oxide in the vertical direction from occurring during normal device operation.

Next, 50 nm of thermal oxide is grown followed by the deposition of 250-nm of SiO_2 by PECVD. The oxide layers are an etch mask for patterning the emitters. The patterning and



Figure 5-20: filling the trenches surrounding the active device region and planarization.



Figure 5-21: Rough emitter tip etch.

etching process to form the nanoscale-sharp silicon tip and the silicon nanowire current limiter begins with an i-line stepper photolithography using a positive photoresist (Shipley SPR-700) to form 1- μ m pitch arrays of 0.5 μ m photoresist dots. It is critical that the array patterns be well-aligned (misalignment < 50nm) to the mesa regions. The oxide hard mask is patterned by a CF₄/CHF₃/Ar reactive ion etch. After this etch, the tip has a diameter of about 200 nm. An SF₆ plasma is used to etch the silicon isotropically. The result of this etch, shown in Figure 5-21, forms what will become the field emitter tips.

Deep reactive ion etching (DRIE) is used to form the high-aspect-ratio silicon pillars. The pillars after the DRIE step with the SiO_2 hard mask removed are shown in Figure 5-22. The DRIE etch uses the optimized etch parameters as developed in the previous section.

At this point, the pillars have a diameter of 400 nm, and are 10 microns tall. The diameter is somewhat less than the diameter of the hardmask used due to undercutting of the oxide during





Figure 5-22: Silicon nanowire etch

the DRIE step because the etch is not perfectly anisotropic.

The remaining photoresist and the hard mask are stripped in an O_2 plasma and 7:1 BHF, respectively. To reduce the dimensions of both the pillar and the tip to their final dimensions, the wafer underwent wet oxidation at 950 °C. 950 °C is the temperature above which viscoelastic flow of silicon dioxide occurs, which would deform the shape of the silicon dioxide and potentially result in blunt silicon tips. [61, 86, 60, 39].

A 2-D finite element semiconductor process simulator (Silvaco ATHENA) was used to design the etching and oxidation steps to ensure that the emitter tip would be nano-scale sharp and that the silicon nanowire would be < 200 nm diameter after this oxidation [55, 54]. The silicon nanowire has a diameter of \approx 200 nm and the column height is \approx 10 μ m. The silicon tip is formed by oxidation sharpening resulting in a tip radius of \approx 6 – 8 nm [55].

The deposition of the dielectric stack is, described in detail in Section 5.3.2, resulting in completely filled gaps between Si nanowires as shown in Figure 5-23. A self-aligned polysilicon gate is then defined. To define the gate aperture, first a timed polishing of the nitride is performed using CMP to bring the surface of the nitride to within approximately 100 nm of the top of the oxide domes.

The oxide domes are exposed as shown in Fig. 5-24 through a timed wet etch using phosphoric acid at 160 °C (etch rate ~ 40 Å/min). Phosphoric acid etches silicon nitride with a



Figure 5-23: Void-free dielectric matrix with vertical silicon nanowires embedded in them.

selectivity of ~ 20 : 1 relative to silicon dioxide. The oxide domes will assist in the formation of the gate apertures by allowing the gate material deposited on the remaining nitride to survive, CMP, but the material deposited on the domes to be removed.





Figure 5-24: Oxide domes that form the gate aperture.

800 nm of poly-silicon gate electrode material that is in situ doped with phosphorus is next deposited as shown in Figure 5-25. A thickness of at least $3\times$ the height of the oxide dome, so that there is full coverage of the gaps between the domes, and that the surface topology is smooth. After polishing, the dopants are activated using rapid thermal annealing (RTA) at 950 C for 30 s.

A third CMP step planarizes the gate, and is timed to stop within 50nm of the emitter tip. The grown and deposited oxide around the emitter tip defines the gate aperture, and the depth of the silicon nitride wet etch determines the gate thickness. Similar to the character-



Figure 5-25: Cartoon and SEM of the gate polysilicon deposition

ization of the nitride polishing shown in section ??, characterization of the polishing rate of the polysilicon was performed using spectroreflectometry and confirmed with 4 point probe electrical measurements. Figure 5-26 shows the results of the characterization of the polishing rate of polysilicon. the rate was approximately 47 A/s, and both the optical and the electrical characterization are in agreement. Figure 5-27 shows the gate apertures after CMP.

After gate apertures are formed, the poly silicon gate is patterned with photolithography and a dry etch process. Ni/Ti/Au contact metal is deposited and patterned using a lift-off tech-



Thickness Measurements from Ellipsometer and 4pt Probe

Figure 5-26: Characterization of the polysilicon polishing rate.





Figure 5-27: The gate apertures after CMP.

nique. The metal stack is sintered at 400 °C under forming gas for 30 minutes. During the sintering process, the nickel reacts with the polysilicon to form a nickel silicide, ensuring low contact resistance. Finally, a commercial pad etchant (Silox Vapox III, Transene Co., Danvers, MA) removes the oxide encasing the tips to expose the tips. The sample immediately is then dried with N_2 and loaded into the ultra-high-vacuum characterization chamber for I-V characterization. A tilted SEM image of the completed and released device is shown in Figure 5-28, and a cross section is shown in Figure 5-29.

As can be seen from the the SEM images, the silicon nanowires are 8 μ m tall and < 200 nm in diameter, with the characteristic scalloping from the Bosch process. The emitter tips have a radius < 10 nm and the gate aperture diameter is ≈ 350 nm. The oxide outside of the array is 2 μ m thick, and the polysilicon gate is ≈ 200 nm thick, with the emitter tip recessed from the top by ≈ 50 nm. Due to the sidewall slope of the mesa etch, there is a sharp, $\approx 1 \ \mu$ m tall ridge around the perimeter of the array, located $\approx 1 \ \mu$ m from the poly-Si/SiO₂ interface.



Figure 5-28: SEM cross-section of the completed Device structure.



Figure 5-29: SEM of the completed Device structure.



Figure 5-30: SEM of the completed Device structure with oxide removed to show detail of the Si nanowire. Inset: Detail of an emitter tip.

5.4.3 I-V Characterization

During I-V characterization, an ion pump maintained the test chamber under ultra-highvacuum. The pressure was continually measured to monitor outgassing using the ion pump current and a Bayard-Alpert gauge. The pressure measured is below 3×10^{-10} Torr during the course of the experiments. The electrons emitted by the field emitter array is collected by a stainless steel, bakeable, Faraday cup anode placed approximately 1 cm above the surface of the chip. Three Keithley 237 high voltage SMUs source voltage and measure the emitter, gate and anode currents. These SMUs have ± 1100 V range, can source current up to 10 mA, and measure current with resolution < 10 pA. MHV-5 electrical feedthroughs allow for electrical connection to the sample under test. Contact to the cathode was made to the backside through



Figure 5-31: (A) I-V Characteristics (current per tip on a semilog scale), (B) IV Characteristics (current per tip on a linear scale), (C) F-N Plot per tip of a variety of different array sizes, demonstrating that current scales with the number of tips. In each of these devices except the 50×50 arrays at $V_{GE} > 70$ V, the total gate interception was less than 5 % of the emission current (anode efficiency > 95%.)

the chuck, and the gate was contacted with a tungsten probe.

Figure 5-31 shows the transfer characteristics for several different array sizes scaled by the number of tips in the array. Multiple sweeps were taken to ensure that the characteristics were stable. From the I-V characteristics, we see that the arrays do show good agreement with each other when scaled by the number of tips in the array. In each of these devices except the 50×50 arrays at $V_{GE} > 70$ V, the total gate interception was less than 5 % of the emission current (anode efficiency > 95%.

Figure 5-31 also shows the Fowler-Nordheim (F-N) plot, wherein $\ln (I_{A,tip}/V_{GE}^2)$ is plotted against the inverse of gate-emitter voltage. We define the Fowler-Nordheim coefficients in the

Array Size	Turn-on	Maximum	Maximum	Maximum	FN	FN Intercept	
	Voltage	Voltage	Current	Current/tip	Slope	per tip	
Single							
Emitter	31 V	60 V	2.5 μA	2.5 μA	785	-9.39	
10x10	27 V	70 V	67.8 μA	678 nA	610	-15.5	
25x25	22 V	60 V	11 µA	176 nA	445	-16.3	
32x32	23 V	65 V	1.6 mA	1.55 μA	498	-15.5	
50x50	22 V	80 V	2.6 mA	1.04 µA	468	-15.5	

Table 5.4: Summary of device performance for the characteristics

following way [116]:

$$\ln\left(\frac{I_A}{V_{GE}^2}\right) = \ln\left(a_{FN}\right) + \frac{b_{FN}}{V_{GE}}$$
(5.2)

where:

$$b_{FN} = \frac{0.95 \cdot 6.87 \times 10^7 \cdot \chi_{Si}^{3/2}}{\beta}$$
(5.3)

and

$$a_{FN} = \frac{\alpha A \beta^2}{1.1 \chi_{Si}} \exp\left[\frac{B \cdot 1.44 \times 10^{-7}}{\chi_{Si}^{1/2}}\right]$$
(5.4)

where χ_{Si} is the barrier height (eV), assumed to be the electron affinity of silicon (4.05 eV, and β is the field factor (cm⁻¹) that converts the applied gate-emitter voltage to an effective tip surface field.

The Fowler-Nordheim plot is linear when the current is from electrons tunneling through the surface barrier. Small deviations from a linear relationship could be due to quantum effects in the emitter tip [100], a non-triangular tunneling barrier due to emitter geometry [79], or space charge limitation following Child's law [23].

The saturation of the anode current observed in the 50×50 array above a gate-to-emitter voltage of 70 V shown in Fig. 5-31 is not due to current limitation by the silicon nanowire, or any of these previously stated effects. We observed that the saturation of the anode current is



Figure 5-32: The effect of changing the anode voltage on a 500×500 array that exhibits current saturation.

accompanied by a corresponding increase of the gate current. Thus, the saturation is likely due to diversion of electrons to the gate after leaving the emitter tip due to insufficient anode field. This phenomenon could be mitigated by bringing the anode physically closer to the surface of the emitter or by increasing the voltage on the anode. Figure 5-32 shows a 500 × 500 array that demonstrated similar saturation as the 50 × 50 array. By modifying the anode voltage by 10 % (i.e. $V_A = 1000 \pm 100$ V, I_A changed by $\pm 10\%$, accordingly, while the emitter current remained approximately constant.

The devices reported were designed to achieve high current and hence they do not demonstrate clear evidence of current limitation during normal DC operation. They do demonstrate, however, that they are robust, uniform, and their current scales with the number of tips in the array, thus implying that the silicon nanowires prevent catastrophic breakdown and improve uniformity.



Figure 5-33: Output Characteristics for a 500x500 Array.



Figure 5-34: Transfer characteristics for the same 500x500 array shown in Figure 5-33

Figure 5-34 shows the output characteristics, and Figure 5-33 shows the transfer characteristic for a 500×500 array (250,000 emitter tips). The device turns on at a gate-emitter voltage of approximately $V_{GE,ON} = 14$ V, and reaches > 3 μ A of current (> 16 pA/emitter) at $V_{GE} < 20$ V. As expected, for gate voltages > $V_{GE,ON}$ the anode voltage has very little dependence on anode voltage. Up until this point, we have only considered the effect of the gate-to-emitter voltage on the emission current, but in actuality, the voltage between the anode and teh emitter, V_{AE} , also applies an electric field to the emitter, and the electric field on the surface of the emitter is a superposition of the two, and we can consider an analogous field factor for the anode, which we will call γ [103] such that:

$$E_{surf} = \gamma V_{AE} + \beta V_{GE} \tag{5.5}$$

There is not much variation in emission current at high anode voltages, implying that the magnitude of γ is much less than β . This agrees with the fact that the anode is about 1 cm away from the emitter tip, while the gate directly surrounds the tip and is less than 200 nm away.

When the anode is at 0 V, the current collected by the anode approaches 0. This can be explained through the band diagram shown in Figure 5-35. In field emission, the electron are emitted from near the fermi level, and there is negligible scattering or momentum relaxation in vacuum. If the anode is at 0 V, there is a potential barrier at the anode with the height of barrier equal to the work function of the anode. This barrier repels incoming electrons, and directs them back towards the gate.



Figure 5-35: Band diagram from the emitter tip to the anode at $V_{AE} = 0$ V

When the anode voltage is increased to the work function of the anode (i.e. $V_{AE} = \phi_A$ as shown in Figure 5-36), electrons are collected by the anode. However, it is not a completely sharp on-off transition. If electrons are emitted from below the Fermi energy, they will not



Figure 5-36: Band diagram from the emitter tip to the anode at $V_{AE} = \phi_A$

not have sufficient energy to be collected by the anode. In addition, due to the wave-particle dual nature of electrons, even though classically there should be no reflection of the potential barrier on the transmission of electrons, this is not the case, and a certain fraction of electrons are still repelled by the anode. To model this effect, we assume that the potential near the anode is approximately constant, and equal to $E_F - V_{AE} + \phi_A$ where V_{AE} is the anode-emitter voltage and ϕ_A is the work function of anode. Assuming a workfuction of 4.5 eV, we calculate a collection efficiency by the anode as shown in Figure 5-37. While the collection efficiency is not quite a step function, is it quite abrupt, with $\approx 75\%$ of electrons collected by the anode at 5 V, and > 90% collected by the anode at 5.5 V.

Although there is still a retarding field to overcome, the electron should attain enough energy through the initial acceleration by the gate field to overcome it. This may not be the case if the electrons are accelerated with a large transverse velocity component. The retarding field could cause the longitudinal velocity to go to zero and reflect electrons.

Increasing the anode voltage beyond the work function of the anode further lowers the potential barrier and allows a higher percentage of electrons to be collected by the anode (Figure 5-38). This can be explained both through the transmission coefficient of the quantum-mechanical barrier increasing, and there being a lower repulsive field that the electrons have to overcome. When we reach the typical regime of operation $V_{AE} \gg V_{GE}$ all of the electrons emitted through the gate aperture are collected by the anode.

Differentiating the transfer curve of $V_{AE} = 1100$ V yields the transconductance (gm) shown in Figure 5-39. At $V_{GE} = 20$ V, the gate transconductance reaches about 30 μ S (0.12 nS/tip).



Figure 5-37: Collection probability of the anode as a function of anode-emitter voltage for $\phi_A = 4.5$ eV.



Figure 5-38: Band diagram from the emitter tip to the anode at $V_{AE} < V_{GE}$ and $V_{AE} > V_{GE}$ V

Extension to higher voltages and currents, such as those reported by Whaley et al. [126] would yield similar results, as the transconductance for a cathode scales exponentially with gate voltage, and the currents for which they report transconductance were for $V_{GE} > 60$ V.

Comparing these results to the output and transfer curves for a single emitter shown in Figure 5-41, we see some similarities. The 500x500 array turns on at a much lower voltage than the single emitter. This low turn-on voltage can be attributed to the noise floor of the measurement system, but also to the log-normal statistical variation of tip radius of emitter tips [55], that arises from the fabrication process we employed to make the nanometer-sharp silicon



Figure 5-39: Gate Transconductance of the transfer characteristics shown in Figure 5-33.

emitters. Because of the distribution of tip radius, at low voltages, the several very sharp (radius ≈ 1 nm) emitters dominate the current characteristics. After prolonged emission or emission at higher voltages, the turn-on voltage shifts to higher voltages, perhaps due to emitter tip becoming blunt or to charging of the insulating oxide.

Ranadive developed an empirical model for the proportion of the emission current intercepted by the gate as opposed to collected by anode [106]. He argued that the anode current can be described as $I_A = \alpha_f I_E$ and that the gate current is $I_G = (1 - \alpha_f)I_E$, where α_f is the transmission coefficient. Ranadive made the approximation that $al \ p h a_f$ could be approximated as $1 - \exp(-k V_{AE})$ and a linear relationship was discovered when $\ln[1 - I_A/I_E]$ was plotted against V_{AE} . This plot for a single emitter with $V_{GE} = 40$ V is shown in figure 5-40. From the plot, we confirm that the relationship is roughly linear, with $k \approx -0.097$.

100 A/cm² Devices

In Figure 5-42, the transfer characteristics of 5 different field emitter arrays of different sizes ranging from a single emitter to 2500 emitters are overlaid on each other and scaled by the number of the tips in the array. Quantitatively, the single emitter has a higher turn-on voltage than the other sizes characterized, however, it is likely that the turn-on voltage is limited by the



Figure 5-40: The Ranadive gate current model



Figure 5-41: Output and transfer characteristics for the single device



Figure 5-42: Comparison of the transfer characteristics of 5 different devices scaled by the number of emitters.



Figure 5-43: Fowler-Nordheim plot of the several different field emission arrays shown in the previous graph

noise floor in our measurement system. These results are summarized in Table 5.5

We find that as the array size increases, the effective tip radius decreases. There are two complementary explanations for this. First, proximity effects during the patterning cause the features in the center of the array to be slightly smaller than the features at the edge of the array, resulting in slightly sharper tips in the center versus the perimeter. The larger arrays have

	Single	5×5	25×25	32×32	50×50
$V_{GE,on}\left(\mathrm{V} ight)$	31	30	22	23	22
$Max V_{GE} (V)$	60	75	60	65	80
$Max I_A (mA)$	0.0025	0.240	0.655	1.55	2.6
$Max I_A / Tip (\mu A)$	2.5	0.96	1.04	1.55	1.04
Max J_A (A/cm ²)	250*	96	104	155	104
FN Slope, b_{FN} (V)	785	645	503	498	468
FN Int., $\ln(a_{FN})$					
$(\ln(A/V^2))$	-9.39	-10.93	-7.93	-8.49	-7.72
FN Int./tip, $\ln(a_{FN})/tip$					
$(\ln(A/V^2))$	-9.39	-14.15	-14.37	-15.42	-15.54
Field Factor, β					
$(cm^{-1} \times 10^6)$	0.669	0.821	1.054	1.068	1.137
Tip Radius (nm)	6.0	5.0	4.03	3.20	2.93

Table 5.5: Summary of device performance for the characteristics

a smaller number of perimeter emitters, thus there are a higher percentage sharper emitters in the larger arrays. This effect can be mitigated by using a more advanced lithography technology that employs edge proximity correction in the mask layout of subsequent revisions. Another explanation is that the emitter-tip radius follows a well-documented log-normal or Gaussiandistribution [69]. As the array size increases, there is an increased likelihood that there are several sharper emitters in the array. These sharper emitters will dominate the performance at lower gate-emitter voltages and decrease the slope of the Fowler-Nordheim plot [95]. It is likely that both effects are contributing to the reduction in tip radius observed in our characteristics for the larger array sizes.

The saturation of the anode current observed in the 50×50 array above a gate-to-emitter voltage of 70 V is not due to current limitation by the silicon nanowire. The saturation of the anode current is accompanied by a corresponding increase of the gate current. The anode current saturation can thus be explained by the diversion of electrons to the gate after leaving the emitter tip. Due to strength of the gate-to-emitter (lateral) electrostatic field relative to the anode-to-emitter (vertical) electrostatic field, the electrons preferentially arrive at the gate instead of the emitter at gate-emitter voltages greater than 70 V due to insufficient anode electric

field. By increasing the voltage on the anode, or by bringing the anode physically closer to the surface of the emitter, the anode electric field can be increased, leading to an increase of the gate-emitter voltage at which the electrons are intercepted by the gate can occur. This effect was measured and is shown in Figure 5-32.

In Figure 5-44, we compare the I-V characteristics of a single emitter with a simulated device in SILVACO. The details of the simulation are given in Section 5.2. A 2.5-D simulation structure was used were considered, where a 2-D axisymmetric structure was generated programmatically in MATLAB so that the tip radius and meshing is precisely defined. We find that using the doping and the tip radius as parameters, we can get good quantitative agreement between the simulation and the current measured for reasonable tip radii and doping densities.

To fit the simulation results to the measured I-V characteristics, a tip radius of 6.5 nm and a doping density of 2.5×10^{16} cm⁻³ were used. A simulation of the wet thermal oxidation used to sharpen the tips and shrink the nanowire suggests that the doping density inside the nanowire is slightly lower, ~ 1.5×10^{16} cm⁻³. The simulated I-V characteristic had an FN slope of 624, in agreement with the measured FN slope of 626. The FN intercepts of the simulated and measured characteristics were -12.38 and -12.63. Using the model developed in Section 5.4.4 below, we extract a tip radius of 8.3 nm. The experimental FN characteristic appears to deviate from its linear relationship at $V_{GE} \approx 60$ V, and the simulation suggests that the current limiter saturates at $V_{GE} \simeq 60$ V.

5.4.4 New technique for electrical estimation of emitter tip statistics

It is a significant challenge to image the fine structure of the tip with TEM and SEM due to the many different material layers in the field emitter structure, and the high packing density. The particular challenge of imaging the field emitter tip with high resolution SEM is charging of the surrounding oxide layers. With TEM, there is a significant risk of damaging the tip geometry and re-depositing of material onto the tip during sample preparation with focused ion beam (FIB) milling. In the absence of traditional tools of analyzing the uniformity of our field emitter tips, we explore below the extraction of the tip distribution using our current voltage characteristics.

A simple analytical model relating the field factor, β , to the emitter tip radius is the ball-



Figure 5-44: Comparison of simulated I-V characteristics (left) and FN chart (right) to experimental data for a single emitter (black dots).

in-sphere model [32]. In the ball in sphere model, the emitter tip is modeled as a hard, metal sphere of radius r in a large, spherical shell of radius R. The relationship of the voltage on the sphere to the electric field at the surface of the sphere is simply given by the solution of Laplace's equation in spherical coordinates.

$$\beta = \frac{1}{r} - \frac{1}{R - r} \approx \frac{1}{r} | r \ll R \tag{5.6}$$

While this model is very simple to use and could give results that are reasonable to within an order of magnitude, it is not very accurate for realistic tip and gate geometries and underestimates the tip radius for emitters that are sharper than 20 nm. To build a better estimate for tip radius for our structure, we developed a finite element model for the silicon nanowires capped with field emitter tips embedded in a dielectric matrix using COMSOL Multiphysics. The element (silicon nanowire capped with silicon tip) is modeled using cylindrical symmetry around the central axis of the emitter (i.e. r = 0). The emitter is simulated as having an emitter half-cone angle of 30°. The nanowire diameter is 200 nm, and the nanowire is 8 microns tall. The emitter tip radius is varied between 0.5 nm and 50 nm. The gate aperture is 350 nm, and the gate thickness is 250 nm. Aside from the tip, the pillar is buried in silicon dioxide with relative dielectric constant $\epsilon_{ox} = 3.9$. The surface is assumed to be ideal with no fixed charge or



Figure 5-45: Finite element electrostatics simulation to determine the relationship between tip radius (r) and field factor (β) for the silicon nanowire with self-aligned gated tip. The nanowire is 200 nm in diameter and 10 microns tall, with a tip half-cone angle of 30 degrees. (A)-(C) Detail of the electric field around the apex of the tip for tip radii of 10 nm (A), 5 nm (B), and 1 nm (C).

surface states. In addition, it is assumed that there is no current or space charge in the system. The emitter is assumed to be a perfect conductor, that is, the voltage along the emitter surface is set to 0 V. The polysilicon gate is also modeled as a perfect conductor, and the voltage on the gate is 1 V. The boundary condition at r = 0 is set to $(\partial E_r)/\partial r = 0$, and the solution of the potential and the electric field in the structure is reduced to a boundary value problem. With these boundary conditions (i.e. $V_{GE} = 1$ V), the maximum electric field at the apex of the field emitter in V/cm is the field factor, β .

The finite element simulation of the tip geometry, shown in Figure 5-45(A)-(C), plots the electric field proximal to tips of various emitter radii. By reducing the emitter tip radius from 10 nm to 1 nm, the field factor is increased from 0.76×10^6 cm⁻¹ to 4.0×10^6 cm⁻¹. If we plot the results for emitters in the range 0.5 < r < 50 nm, we find the result shown in Figure 5-46. Dvorson et al. [37], Jensen et al. [70] and Ding et al. [32] developed analytical models that show that the field factor for conical emitters follow the form $\beta = k/r^n$. By fitting an equation of that form to the field factors extracted from the numerical model, we obtained a very good fit to our data with a $\beta = 4.075 \times 10^6/r^{0.7417}$ cm⁻¹ for r in nm.

It is unsurprising that the finite element model indicates that β does not strictly follow 1/r. for $\beta = 1/r$ to be precisely true, spherical symmetry would need to be maintained. instead, the gate is roughly cylindrical, the emitter tip is a cone, and the gate only subtends a fraction



Figure 5-46: Fit of field factors with the model $\beta = k/r^n$ from the finite element electrostatics model shows good experimental agreement for this structure for r = 0.5 - 50 nm.

of the angle of the emitter tip surface.

From the relationship between tip radius and the field factor for our structure, we can use the slope of the Fowler-Nordheim plots of our I-V characteristics to estimate the average radius of the emitter tips in the array assuming that the silicon workfunction is its electron affinity. Over 150 plots from 20 different devices were analyzed for their Fowler-Nordheim slopes, and their tip radii were extracted. Table 5.6 shows the number of characteristics used for each device size. The results for the tip radii estimates for the different plots are summarized in the box chart shown in Figure 5-47.

The mean of the average tip radii extracted from the current voltage characteristics of the filed emitter arrays decreases from 8.7 nm for the single emitter to 4.8 nm for the 25×25 array. It is likely that in the smaller arrays, edge effects have a large effect on the tip radius distribution. For example, in the 10×10 array, 36 % of emitters are on the perimeter, and 64% of emitters are either on the perimeter, or nearest neighbors to the perimeter. As the size of the array increases, the percentage of emitters along the perimeter becomes. It is expected that the lithographic
conditions (photon flux distribution) is not symmetric for nanowires/emitters situated on the perimeter. This will result in features that are slightly smaller. During oxidation, these slightly smaller features might have been over-oxidized, resulting in an emitter tip that is not as sharp.

Array Size	Number of Characteristics Used			
1	39			
100	11			
625	28			
1024	20			
2500	15			
10000	25			
250000	11			

Table 5.6: Summary of device performance for the characteristics



Figure 5-47: Box chart of the distribution of average effective tip radii extracted from I-V characteristics and fitted to tip radius using the extrapolation from the finite element model. The square in the center of each box indicates the mean value of tip radius of the distribution. The horizontal line inside of each box is the median value of tip radius. The upper and lower bounds of the box show the first (Q_1) and third quartiles (Q_3) of the tip radius distribution, the height of the box is the inter-quartile range (IQR). The terminal points of the vertical lines extending from the boxes show approximately $\pm 3\sigma$ (i.e. $Q_1 - 1.5 \times IQR$ and $Q_3 + 1.5 \times IQR$). The starred points are outliers.



Figure 5-48: Probability density functions of the average effective tip radii of different array sizes based on the statistics of the data summarized in Figure 5-47 superimposed a histogram of 200 tip radii measured on a representative sample.

This effect will be reduced by adding perimeter corrections to the mask.

For arrays larger than 625, the average tip radius tends to be between 4 and 5 nm, consistent with previous work on silicon field emitter arrays. For arrays larger than 1000 emitters, the variation tends to be smaller. This is probably due to the law of large numbers, and the averaging effect that takes place with the large number of emitters. It was expected that there would be a slight trend for the tip radius to decrease with array size due to the presence of a few sharper "hero" emitters dominating the emission at lower voltages; however, this effect does not appear to be present in the data presented. Either the arrays are extremely uniform that there are not many of these very sharp emitters, or the presence of the silicon pillar nanowire prevents them for contributing an overly large percentage of the current. If the statistics obtained from the IV characterization are plotted as Gaussian probability density functions, as shown in Figure 5-48, the larger arrays all converge around 4.8 nm. The distribution is consistent with those obtained from SEM measurements of tips without gates previously reported in Guerrera et al. [55]. We

should add a caveat that the distribution extracted from the IV characteristics would naturally favor smaller tip radius and perhaps tighter distribution principally due to the exponential relationship between the field factor (and hence the tip radius) on the emission current.

5.4.5 Emission Angle Measurements

This section examines the angle of emission of the silicon FEA through the use of a low voltage phosphor screen. From the spot size, we can obtain estimates about the distribution of transverse energy of the electrons that are being emitted and make some estimates about the brightness of our electron source.

Figure 5-49 shows a schematic of the experimental setup used for these measurements. The field emitter array with width a emits electrons. These electrons have a certain transverse velocity v_x that gives rise to a divergence in the beam once it leaves the emitter. A ZnO phosphor screen, biased at 900 V is placed at a height h above the field emitter array. When the emitted electrons impact the phosphor screen, they excite phosphors in a spot on the phosphor screen with width d through cathodeluminescense (CL). In CL, the electrons generate electron-hole pairs called excitons. These excitons transfer energy to an impurity "activator" ion, which is then excited to a higher level, relaxes to a radiating level, and then radiatively transitions back to the lower level to produce light[107]. This light is then collected by a microscope objective, and imaged onto a scientific CCD digital camera (QImaging Retiga camera).

A 25 × 25 array was used for these experiments. The I-V and F-N characteristics of the device are shown in Figure 5-50. The device turned on at ~ 30 V and had an average tip radius of ~ 10 nm based on the Fowler-Nordheim slope of 700 V. During I-V characterization, the phosphor screen anode was kept at ~ 5 mm, and biased at $V_{AE} = 900$ V, while the gate voltage, V_{GE} was ramped from 0 to 60 V. For the imaging of the phosphor screen, the gate electrode was biased at a constant gate-emitter voltage $V_{GE} = 60$ V and the anode at $V_{AE} = 900$ V. The array produced an anode current of $I_A \approx 3.5 \ \mu$ A. The gate leakage is about 50 nA at 60 V, and follows Fowler-Nordheim behavior. Images of the phosphor screen were collected at various heights between 4.6 mm, and 8.3 mm, as shown in Figure 5-51.

To calibrate the height of the phosphor screen above the sample, the trendline of the width of the spot above the sample was extrapolated back to a separation of 0, and set to the width of the array, 25 μ m. The relative spacing between the different spots is measured by a micrometer and accurate to ±25.4 μ m. To obtain repeatable results, care was taken to ensure that each image was collected and processed exactly the same. The anode was kept at a constant bias of 900 V, and the camera used a 90 second exposure with a gain of 8. Such a long exposure was needed because of the low efficiency of low-voltage phosphor screens. Recombination at the surface quenches phosphorescence and reduces the brightness, and because the phosphor screen is not metallized, half of the light that is generated is directed downwards and away from the camera, and more of the light is piped away from the camera in the glass substrate of the phosphor screen due to total internal reflection.

After the image was collected and saved, it was imported into photoshop. An image of the substrate was used to calibrate the size of the pixels to quantitatively measure the spot size. In photoshop, the width of the spot was measured, and the brightness/contrast was changed to +75/-25 respectively to obtain a clear reproduction when printed.

The spot on the phosphor screen appears oblong, with one axis cut off on one end. The end is that is cut-off is likely due to the probe deforming the electric field and occluding the phosphor screen. The spot may be longer in one axis than the other because of non-uniformity



Figure 5-49: Diagram of the measurement Setup



Figure 5-50: I-V characteristics (left) and Fowler-Nordheim characteristics (right) of the device used for emission angle experiments

in the chemical-mechanical polishing, resulting in the top not being exactly in the center of the aperture. Figure 5-52 plots the width of the spot vs. height. While the accuracy of the spacing between the points is $\pm 25 \ \mu$ m, to calibrate the height of the phosphor relative to the substrate, the points were shifted so that the y-intercept of the least squares linear regression was half the array width.

From the spot size, we can calculate the emission half-angle through:

$$\tan\theta = \frac{1}{2}\frac{dx}{dh} \tag{5.7}$$

Where dx/dh is the slope of the half-width of the spot size. We find that the emission angle is $\approx 12.3^{\circ}$ (≈ 200 mrad). This number agrees quite well with previous results reported in the thesis of M. Ding, who found that Si FEAs with apertures formed with CMP had an emission angle of 12.6° [30].

From the spot size, we can obtain an estimate of the transverse velocity and transverse energy of emitted electrons. When an electron is emitted at an angle away from the apex of the emitter tip, it will gain some transverse energy due to the roughly spherical gate field profile around the tip. Once the electron is far enough from the gate electrode (approximately several gate diameters), it will be accelerated by the anode field (which is roughly constant and perpendicular to the substrate), but the electrons will retain the transverse velocity they attained from acceleration from the gate field. Hence, the transverse velocity should depend on the gate



Figure 5-51: Dependence of the spot size on the height of the phosphor screen. Spots produced on the phosphor anode with $V_G = 60$ V, $V_E = 0$ V, $V_A = 900$ V, $I_A \approx 3.5 \mu$ A. The gate leakage is about 50 nA at 60 V, and appears to follow Fowler-Nordheim behavior. The spot size measurement was taken on the short axis of the spot. There was a linear relationship between spot size and the height.

voltage, but not the anode voltage.

From the height, h, we can calculate the travel time Δt by assuming that the force on the electrons is constant and equal to qE where the electric field is approximately equal to the anode voltage divided by the separation between the phosphor screen anode and the FEA, i.e. $E = V_A/h$. With a known travel time, we can then estimate the transverse velocity, v_x given the displacement Δx , assuming that Δx is the half-spot size.

$$h = \frac{1}{2} \frac{q V_A}{m h} (\Delta t)^2 \to \Delta t = h \sqrt{\frac{2m}{q V_A}} \to v_x = \frac{\Delta x}{h \sqrt{\frac{2m}{q V_A}}}$$
(5.8)



Figure 5-52: Extraction of emission angle from spot size measurements



Figure 5-53: Transverse velocity (left) and energy (right) for the different anode-emitter separtions measured.

Using this approach, we find a transverse velocity of $\approx 2 \times 10^6$ m/s. Estimating the transverse energy as $E_x = \frac{1}{2}mv_x^2$, we find that the electrons have a transverse energy of ≈ 12 eV. Figure 5-53 shows the transverse velocity and energy of the different anode heights measured, showing good agreement between the different measurements.

5.4.6 Failure Analysis

When gate-emitter voltages greater than approximately 65-80 V are applied, the devices can sometimes exhibit sudden catastrophic failure. These failures are associated with a sharp pressure rise and a drop in emission current. Figure 5-54 is an example of the I-V sweep of a device that failed at $V_{GF} = 70$ V.

Before failure, the device was emitting approximately 650 nA/emitter, for a current density of 65 A/cm². The gate current was $\approx 1 \ \mu$ A, less than 1 % of the total current emitted by the anode. During the 6th sweep, emission suddenly ceased, and the pressure increased from $\approx 2 \times 10^{-10}$ Torr to 1×10^{-9} Torr. The gate current characteristic changed: the gate current decreased at higher voltages, but increased at lower voltages.

A plausible explanation for the shift in the gate current is thus: Before failure, the gate current at high voltages is due to interception of emitted electrons by the gate. This is corroborated by the almost linear Fowler-Nordheim characteristic of the gate current at higher gate-emitter voltages. After the sudden failure, the electron emission is "turned off" but a resistive leakage path between the substrate and the gate is formed, causing the gate leakage to be higher at low voltages, but lacking the exponential nature of field emission.

There are several possible causes of this dielectric failure:

1. Charge injection and breakdown inside the oxide



Figure 5-54: I-V and FN plots of a device that demonstrated sudden catastrophic failure at $V_{GE} = 70 \text{ V}$



Figure 5-55: Comparison of the finite element simulation of the ridge found at the edge of the array with a cross-section SEM of the ridge. There is a field enhancement of approximately 5 at the apex of the ridge.

2. Flashover and vacuum breakdown at the gate, SiO2, vacuum triple point [5, 50, 115].

Cross-section SEMs of the completed device revealed a possible failure point is a sharp ridge that formed around the active device region. This ridge was inadvertently created during the mesa etch due to the etch being unoptimized and thus having an isotropic component. This ridge has a prominence of 1 μ m above the substrate, and a radius at the tip of approximately 120 nm. Finite element simulations of the ridge indicate that the electric field at the apex of the ridge is > 5× greater than the electric field in the rest of the oxide, and thus is a likely location for failure. Figure 5-55 compares the finite element model (with equipotential lines drawn) with the SEM of the structure.

With 70 V on the gate and the substrate at ground, the electric field at the apex of the ridge is approximately 3 MV/cm. The intrinsic dielectric strength of SiO_2 is often reported at 10 MV/cm [119], however, this SiO_2 has undergone significant plasma processing, as well as CMP. It is possible that some of these processes have introduced defect sites into the oxide, reducing its dielectric strength and causing failure at lower than expected fields.

5.5 Summary

This chapter covered the design, fabrication process development, fabrication, modeling, and characterization of silicon field emitter arrays with self-aligned gates and integrated silicon nanowire current limiters. Exploratory experiments in filling the voids between high-aspectratio silicon pillars revealed that it was a significant materials and processing challenge, and extensive work was done to develop a process for the void-free, filling of these gaps with a high quality dielectric material that could support the polysilicon gate. A process that used a combination of undoped polysilicon deposition and oxidation, as well as low-stress silicon nitride was presented, as well as a method that used selective etching of silicon nitride over silicon dioxide to form self-aligned gate apertures. These exploratory experiments yielded functional field emitter arrays, demonstrating a turn-on voltage ≈ 20 V and current saturation at 30 V, however, these devices suffered from large gate-substrate leakage current. To address this leakage current, a new device architecture was proposed that incorporates a thick dielectric surrounding the array, and a process to ensure that the gate polysilicon remains outside of the active array after processing.

These new devices exhibit record current density for Si field emitter arrays, with sustained currents > 1 μ A/tip and current densities > 150 A/cm² @ V_{GE} < 70 V. The results are consistent with numerical simulations and analytical models. A number of different-sized arrays were measured, showing that current scales by the number of emitters in the array, suggesting both that the emitter tip radius distribution is quite uniform, and that the current limiter is preventing the destruction of sharper emitters. From the I-V characteristics, a new technique from extracting the tip radius statistics was developed, and the results of the method agree well with tip radii measured with SEM.in

Chapter 6

Lifetime Characterization of Silicon Field Emitter Arrays

6.1 Introduction

In this chapter, the lifetime and reliability of silicon field emitter arrays with integrated silicon nanowire current limiters and self-aligned gate apertures is explored. Lifetime and reliability are important metrics for some applications of electron sources, with requirements ranging from several minutes to years. To test the lifetime of our FEAs, we operate them with the Keithley 237 that biases the emitter in constant current mode, with the gate power supply set to 0V. The Keithley 237 has an internal feedback loop to maintain the current, and will automatically adjust the voltage should the I-V characteristics change. At regular intervals of every 1 or 5 hours, the measurement is paused and an I-V sweep is taken to monitor the health of the device. After the I-V sweep, the current is ramped up and lifetime test is continued.

During lifetime measurements, the anode was biased at a voltage of 1 kV. The power dissipated by our anode is limited to 1 W to prevent heating of the anode and outgassing, which would have the potential to cause back-ion bombardment and damage the emitter, thus the beam current is limited to 1 mA during lifetime characterization. We tested several different sized arrays, and have achieved up to 100 hours @ 100 A/cm², and over 10 days @ 1 mA and 0.1 A/cm^2 . Only the 50 × 50 array at 100 A/cm² demonstrated irreversible failure when the test was concluded.

Array size	1000×10000	100×100	50 × 50	Single Emitter
Current (mA)	1	1	2.5	0.001
Current Density(A/cm ²)	0.1	10	100	100
Starting Voltage (V)	43	54.5	71	55
Lifetime (hours)	320	100	6	100
Failed?	n	n	у	n

Table 6.1: Devices Characterized for Lifetime

Table 6.1 Summarizes the lifetime results.

6.2 The S-K Chart

In this chapter, we examine the behavior of a single sample over time, and plot the slope and intercepts from linear fits of different FN-plots. These plots are referred to as Seppen-Katamuki plots or SK-plots for short. An example S-K plot is shown in Figure 6-1. The name Seppen-Katamuki comes from the Japanese words for slope and intercept, following the work and convention of Gotoh et al. [49, 48]. In SK-plots, a linear relationship between the slope and the intercept is often observed, however, this linear relationship cannot be explained assuming only a change in a single parameter, such as the radius or the work function. For a possible explanation using the simple FN-equation, a radial dependency of the emission area must be included [66, 47].

6.3 0.1 A/cm² Device

We have demonstrated 10 days of continuous operation of a 1000×1000 array (1M emitters) at an anode current of 1 mA (J = 100 mA/cm^2) (Fig. 6-3). During the course of these experiments, the operating voltage increased 4.8%. This voltage increase is likely indicative of charge injection and trapping at the surface of the oxide, resulting in de-biasing of the gate and a reduction in the electric field at the surface of the emitter. The plot of voltage vs. time in Fig. 6-3 shows spikes at regular intervals. These spikes occur each time the lifetime measurement is paused to collect the I-V characteristics.

The S-K chart of I-V characteristics taken during the course of the lifetime characterization



Figure 6-1: Example S-K chart. From [49], reproduced with permission. ©2001 AIP.

is shown in Figure 6-2. One I-V characteristic per day was used to generate the S-K. The S-K chart shows a clear trend towards larger apex radius and duller emitters over the course of the experiment. Although each time the lifetime test was paused for I-V characteristics the voltage required to maintain 0.1A/cm² was lower, and the voltage increases back to its previous value. It was initially thought that perhaps this voltage change suggests that it is possible that



Figure 6-2: S-K Chart of the 0.1 A/cm² device.

the degradation may be due to charging of the gate oxide or the gate itself, and that the effect could be reversible, however, the S-K chart indicates that there is also a change in the effective structure of the tips.

6.4 10 A/cm² Device

The 100×100 array (10^4 emitters) was operated continuously for 100 hours at an anode current of 1 mA. The results are plotted in Figure 6-7. This current corresponds to a current density $J = 10 \text{ A/cm}^2$. This device demonstrated a more severe voltage increase of 14.8% over the 100 hours to maintain 10 A/cm² current density. Figure 6-4 shows that the voltage shift is worse for the first 10 hours, and then increases linearly at a rate of 0.05 V/hour. Interestingly, by allowing the device to relax for 24 hours, the device recovered and the original I-V characteristics were restored as shown in Figure6-5.

From the I-V characteristics collected every 5 hours, we calculated the Fowler-Nordheim $\ln(a_{FN})$ and b_{FN} parameters and plotted $\ln(a_{FN})$ and b_{FN} to make the Seppen-Katamuki (S-K) chart in Figure6-6. From the S-K plot, we see that over time, the voltage required to achieve emission is increasing, consistent with either work function increasing or de-biasing due to charge trapping. Because the initial I-V characteristics were able to be recovered, it is more likely that de-biasing due to charging is the reason for the shift in characteristics.



Figure 6-3: Lifetime summary - 1mA, 0.1 A/cm², 10 Days after initial 70 hour experiment - 310+ hours total.



Figure 6-4: The operating voltage of the 100x100 array increased linearly with time.



Figure 6-5: I-V characteristics showing the recovery of the 100x100 device.



Figure 6-6: S-K chart of the FN characteristics showing the recovery of the 100x100 device.



Figure 6-7: Lifetime summary - 100x100 array 1mA, 10 A/cm², 100 hours total.

6.5 100 A/cm² Devices

A 50x50 array was operated at an anode current of 2.5 mA (Current density, $J = 100 \text{ A/cm}^2$) for a period of 5 hours before failing. Figure 6-10 shows a summary of the data taken during lifetime characterization. The voltage increased linearly over 6.5 hours of operation, beginning at 70 V until it reached 75 V when a failure occurred. This increase of 0.77 V/hour is more drastic than the voltage increase in the 10 A/cm² devices and is probably attributed to the higher operating voltage injecting more carriers into the dielectric film. Thus, this event is likely dielectric breakdown which resulted in outgassing which damaged the emitters. Due to the emitter damage, there was a sharp reduction in emission current, increasing the driving voltage to the compliance limit of 90 V. After 5.5 hours at 90 V and anode current between 1 and 1.5 mA, the device demonstrated irrecoverable failure. Both of these failures were accompanied by spikes in the pressure due to arcing and emitter outgassing.

A single emitter operating at an anode current of 1 μ A (for an equivalent array current of 100 A/cm²) maintained that current for over 100 hours with negligible degradation (Fig. 6-11). The 4.4% standard deviation in voltage is likely due to fluctuations in the state of the emitter surface [65], as shown by the S-K chart in Figure 6-9). There is no clear trend in the progression of the F-N characteristics of the 1x1 array, however, these scatter in this S-K characteristic is consistent



Figure 6-8: S-K chart of the 50x50 Array.



Figure 6-9: S-K Chart of the single emitter shows random walk over time.

with the S-K characteristic of a single Pt Spindt-type emitter demonstrated by Gotoh et al. [48].



Figure 6-10: Lifetime summary - 50x50 array, 2.5 mA, 100 A/cm², 5 hours.



Figure 6-11: Lifetime summary - Single Emitter, 1µA, 100 A/cm², 100 hours total

6.6 Discussion

The variation of the S-K chart for single silicon emitters has been observed by several in the literature, however, a satisfactory explanation for the behavior still does not exist. As Persaud observes [101], because the distribution of the points along the line appears random, explaining the data is a challenge. The linear relation shown in the SK-plots seems to be orthogonal to a change in work function. Charbonnier *et al.* also investigated further possible explanations for the linear behavior in SK-plots, and considered whether nano-protrusions on top of a single emitter can be the cause. They found that a single protrusion cannot fit the experimental data but when several protrusions are used, good linear fits to the SK-plots can be obtained [20].

The body of literature that examines the long-term temporal stability of silicon field emitter cathodes is particularly sparse. Most papers that examined temporal stability, only presented short term stability such as those presented by Itoh and Hong[67, 62] and many others. Temple et al. demonstrated 25 hour operation of a 3255 tip array operating at $\approx 100 \ \mu$ A for a current of 30 nA/tip, or a current density of 0.03 A/cm² [121]. When we expand our literature search to carbon nanotube and metallic emitters, the picture improves somewhat, but perhaps the only comparable results in terms of lifetime at high current and high current density come from metal cathodes fabricated by SRI, with published reports of operating at 120 mA, 13 A/cm², and 2.4 μ A/tip for several days [115].

Our results stand apart from the body of literature for silicon FEAs in terms of current density and lifetime. We can attribute this enhanced performance to our device architecture which prevents the failure mechanisms. Further work needs to be done to quantify these lifetime results and demonstrate how this devices will eventually degrade and fail. If these failure mechanisms could be well understood, we could develop approaches for performing accelerated lifetime testing such as is done with transistors.

6.7 Summary

Lifetime characterization demonstrated > 100 hours of operation for a single emitter operating at 1 μ A (for an equivalent array current of 100 A/cm²) and of > 100 hours for a 10 × 10 array operating at 10 A/cm². The emission characteristics degraded during this time however, by allowing the device to relax, the original characteristics could be recovered. Only a very limited subset of devices and potential failure mechanisms have been explored and there is very rich set of issues that can be explored in this area. However, these limited results suggest that a potential degradation mechanism of the 100×100 array could be charging of the insulator between the gate and the emitter.

If the degradation mechanism is indeed dielectric charging, there are several ways that this effect could be mitigated. The first is that the dielectric could be made thicker. Another option is to coat the dielectric with a thin film of conductive, but highly resistive material. This film could prevent the accumulation of charge in the dielectric.

Chapter 7

Potential New Applications of FEAs

In this chapter, experiments in extending the operation of field emitter arrays to challenging applications where a pristine vacuum environment is not available are explored. Two of these experiments include operation in a high-pressure helium ambient, and the use of an electrontransparent graphene gate to isolate the emitters from the ambient environment.

7.1 Operation in Helium Ambient

An alternative to solid-state electronics for high-frequency power application are vacuum electronics; however, the need for vacuum chambers and vacuum pumps limits their utility, particularly in applications where size and weight and power (SWaP) are constrained. In this section, we report an approach that uses a gas channel medium approaching 1 atmosphere (500 Torr) to achieve simultaneously high breakdown field and high electron velocity, without the need for a pristine vacuum.

The operation of the field emitter device in helium can be thought of and modeled as a transistor where the channel medium is a gas rather than a solid state material, i.e. the "He channel transistor." In both a planar MOSFET and in the He channel transistor, a gate electrode modulates a barrier to control the injection of carriers into the channel. In a MOSFET, a thermionic process is used to inject the carriers into the channel. This is because the barrier between the source and the drain is < 1.1 eV, the bandgap of Si, and the gate is only several monolayers of an insulator away from the channel, often a high- \varkappa dielectric. This description of MOSFET operation is particularly obvious when thinking about the subthreshold region of operation. The thermionic injection of carriers from the source is what gives rise to subthreshold current, and the ideal subthreshold swing of 60 mV/dec.

In a gas channel, the barrier is much higher, closer to the work function of the material, $\approx 4-5$ eV, and it is a significant challenge to engineer materials with a work function low enough to make the injection of electrons into the channel via a thermionic process, as reducing the work function of the material thus increases the reactivity of the material (i.e. the definition of oxidation in a chemical sense is the loss of electrons). Hence, our approach is to inject electrons into a gas channel using the field emission arrays presented in Chapter 6, in which all the bias voltages (gate to emitter voltage for electron injection and anode to emitter voltage for electron extraction) are all below the first excited state energy of the gas to ensure that the only interactions with gas molecules is elastic scattering. Figure 7-1 shows a schematic of the



Figure 7-1: Comparision of the helium channel transistor to a planar MOSFET

transistors and their principles of operation, where applying a voltage between the gate and the emitter/source causes the injection of electrons into the channel.

For our channel, we chose helium for its low reactivity and high first ionization energy (≈ 24 eV) due to the tight binding of the valence electrons to the helium nucleus. Furthermore the excited states resulting from electron impact also have very high energy with the lowest having energy greater than 19.5 eV [2]. It is expected that electrons in a helium channel should have a higher mobility than in a solid-state device, as the density of gas molecules is much less than the density of atoms in a solid-state lattice.

To estimate the mobility of electrons in helium, we can use the relaxation time approximation [85], starting with the assumption that the electrons are traveling at a constant drift velocity v_{drift} with no elastic collisions (momentum relaxation), i.e.

$$qF - \frac{m^* v_{drift}}{\tau_m} = \frac{\partial (m^* v_{drift})}{\partial t} = 0$$
(7.1)

where qF is the acceleration force from the electric field applied by the anode, and $m^* v_{drift} / \tau_m$ is the "drag force" from elastic scattering by the gas molecules. m^* is the electron mass, and τ_m is the momentum relaxation time, and is the time required to randomize the momentum. The momentum relaxation time can be thought of intuitively as the average time between scattering events. It is trivial to solve for the drift velocity under this steady state condition:

$$\nu_{drift} = \frac{q\tau_m}{m^*} F \tag{7.2}$$

and hence the mobility, μ is:

$$\mu = \frac{q\,\tau_m}{m^*} \tag{7.3}$$

To calculate the time between collisions, we assume that the helium atoms in the gas are evenly distributed with density N_{He} . If the electrons are traveling with velocity v_e i.e. $v_e = \sqrt{\frac{2qE}{m^*}}$, we can use elastic collision cross-section, σ to calculate to find the momentum relaxation rate:



Figure 7-2: Scattering cross-section for low energy electrons in helium, after [16]

$$\frac{1}{\tau_m} = N_{H_\ell} \sigma \sqrt{\frac{2qE}{m^*}} \tag{7.4}$$

We use the average value of σ as tabulated in [16] for electrons with energy between 0.1 and 20 eV. Figure 7-2 shows that the scattering cross-section for electrons of energy 0 – 20 eV is relatively constant, with values varying between 3 and 6 ×10⁻¹⁶ cm⁻². Not visible in this data is the dip in elastic scattering present at the 19.366 eV due to the 1s²s² resonance. Note that σ is independent of pressure. The scattering rate, however, depends on the density of gas atoms/molecules in the channel, which obviously changes with pressure.

Thus, assuming the average energy of electrons is 20 eV and the scattering cross-section is 4 Å^2 , the mobility of electrons in He at 1 atmosphere is

$$\mu_{e,He} \approx 1700 \ \frac{\mathrm{cm}^2}{\mathrm{V} \cdot \mathrm{s}} \tag{7.5}$$

Surprisingly, this mobility comparable to the mobility numbers for Si and GaAs. A more rigorous calculation of this mobility would take into account the energy dependent nature of the momentum relaxation time and solve the Boltzmann transport equation assuming a uniform electric field, however, the numbers we chose should provide a pessimistic estimate. Furthermore, we can perform an estimation of the saturation velocity of electrons assuming steady state inelastic scattering with energy transfer to the first excited state of helium under electron impact.

In that case:

$$\frac{\partial \langle E \rangle}{\partial t} = 0 = q F v_e - v_\mu E_{transfer}$$
(7.6)

Experiments to measure the differential inelastic cross-section of helium show that that threshold of excitation of the metastable 2^{3} S triplet state of He is 19.819 eV [4], and it could have a lifetime as long as 8000 s [58, 15]. Thus, from energy balance, the saturation velocity, v_{sat} is given by:

$$v_{sat} \approx \sqrt{\frac{E_{inelastic threshold}}{m^*}}$$
(7.7)

Using eq. 7.7 and assuming that the energy is 19.812 eV and the electron mass is the free electron mass, we obtain a saturation velocity of 1.8×10^8 cm/s, more than an order of magnitude higher than the saturation velocity numbers obtained in solid state electronics for silicon and gallium arsenide.

As shown in Figure 7-3, He has breakdown voltage of greater than 160 V at atmospheric pressure with a minimum at about 50 microns [3]. Thus, when the electrode is placed at a distance below the mean free path of He at STP, the breakdown voltage will be greater than 160 V. Hence, a way to increase the breakdown voltage is to reduce the spacing between the electrodes so that it is less than the mean free path of electrons in He. At 1 atmosphere, as shown in Figure 7-4, the mean free path is 9.5 μ m.

Recall from Chapter 1 that the Johnson Figure-of-Merit is (J-FOM) is a figure of merit of power semiconductor devices, and defined as J-FOM = $\frac{E_{bd}v_{rat}}{2\pi}$. The above calculations imply that the J-FOM of a "helium channel transistor" could be greater than 10¹⁴ V/s, compared to the Si J-FOM of 4.5×10^{11} V/s or the GaN J-FOM of 9.5×10^{12} V/s [89]. Electronics with J-FOM of 10^{14} V/s without the need for high vacuum could result in a new architecture for devices that require high power amplification at high frequencies with small SWaP requirements.

To move towards a helium channel transistor, we have demonstrated the operation of low-



Figure 7-3: Paschen's curve for helium at 1 atmosphere



Figure 7-4: Mean free path of electrons in helium as a function of pressure

voltage field emission cathodes suitable for operation in helium. Figure 7-5 shows a 1000 × 10000 field emitter array that turns on at $V_{GE} = 8.5$ V and emits over 1 nA of current at $V_{GE} < 10$ V. This device shows a current swing > 2 orders of magnitude with $\Delta V_{GE} = 1.5$ V.

Another cathode emitted over 1 nA of current at V_{GE} < 14 With the anode biased at 19 V, below the first ionization potential to prevent plasma formation and back-ion bombardment.



Figure 7-5: I-V and FN Characteristics of a 1000x1000 array that turns on at 8.5 V



Figure 7-6: I-V (top) and FN (bottom) Characteristics of a 1000 \times 1000 array operated in 100 mTorr of helium with V_{A} = 19 V

After characterizing this device in ultra-high-vacuum, the ion pump was isolated from the rest of the chamber and switched to the turbo pump. Helium was introduced into the vacuum chamber through a needle valve, with the flow rate moderated to set the pressure. I-V characteristics at various pressure have been recorded. Figure 7-6 shows the operation of the device in 100 mTorr of helium.



Figure 7-7: I-V (A) and FN (B) Characteristics of a 1000×1000 array operated in 500 Torr of helium with $V_A = 19$ V

At pressures greater than $\sim 10^{-7}$ Torr, the emission current begins to be suppressed when compared to the current obtained under UHV. We postulate that this suppression is due to adsorption of helium on the emitter surface, increasing the barrier for emission. Operation of the devices exposed to higher pressures in UHV can cause a subsequent increase in current, similar to the ionizers reported by Fomani et al. [41]. These cathodes have been operated in He pressures up to 500 Torr (Figure 7-7).

When operated in 500 Torr of He, the device characteristics demonstrated significant hysteresis, with the device "turn on" at $V_{GE} \simeq 15$ V and "turn off" at $V_{GE} \simeq 18$ V. This change is too large to be explained by a work function difference. If we assume that the change is only due to work function, the ratio of the work functions between the up sweep and down sweep would be a factor of 3. In addition, the S-K chart shown in the inset of Figure 7-7(B) suggests that the up-sweep characteristic exhibits an smaller effective tip radius compared with the down-sweeps, with β decreasing from 2×10^6 to 8.5×10^5 . It appears that as we drive to higher voltages, the slop of the FN plot changes as the emission current from sharper emitters is quenched, and then only larger emitters contribute to the current in the down sweep. It is possible that higher emission current allows helium to better passivate emission sites, and that there are fewer atomic sites contributing to emission in the smaller emitters. We have not been able to ascertain the exact mechanism of this effect.

At higher pressures, the emission current was completely suppressed and there was no ob-

servable emission current at 1 atmosphere. The recovery of the I-V indicates that it is surface adsorption of gas molecules on the emitter surface that is limiting the emitter performance in high-pressure environments. This adsorption-desorption and its effect on emitter stability and lifetime is the primary challenge for this proposed class of devices, and may be mitigated by novel surface coatings or device architectures.

7.2 Electron-transparent Graphene-Gated FEAs

Based on challenge of gas adsorption on the emitter surface when operating the cathode in poor vacuum conditions, it is attractive to develop a method to isolate the cathode from the ambient atmosphere to ensure that the emitter tips remain in a pristine vacuum environment regardless of the ambient environment. One possible way to do this is to encapsulate the cathode, maintaining a pristine ultra-high-vacuum environment in which the emitters to operate. This structure requires an "electron window" that is transparent to electrons, but is impervious to gas molecules. In practice, this window can be a membrane that that is thin enough to allow electrons of a certain energy to be transmitted and thick enough to prevent the percolation of gas molecules in the opposite direction. There are applications for electron transparent windows in microscopy (particularly for the imaging of biological or other "wet" samples), lithography, x-ray sources, ion sources, electron-beam processing, and lasers.

Previous work on electron-transparent materials can be traced back to Lenard windows on cathode ray tubes [81] and Rutherford's experiments measuring the transmission and scattering of β particles by metal foils reported in 1911 [110]. Groups have utilized metal films such as aluminum-coated mylar [19]. Other groups have studied the use of layered organic/inorganic materials [43], silicon [24], and dielectric materials such as alumina [34], silicon nitride [52], and boron nitride [57].

All of these approaches share several drawbacks. First, these structures require a rigid support structure to maintain the integrity and prevent the membrane from tearing when a large vacuum differential is applied across the material. This support structure is necessarily thick, and reduces the transmission efficiency of the window. Next, the membranes range from tens of nanometers to several microns thick, requiring electrons to have energy of > 1 - 10 keV

to attain adequate transmission through the electron window. An ideal approach would use a low-Z material to reduce nuclear scattering of the electron beam and would be atomically thin, allowing transmission of electrons with energy < 100 eV.

Graphene, an allotrope of carbon which has garnered lots of publicity recently as an interesting electronic and mechanical material, can be as thin as a single atomic layer. It is a single sheet of graphite, i.e. carbon atoms arranged in an hexagonal pattern and has excellent mechanical properties and thermal properties. The atomic layer thickness (~ 5 Å)and the low atomic number of carbon (Z = 6) guarantees that the scattering cross-section will be small even for low energy electrons. Previous experimental work performing electron holography of graphene has shown that graphene can be transparent to electrons with energy > 100 eV, absorbing or backscattering $\approx 27\%$ of the incoming electrons per layer at 66 eV [84]. Following [91], an effective attenuation length (l_{EAL})of graphene can be modeled theoretically or measured experimentally, and transmission can be modeled as an evanescent wave:

$$I_{transmitted} = I_0 \exp\left(-\frac{t}{l_{EAL}}\right) \tag{7.8}$$

At low energies, this effective attenuation length (l_{EAL}) has been shown to be on the order of 5 Å[77]. Graphene has also been previously demonstrated to work as an electron-transparent gate electrode for field emission devices, demonstrating transmission through the triode at ~ 400V [82].

Important for this application is that an atomic layer of graphene has been shown to be impermeable to gas molecules as small as helium [17]. From the measured leak-rate data reported in [17], they have estimated the upper bound of the transmission probability of a helium atom impinging on a graphene membrane to be 10^{-11} . The pressure on either side of the membrane equilibrates in a time on the order of hours or days, and the authors speculate that that it is due to leakage at the graphene-SiO₂ interface, or diffusion of He through the SiO₂.

Given the tensile strength of graphene (~ 130 GPa [80]), it thus possible to have ultra high vacuum on one side of a atomic layer graphene membrane which has the electron emitter and on the other side have one atmosphere (or greater) of ambient gasses or others. Corroborating this, the results in [17] indicate 4.75 μ m × 4.75 μ m microchamber sealed with a single layer of



Figure 7-8: Process for transferring graphene from Cu catalyst foil to a device substrate using PMMA. Figure from [78].

graphene was able to withstand several atmospheres of pressure without failure. Above several atm, the failure mechanism was slipping of the graphene film, indicating that perhaps with a better clamping mechanism, an even larger pressure differential can be maintained.

We conducted experiments in which graphene was synthesized by hot-walled thermal chemical vapor deposition (CVD), as similarly reported in detail in [6] 5 sccm CH₄ (99.5% purity) was introduced to a 500-nm-thick physical vapor deposited Cu (and Ni) catalyst on 200 nm thermally oxidized Si, at 1000° C in an Ar: H₂ ballast (960 (99.9997%): 40 (99.9992%) sccm, at 25 mbar. Following the growth phase, samples were quenched using 2000 sccm N₂ (99.99%) to 250° C prior to system venting. The graphene was transferred to our field emitter as shown in Figure 7-8.

To transfer the graphene, PMMA (an electron-beam resist) was spun onto the graphene. By etching the copper catalyst layer in ferric chloride (i.e. PCB etchant), we transferred the graphene layer to the substrate, on top of the polysilicon gate of our our silicon field emitter. The PMMA was dissolved in acetone, leaving the graphene layer in intimate (and electrical) contact with our polysilicon gate. The final structure is shown in Fig 7-9. It should be noted that the tip is recessed from the top of the poly-silicon gate aperture by about 50 - 75 nm.

Figure 7-10 shows the I-V and F-N characteristics for a 100×100 array that has been



Figure 7-9: Structure of a field emitter device with an electron-transparent graphene gate.



Figure 7-10: I-V (left) and FN Characteristics (right) of a 100×100 array with a graphene, electron transparent gate electrode.

covered with single layer graphene. These results demonstrate that the graphene gate layer is transparent to electrons emitted by the tip. The current shown in Fig. 7-10 is collected by the anode placed about 1 cm above the silicon field emitter array with graphene gate layer.

Generally, 100 \times 100 arrays turn on at V_{GE} < 20 V. With graphene on the sample, the
turn-on voltage has shifted to ≈ 40 V. While the graphene could change the electrostatics, it appears that energy transmission through atomically thin graphene sheets is limited by the "bulk" (in as much as an atomically thin layer of material can have a bulk, at least) plasmon excitations in the graphene. In plasmon interactions, the electron excites charge density waves in the ensemble of loosely bound electrons within the graphene. Plasmon interactions are the dominant inelastic scattering mechanism for electrons in solids and typically involves energy losses of 10 to 30 eV [10] hence, the fact that our device seems to strongly turn on between 30 and 40 eV is consistent with plasmon excitation limiting transmission through the graphene at lower energies. This sharp turn on is hence likely more due to the transmission probability through the graphene rather than emission from the tip. The transmission probability manifests itself as an effective higher $b_{FN} \approx 1000 \text{ V}^{-1}$ vs. 300-700 for typical 100 × 100 arrays.

The process is not yet perfected, and one of the consequences is that the gate leakage current is several orders of magnitude higher than the emission current. The high gate leakage current is probably due to the fact that the graphene is unpatterned, and thus shorts to the substrate in areas outside of the active device region (e.g. along the periphery of the substrate). A way to improve this process is to perform photolithography (electron-beam lithography) of the graphene after transferring to the target substrate. Because of the challenges inherent in performing 2-D transfers on large substrates, it may be better to transfer on small pieces, and *after* field emission characterization, to compare the performance of arrays before and after graphene transfer.

Graphene is readily etched in an oxygen plasma, allowing for an simple method for patterning the graphene after lithography to define the regions where the graphene will remain. A solvent clean can remove the resist after the O_2 plasma etch, which should not damage the graphene layer. It is also possible that subsequent annealing of the graphene after the transfer and solvent processing could improve the graphene, or alternative polymer scaffolds yield better quality graphene [127].

There are three potential approaches for using a graphene membrane to protect field emitter tips from their environment, summarized in Figure 7-11. The first, explored above, is where the graphene membrane is in intimate contact with the gate, so that each emitter tip is in its own microenvironment. The second architecture is where the graphene is used as a transparent anode, used to separate the entire cathode from the ambient environment. In this case, the



Figure 7-11: Three different approaches for using graphene to separate field emitter tips from there operating envrionment

graphene would be separated some distance from the emitter tips, and a larger volume would be maintained under UHV. The third — and most complex — architecture is similar to the transparent graphene anode, but includes beam-forming electrodes to form a focused or collimated electron beam in the ambient environment. Further work needs to be done to test the viability of each of the architectures.

7.3 Summary

In this chapter we reported experiments to explore extending the operation of field emitter arrays to challenging applications through the operation in high pressures and using an electrontransparent graphene gate. Both of these experiments show promise, however, more work needs to be done to make both technologies higher performance.

The ability to operate field emission cold cathodes in a variety of environments – not just ultra-high vacuum – has considerable implications. First, the cost, size, weight, and power of the systems that they are operated in could be significantly reduced because the need for large, inefficient vacuum pumps and vacuum chambers is lessened. We can also think about using these cathodes for novel applications such as environmental SEM, food processing, and high throughput processing of semiconductor wafers.

To improve the emission characteristics of emitters in helium, surface coatings that prevent the adsorption of gasses on the emitter surface, while still allowing for electron emission. It could be that higher work function materials could limit the reactivity of the emitter surface and help prevent gas adsorption, however, the emission current will be reduced correspondingly. The operation should be extended to 1 atmosphere.

The electron-transparent graphene gate is exciting, as it is the lowest gate-emitter voltage reported for an electron-transparent triode. Several challenges, particularly the leakage current of the graphene gate needs to be addressed with careful processing. Sealing techniques for thin graphene membranes should be explored. Further analysis of the transmission of electrons through graphene, and experiments measuring the inelastic scattering mechanisms of electrons in graphene at low energies should be performed.

Chapter 8

Summary and Future Directions

8.1 Summary

This thesis reported the design, fabrication, modeling, and characterization of silicon field emitter arrays with self-aligned gates and integrated silicon nanowire current limiters. First, detailed modeling and characterization of the silicon nanowire current limiter was performed, including the role of surface states and recombination in determining the I-V characteristics of the current limiter, the thermal failure and breakdown of the nanowire, and 3-D effects of building arrays of silicon nanowires. This modeling suggests that for high current operation, the Si nanowire must have higher doping and/or larger radius than previously expected to prevent full-depletion due to the silicon interface. Ungated arrays with high-aspect-ratio current limiters verified that the silicon nanowire successfully limits the current to individual emitters, evidenced by the F-N plot deviating from its linear relationship.

Exploratory experiments in filling the voids between high-aspect-ratio silicon pillars revealed that it was a significant materials and processing challenge, and extensive work was done to develop a process for the void-free, filling of these gaps with a high quality dielectric material that could support the polysilicon gate. A process that used a combination of undoped polysilicon deposition and oxidation, as well as low-stress silicon nitride was presented, as well as a method that used selective etching of silicon nitride over silicon dioxide to form self-aligned gate apertures. These exploratory experiments yielded functional field emitter arrays, demonstrating a turn-on voltage ≈ 20 V and current saturation at 30 V, however, these devices suffered

from large gate-substrate leakage current. To address this leakage current, a new device architecture was proposed that incorporates a thick dielectric surrounding the array and probe pads outside the active array.

These new devices exhibit record current density for Si field emitter arrays, with sustained currents > 1 μ A/tip and current densities > 150 A/cm² @ V_{GE} < 70 V. A number of different-sized arrays were measured, showing that current scales by the number of emitters in the array, suggesting both that the emitter tip radius distribution is quite uniform, and that the current limiter is preventing the destruction of sharper emitters. Lifetime characterization demonstrated > 100 hours of operation for a single emitter operating at 1 μ A (for an equivalent array current of 100 A/cm²) and of > 100 hours for a 10 × 10 array operating at 10 A/cm². The emission characteristics degraded during this time however, by allowing the device to relax, the original characteristics could be recovered.

The new devices are suitable for a variety of different applications where high performance cold cathodes are required, and we conducted some experiments to try to extend their operation to areas where high-vacuum is no-longer required. We demonstrated that graphene is an interesting material for impermeable, electron transparent windows, and that 40 eV electrons can penetrate graphene. We also demonstrated field emitter arrays that operate in up to 500 Torr of helium at voltages under 19 V, the first ionization potential of He.

8.2 Significant Contributions of This Work

Here is a brief list of the main contributions of this work:

- Development and demonstration of an original process for the fabrication of silicon field emitter arrays with self-aligned gate and integrated silicon nanowire current limiters. As executed, this process produced current limited field emitter arrays with 100× higher packing density than arrays with current limiters previously reported, and are the first reported arrays to combine both individual current limiters and self-aligned gates.
- 2. Demonstration of field emitter arrays that demonstrate operation at > 150 A/cm² @ 1.5 mA of total current. These are the highest reported Si field emitter array current

densities and represent a > $10 \times$ improvement over state-of-the-art semiconductor field emitter arrays.

- 3. Demonstration of single tip field emission current from silicon FEAs > 2.5 μ A.
- 4. Field emitter arrays that demonstrate > 100-hour lifetimes at a current density > 100A/cm²,
 > 100-hour lifetimes at a current density > 10 A/cm², and > 320 hours at 0.1 A/cm²,
 suggesting that the current limiter addresses failure mechanisms of field emitter arrays.
- 5. Development of a new technique to electrically estimate the tip radius statistics of field emitter arrays from a series of I-V characteristics.
- 6. Study of the beam divergence of silicon field emitter arrays with current limiters and self-aligned gate, demonstrating $\simeq 12^{\circ}$ angular divergence.
- Study of the electron transparency of graphene, demonstrating that at voltages as low as 40 V, there is transmission of electrons through graphene.
- 8. Demonstration of turn on-voltages as low as 8.5 V, enabling operation of Silicon field emitter arrays in up to 500 Torr of He. The low turn-on voltage makes it possible to keep all voltages lower than the first ionization potential of He and prevent ionization of the gas and arcing. Devices recovered when re-introduced to UHV and operated at low pressure, indicating surface absorption is causing emission current degradation at high pressures.
- 9. Developed an approach for self-consistent device simulations of a single emitter were developed that agree with measured I-V characteristics, using only doping density and tip radius as fitting parameters.
- 10. Explored the effect of operating the current limiters in an array instead of as isolated structures.

8.3 Future Directions

The gated field emitter arrays with integrated nanowires could be further optimized and improved. Perhaps first they should be improved by extending their operation to higher voltages. Typically devices exhibited failure between 60 and 75 V. A systematic study of the reliability of the dielectric, including time-dependent dielectric breakdown studies (TDDB) [87] should be performed, and this study used to optimize the structure and extend operation to higher voltages. An obvious first step is to reproduce the fabrication with careful optimization of the mesa etch to ensure that a sharp ridge does not form around the perimeter of the active area. Second, the dielectric surrounding the arrays could be thickened from 2 to 3 - 5 microns, increasing the breakdown voltage of the device. A better understanding of vacuum breakdown could also help extend field emitter operation to higher voltages without failures.

Now that the modeling of silicon nanowire current limiters has been improved — examining both individual nanowires and the effects of putting them in an array — the models should be coupled and turned into framework to model field emitter arrays with integrated current limiters. As part of this, measuring the interface trap density and recombination velocity of the SiNW surface should be performed. These experiments could include pump-probe techniques [53]. This modeling could better inform field emitter array designers and enable better optimized field emitter arrays with nanowire current limiters.

Of course, the arrays could be further scaled. Gated field emitter arrays with pitch as small as 200 nm have been reported [102]. If these field emitter arrays with Si nanowire current limiters were shrunk to 200-nm pitch, and the current/tip maintained at 1 μ A, the current density could be increased to > 2000 A/cm². At this scale, it is likely that a more precision deposition technique such as atomic layer deposition will need to be employed to fill in the very small gaps between naowires. Assuming a nanowire diameter of ~ 10nm, height of 1 μ m and moderate D_{it} , it is likely that a high doping density of $10^{18} - 10^{19}$ cm⁻³ will need to be used to prevent the nanowire from being completely depleted.

These devices are approaching the performance required to make them an electron source for systems that require $\approx 100 \text{ A/cm}^2$ current density and $\approx 1 \text{ mA}$ circuit current, such as the 0.85 THz amplifier reported in [8]. In order to make these arrays implementable in systems, a packaging technology needs to be developed in order to create electrical contact to the gate and the substrate while allowing the beam to be injected into the input of the system.

A systematic study of the operation of cathodes in a high pressure, inert gas ambient, including potential surface coatings to prevent adsorbed gas molecules from quenching field emission should be performed. Electron transparency of graphene at low beam energy should be conducted, and work on developing architectures that separate the cathode from the ambient environment should be performed. The graphene-encapsulated cathode could enable a longlife electron impact ionizer in systems such as portable mass spectrometers. Work should be continued to explore the use of this device architecture in ionization experiments.

Appendix A

Table of Fundamental Physical Constants

Physical Constant	Symbol	Value		
		SI units	"microelectronic" units	
Boltzmann constant	k _B	$1.38 \times 10^{-23} \text{ J/K}$	$8.62 \times 10^{-5} \text{ eV/K}$	
Electron charge	9	$1.60 \times 10^{-19} \text{ C}$	$1.60 \times 10^{-19} = 1 \ e$	
Electron rest mass	m_0	9.11 × 10 ^{−31} kg	$5.69 \times 10^{-16} \text{ eV} \cdot \text{s}^2/\text{cm}^2$	
Planck constant	h	$6.63 \times 10^{-34} \text{ J} \cdot \text{s}$	$4.14 \times 10^{-15} \text{ eV} \cdot \text{s}$	
	カ	$1.05 \times 10^{-34} \text{ J} \cdot \text{s}$	$6.58 \times 10^{-16} \text{ eV} \cdot \text{s}$	
Speed of light in vacuum	С	3.00×10^8 m/s	$3.00 \times 10^{10} \text{ cm/s}$	
Permittivity of vacuum	ϵ_0	8.85×10^{-12} F/m	8.85×10^{-14} F/cm	

Conversion between both systems:

$$1 \text{ eV} = 1.60 \times 10^{-19} \text{ J}$$

1 kg = $6.24 \times 10^{14} \text{ eV} \cdot \text{s}^2/\text{cm}^2$

From [119]

Appendix B

Table of Important Material Parameters of

Si and GaAs at 300 K

Physical Parameter	Sym	Si Value	GaAs Value	units
lattice constant	a	0.543	0.565	nm
interatomic distance		0.235	0.245	nm
atomic density	N_{a}	5.0×10^{22}	4.4×10^{22}	g
density	d	2.33	5.32	g·cm ^{−3}
linear thermal expansion coefficient	α	2.59×10^{-6}	5.73×10^{-6}	K ⁻¹
relative dielectric constant	e,	11.7	12.9	-
electron affinity	χ	4.04		eV
bandgap energy	E_{g}	1.124	1.422	eV
DOS electron effective mass	m_{de}^{*}	$1.09 m_0$	0.066 m ₀	eV·s ² /cm ²
DOS hole effective mass	m_{dh}^*	$1.15 m_0$	$0.52 m_0$	eV·s²/cm²
conduction band effective DOS	$\tilde{N_c}$	2.86×10^{19}	4.21×10^{17}	cm ⁻³
valence band effective DOS	N_{ν}	3.10×10^{19}	9.51×10^{18}	cm^{-3}
intrinsic carrier concentration	n _i	1.07×10^{10}	2.25×10^{6}	cm ⁻³
optical phonon energy	Eopt	0.063	0.035	eV
conductivity electron effective mass	m [*] _{ce}	$0.28 m_0$	$0.070 \ m_0$	eV·s ² /cm ²
conductivity hole effective mass	m_{de}^*	$0.41 \ m_0$	$0.44 m_0$	eV·s ² /cm ²
phonon-limited electron mobility	μ _e	1430	8000	cm²/V⋅s
phonon-limited hole mobility	μ_{b}	480	320	cm²/V⋅s
electron saturation velocity	v _{esat}	1.0×10^{7}	$1.0-1.5 \times 10^7$	cm/s
hole saturation velocity	v _{hsat}	6.0×10^{6}		cm/s
optical G/R rate coefficient	r _{rad}	2.0×10^{-15}	7.2×10^{-10}	cm ³ /s
electron-electron Auger coefficient	r _{eeb}	1.8×10^{-31}	1.8×10^{-31}	cm ⁶ /s
hole-hole Auger coefficient	r _{ehh}	9.5×10^{-32}	4.0×10^{-30}	cm ⁶ /s
impact ionization threshold energy	E_{ii}	1.12	1.72	eV

From [119]

Appendix C

Table of Important Material Parameters of SiO_2 and Si_3N_4 at $300\,{\rm K}$

Physical Parameter	SiO ₂ Value	Si ₃ N ₄ Value	units
Structure	Amorphous	Amorphous	
Melting Point	≈ 1600	-	°C
Density	2.2	3.1	g/cm ³
Refractive Index	1.46	2.05	
Dielectric Constant	3.9	7.5	
Dielectric Strength	107	107	V/cm
Infrared Absorption Band	9.3	11.5 - 12.0	μm
Energy Gap	9	5	eV
Thermal Expansion	5×10^{-7}	-	°C ⁻¹
Thermal Conductivity	0.014	-	$W cm^{-1} K^{-1}$
DC Resistivity			
@25 °C	$10^{14} - 10^{16}$	1014	Ω· cm
@500 °C		$\approx 2 \times 10^{13}$	Ω· cm

Appendix D

Tunneling Through a Surface Barrier

The one-dimensional problem of an electron tunneling through a sharp triangular barrier can be readily solved by applying the Wentzel-Kramers-Brillouin (WKB) approximation to estimate the transmission probability, $T(E_x)$, through the barrier. For a single electron with energy in the direction of the barrier E_x , the one-dimensional time-independent Schrödinger equation is:

$$E_{x}\psi(x) = -\frac{\hbar^{2}}{2m}\frac{d^{2}}{dx^{2}}\psi(x) + V(x)\psi(x)$$
(D.1)

The tunneling distance, W, depends on both the energy of the electron as well as the applied electric field, F. In this section, the vacuum level at x = 0 is used as the energy reference.

$$W(E_x) = \frac{\phi + E_F - E_x}{qF} \tag{D.2}$$

First consider a particle incident traveling in the +x direction incident on a barrier with wavefunction ψ_0 . The particle inside of the barrier can be modeled as an evanescent wave and ψ inside of the barrier can be approximated as decaying as $\psi_0 e^{-\alpha x}$. From this model, an expression for the fraction of the wave that is transmitted through the barrier, i.e. the transmission probability is obtained:

$$T(E_x) = \frac{\psi(W)}{\psi(0)} \approx e^{-\alpha W}$$
(D.3)

If instead of a square barrier, the barrier height varies as a function of position, the WKB approximation can be used to calculate the transmission probability. Essentially, the WKB approximation breaks up the barrier into a number of barriers win infinitesimal width and calculates the transmission through each. Mathematically, the WKB approximation states:

$$T(E_x) \approx e^{-2\int_{x_1}^{x_2} k(x) \, \mathrm{d}x} \tag{D.4}$$

where x_1 and x_2 are the classical turning points. The x-directed wave vector inside the classically forbidden region, k(x), from the Shrödinger equation, is

$$k(x) = \sqrt{2m[V(x) - E_x]/\hbar^2}$$
 (D.5)

From this equation, the relation between α and k can be found.

$$\alpha W = \int_{x_1}^{x_2} k(x) dx \tag{D.6}$$

Thus, the transmission probability $T(E_x)$ is:

$$T(E_x) \approx \exp\left[-2\int_{x_1}^{x_2} \sqrt{2m[V(x) - E_x]/\hbar^2} \, dx\right]$$
 (D.7)

Where the potential difference $V(x) - E_x = -qFx + \phi + E_F - E_x$. From the potential V(x) = -qFx, the limits of the classically forbidden region are $0 \le x \le \frac{\phi + E_F - E_x}{qF}$, giving the limits of the integration. The resulting equation is:

$$T(E_x) \approx \exp\left[-2\int_0^W \sqrt{\frac{2m}{\hbar^2}} \sqrt{-qFx + \phi + E_F - E_x} \, dx\right]$$
(D.8)

The equation can be readily integrated to find:

$$T(E_x) \approx \exp\left[-\frac{4}{3}\sqrt{\frac{2m}{\hbar^2}}\frac{(\phi + E_F - E_x)^{3/2}}{qF}\right]$$
 (D.9)

Alternatively, by noting that apart from the factor of $\sqrt{2m/\hbar}$, this area resembles a triangle with base $(\phi + E_F - E_x)/qF$ and height of $\sqrt{\phi + E_F - E_x}$, the same result can be obtained.



Figure D-1: Tunneling probability through a triangular barrier shows an exponential dependence on electric field. The electron has energy normal to the surface $E_x = E_F$. The dashed line shows where significant tunneling occurs, at 2 V/nm.

Because electrons inside the metal obey Fermi-Dirac statistics and the metal is assumed to be at 0 K, we can assume that there are no occupied states with energy greater than the Fermi energy. In addition, for energies below the Fermi energy, the transmission probability is vanishingly small because. Thus, we can make the approximation that $E_x \approx E_F$, simplifying equation D.9 to:

$$T(E_x) \approx \exp\left[-\frac{4}{3}\sqrt{\frac{2m}{\hbar^2}}\frac{\phi^{3/2}}{qF}\right]$$
 (D.10)

From the above, the tunneling probability has exponential dependence on the applied electric field, as well as the barrier height. Assuming that the electron with x-directed energy E_x equal to the Fermi energy, the height of the barrier is the work function, ϕ , for n-type Si (~ 4.04 eV). A quick calculation can be performed to understand the field required for significant electron emission to occur. Figure D-1 shows the relationship of the tunneling probability to the electric field. There begins to be significant tunneling probability at roughly 2 V/nm (2 × 10⁷ V/cm), indicating that the width of the potential barrier must be less than approximately 2 nm. This thickness is on the order of several electron wavelengths, which is the same result that is obtained for significant transmission of an evanescent wave.

A more careful calculation of the tunneling probability was performed by Fowler and Nord-

heim [42] with the result:

$$T = \frac{4\sqrt{E_x(\phi + E_F - E_x)}}{\phi + E_F} \exp\left[-\frac{4}{3}\sqrt{\frac{2m}{\hbar}}\frac{(\phi + E_F - E_x)^{3/2}}{qF}\right]$$
(D.11)

When considering many electrons, it is more appropriate to consider tunneling current density rather than tunneling probabilities. The tunneling current density can be expressed as the tunneling probability multiplied by the differential arrival rate (flux of electrons per unit energy) $N(E_x)$, called the supply function, and then integrated from $-\infty$ to E_F :

$$J_{tnl} = q \int_{-\infty}^{E_F} T(F, E) \cdot N(E) dE$$
(D.12)

where:

$$N(E_x) = \int v(E_x)g(E)f(E_x)$$
(D.13)

Here, $v(E_x)$ is the x-velocity of the electrons, g(E) is the density of states (in p-space), and $f(E_x)$ is the Fermi-Dirac distribution. For a 3-D electron gas, the density of states will be:

$$g(E) = \frac{2}{h^3} dp_x dp_y dp_z$$
(D.14)

and the supply function integral is:

$$N(E_x) = \int_{p_y, p_z} \frac{p_x}{m^*} \cdot \frac{2}{h^3} dp_x dp_y dp_z \cdot \frac{1}{1 + \exp(\frac{E_x - E_F}{k_B T})}$$
(D.15)

Performing a change of variables and evaluating this integral in cylindrical coordinates results in:

$$N(E_{x}) = \frac{4\pi m^{*} k_{B} T}{h^{3}} \ln\left(1 + e^{\frac{E_{T} - E_{x}}{k_{B} T}}\right)$$
(D.16)

Putting equations D.16 and D.11 together into D.12 yields an approximate expression for the tunneling current.

$$J_{tnl}(E_{x}) dE_{x} = q \frac{16\pi m^{*} k_{B} T \sqrt{E_{x}(\phi + E_{F} - E_{x})}}{h^{3}(\phi + E_{F})} \ln\left(1 + e^{\frac{E_{F} - E_{x}}{k_{B} T}}\right)$$
(D.17)
$$\cdot \exp\left[-B \frac{(\phi + E_{F} - E_{x})^{3/2}}{qF}\right] dE_{x}$$

where

$$B = \frac{8\pi\sqrt{2m^*}}{3h} = 6.87 \times 10^7 \text{ cm}^{-1} \cdot \text{eV}^{-1/2}$$
(D.18)

At moderate temperatures, the following simplification can be made in the calculation of the supply function:

$$k_B T \ln\left(1 + e^{\frac{E_F - E_x}{k_B T}}\right) \cong \begin{cases} k_B T e^{\frac{E_F - E_x}{k_B T}}, & \text{for } E_x > E_F \\ E_F - E_x, & \text{for } E_x \le E_F \end{cases}$$
(D.19a)

The simplification results in the tunneling current becoming:

$$J_{tnl}(E_x) dE_x = - \frac{16\pi m^* (E_F - E_x) \sqrt{E_x(\phi + E_F - E_x)}}{h^3(\phi + E_F)} \cdot \exp\left[-B \frac{(\phi + E_F - E_x)^{3/2}}{qF}\right] dE_x \qquad (D.20)$$

for $E_x > E_F$, and:

$$J_{tnl}(E_x) dE_x = q \frac{16\pi m^* k_B T \sqrt{E_x(\phi + E_F - E_x)} e^{(E - E_F)/k_B T}}{h^3(\phi + E_F)} \cdot \exp\left[-B \frac{(\phi + E_F - E_x)^{3/2}}{qF}\right] dE_x \quad (D.21)$$

for $E_x \leq E_F$. Because this function is peaking at $E_x = E_F$, the approximation that $E_F - E_x \ll \phi$ may be made. In addition, by using the approximation:

$$(B/F)(\phi + E_F - E_x)^{3/2} = (B/F)\phi^{3/2}(1 + (E_F - E_x)/\phi)^{3/2} \cong (B/F)\frac{3}{2}\phi^{1/2}(E_F - E_x) \quad (D.22)$$

which is valid because $(E_F - E_x)/\phi \ll 1$ and results in an integral of the form $-\int y e^{\epsilon y} dy$. The modified version of equation D.20 shown below:

$$J_{tnl} = q \frac{16\pi m^* \sqrt{\phi E_F}}{h^3(\phi + E_F)} \exp\left[-B \frac{\phi^{3/2}}{qF}\right] \int_{-\infty}^{E_F} \exp\left[\frac{3}{2}B\phi^{1/2}(E_F - E_x)/qF\right] (E_F - E_x) dE_x$$
(D.23)

Performing the integration, we finally arrive at the Fowler-Nordheim model without image correction in 3-D:

$$J_{tnl} = q \frac{4}{3} \frac{16\pi m^* \sqrt{\phi/E_F}}{h^3 B^2(\phi + E_F)} F^2 \exp\left[-B \frac{\phi^{3/2}}{qF}\right]$$
(D.24)

Appendix E

Donor Ionization Efficiency in Silicon Nanowires

A surprising consequence of the nanowire geometry is that the assumption that all of the donor atoms inside the silicon are ionized is no longer valid, and must be checked. This is because in a bulk silicon lattice, the potential of the ionized impurity is screened by the charge carriers in silicon. In a nanowire, the volume surrounding the impurity is reduced, the medium that the impurity "sees" in no longer isotropic, and an image charge can form at the semiconductor/dielectric interface. The new ionization energy, with respect to the conduction band, can be found through the calculation given by Diarra et al. [29, 28], assuming that the donor atom is sitting directly in the center of the nanowire (the best case scenario):

$$E_{I} = E_{I}^{0} + \frac{2}{\epsilon_{Si}r} \frac{\epsilon_{Si} - \epsilon_{ox}}{\epsilon_{Si} + \epsilon_{ox}} F(\epsilon_{Si}/\epsilon_{ox})$$
(E.1)

where r is the nanowire radius (in nm), ϵ_{Si} and ϵ_{ox} are the relative dielectric constants of Si and SiO₂, respectively, E_I^0 is the ionization energy for the donor in eV ($E_I^0 = 45$ meV for phosphorus donor atoms in silicon) and F(x) is the following function in (nm·eV):

$$F(x) = \frac{0.0949x^3 + 17.395x^2 + 175.739x + 200.674}{x^2 + 50.841x + 219.091}$$
(E.2)

From the modified ionization energy, we can then calculate the density of the ionized impurities as:

$$N_{D}^{+} = \frac{N_{D}}{1 + g_{d} \exp\left(\frac{E_{F} - E_{C} - E_{I}}{kT}\right)}$$
(E.3)

Where g_d , the degeneracy of donor states, is equal to 2 from spin degeneracy [119]. In a silicon nanowire, we can assume that the charge neutrality condition still holds, i.e.

$$N_D^+(E_F) + p(E_F) - n(E_F) - N_A^-(E_F) = 0$$
(E.4)

where:

$$\exp\left(\frac{E_F - E_C}{kT}\right) = \frac{n}{N_C} \tag{E.5}$$

and assuming that in the case of an n-doped semiconductor $N_A^-(E_F) \approx 0$ and $p(E_F) \approx 0$

$$\frac{N_D}{1 + 2\frac{n}{N_C} \exp\left(\frac{E_I}{kT}\right)} = n \tag{E.6}$$

where N_C is the effective density of states in the conduction band of Silicon, and is unchanged from the bulk value for nanowires larger than ≈ 10 nm. Solving for *n*, and assuming that $n = N_D^+$, we find:

$$n = N_D^+ = \frac{N_c}{4} e^{-E_I/kT} \left(-1 + \sqrt{1 + 8\frac{N_D}{N_C} e^{E_I/kT}} \right)$$
(E.7)

Putting these all together, Figure E-1 shows the number of ionized donors and the ionization efficiency (N_D^+/N_D) for a silicon nanowire buried in SiO₂. From the graph, we can assume that the donors are completely ionized only until about a doping of 10¹⁷ for a 100-nm diameter nanowire. For the nanowire current limiters explored in this dissertation, this effect does not appear to be a relevant concern, however, for smaller nanowires or nanowires with higher doping densities, care should be taken to explore this phenomenon.

It is obvious from equation E.1 that the ratio of dielectric constants between the nanowire material (ϵ_{Si}) and the surrounding material (ϵ_{ox}) have a large impact on the incomplete ionization in the nanowire in that the higher the dielectric constant, the lower the effect. Diarra et al. have suggested that a way to mitigate this effect is to bury the silicon nanowire in a high-*x*



Figure E-1: (A) Ionization Efficiency and Ionized donor concentration vs. Net Doping for a 100-nm diameter silicon nanowire in oxide (B) Ionization efficiency for several different doping densities for vs. nanowire diameter

dielectric such as HfO₂. Another option is to use a metal gate around the nanowire to provide better electrostatics and screen the ionized impurities better.

Appendix F

Modeling of Surface States in Si Nanowire Current Limiters

Two non-idealities that effect the electrical characteristics and ultimate performance of silicon nanowires are the effect of the Si/SiO_2 interface along the surface of the nanowire and the incomplete donor ionization that results in silicon nanowires. This section will focus on the impact of interface states on silicon nanowire current limiter performance. For a discussion of incomplete donor ionization, please see Appendix E. As the size of the nanowire shrinks, the surface-to-volume ratio increases, and surface effects can become a limiting factor to performance. These surface effects need to be properly understood in order to build a model for the silicon nanowire as a current limiter. We will start by examining the effect of charges and traps at the Si/SiO₂ interface.

As in the case of a MOSFET, in a silicon nanowire device there are four types of charges or traps that can have an impact on device performance, using the framework developed by Deal [25]. These are [119]:

1. Interface traps of density D_{it} , and trapped charges Q_{it} which are located at the Si/SiO₂ interface with energy states within the silicon forbidden bandgap and can exchange charges with silicon in a short time. D_{it} has units ev⁻¹cm⁻². The interface charge density, Q_{it} , is determined by the Fermi level, so it can change with bias. Interface traps can arise from Si dangling bonds, broken Si-H bonds, impurities, excess oxygen, and excess

silicon (trivalent) silicon.

- 2. Fixed oxide charges, Q_f , which are located at or near the interface and are immobile under an applied electric field.
- 3. Oxide trapped charges, Q_o . These traps are distributed inside the oxide layer.
- 4. Mobile ionic charges, Q_m , such as sodium ions. These mobile within the oxide under bias-temperature stress conditions.

Figure F-1 shows a simple schematic depiction of the state of the Si/SiO₂ interface after a hydrogen anneal, such as a 5% $H_2:N_2$ forming gas anneal at ~ 400 °C. This anneal allows hydrogen to diffuse through the dielectric layer and satisfy dangling bonds at the Si/SiO₂ interface. An anneal such as this can reduce the defects at this interface from about 10¹⁵ cm⁻² to about 10¹⁰ cm⁻².

This dissertation will focus mainly on the effect of D_{it} and Q_{it} on device performance. Q_f is always positive and will act to accumulate the surface of the silicon nanowire, increasing



Figure F-1: Schematic depiction of the state of the Si/SiO_2 interface. Dangling bonds give rise to trap sites, which can accept electrons and give rise to interface charge. An anneal in hydrogen can passivate some of the surface traps sites by allowing hydrogen to bond at the dangling sites, reducing the interface trap density.

current at lower V_{DS} , which does not agree with our experimental results. Q_o and Q_m can deplete the silicon nanowire and impact performance, however through careful processing and the fact that we have a high quality thermal oxide on the surface of the nanowire, we expect that the density of Q_o and Q_m to be low. Further, their effect will be simply to shift the electrostatics of the device, whereas D_{it} can directly address the electrons in the silicon, and effect its density.

 D_{it} is the density of interface trap states as function of E_s , i.e. $D_{it}(E_s)$, where E_s is an energy (in eV) in the bandgap measured with respect to the intrinsic Fermi level at the Si/SiO₂ interface. This is shown schematically in Figure F-2. Typically, D_{it} is relatively constant near the midgap, but it increases as the energy approaches both the conduction band and valence band edges. However, it is possible for deep traps to be present that manifest in the midgap, giving rise to a parabolic increase in the middle of the bandgap. Both cases are depicted in Figure F-2. These states can be integrated to find the interface charge, Q_{it} through:

$$Q_{it} = -q \int_{E_i}^{E_F} D_{it}(E_s) dE$$
 (F.1)

While Figure F-2 shows the states in the center of the gap as being symmetric around midgap, this condition is not necessarily nor often the case. In general, interface traps are amphoteric, meaning that they can act as acceptor-like states or donor-like states depending on their position in the gap and the location of the Fermi energy, E_F . Assuming a Fermi level in the upper half of the bandgap, consistent with n-doped silicon, Then then trap sites between the Fermi energy and the middle of the bandgap are negatively charged, and the trap sites above the Fermi level are neutral.

For materials with a Fermi level below the middle of the bandgap, the converse is true. These site act like donors. That is, the trap sites below the Fermi level are neutral, and the trap sites between midgap and the Fermi level are positively charged. As the Fermi level can move with applied bias, the number of active trap sites can change with applied bias. This is shown schematically in Figure F-3.

Figure F-4 shows a schematic representation of the effect Q_{it} has on the silicon nanowire. The outer part of the nanowire is depleted, however there remains an inner quasi-neutral channel. This can be thought of as reducing the effective diameter of the nanowire so that is is less



Figure F-2: Schematic diagram of a hypothetical silicon nanowire current limiter with interface charge density, Q_{IT} . The interface charge density has the effect of depleting the outer part of the silicon nanowire, even with no voltage applied. If Q_{IT} is too high relative to the doping density and the tip radius, the entire nanowire can be depleted with no potential applied.

than the actual nanowire diameter, *i.e.* $r_{eff} < r_{NW}$. Adapting the analyses presented in [111] and [113], we can model the effect of the interface states, compare it to device data, and put bounds on the density of interface states D_{it} . This work can be used to guide future work so that the doping and nanowire diameter can be chosen so that the nanowire is not fully depleted at $V_{DS} = 0$ V.

Assume that the nanowire is n-type, circular and radially symmetric with radius R. We make the Boltzmann approximation and assume that all of the donors are ionized. The space charge in the semiconductor is given by

$$\rho(r) = q(p(\Phi(r)) - n(\Phi(r)) + N_D) \tag{F.2}$$

where $\Phi(r)$ is the potential difference from the intrinsic Fermi level, E_i .

Thus:

$$n(\Phi) = n_0 \exp\left(\frac{q\Phi(r)}{kT}\right)$$
(F.3)

$$n(\Phi) = p_0 \exp\left(-\frac{q\Phi(r)}{kT}\right) \tag{F.4}$$

where n_0 and p_0 are the intrinsic carrier concentrations. We will use the abrupt depletion approximation, and assume that the silicon is depleted to some radius, r_d :

$$\rho(r) = \begin{cases} 0 & 0 \le r < r_d \\ qN_D & r_d \le r \le R \end{cases}$$
(F.5)

We can solve Poisson's equation in polar coordinates to find the potential, $\Phi(r)$.

$$\nabla^2 \Phi(r) = \frac{1}{r} \frac{\partial}{\partial r} \left(r \frac{\partial \Phi(r)}{\partial r} \right) = -\frac{\rho(r)}{\epsilon_s}$$
(F.6)

This differential equation can be solved through integration and the intermediate steps are shown in Appendix G. The final result is:



Figure F-3: Band diagram of an Si nanowire with no potential applied, showing the outer edge depleted due to interface states. If D_{it} is high enough and the doping low enough, the entire nanowire may be depleted.

$$\Phi(r) = \Phi_0 - \frac{qN_D}{4\epsilon_s} \left(r^2 - r_d^2 - 2r_d^2 \ln\left[\frac{r}{r_d}\right] \right)$$
(F.7)

where Φ_0 is the potential of the neutral semiconductor, i.e.

$$\Phi_0 = k T \ln\left(\frac{N_D}{n_i}\right) \tag{F.8}$$

The surface potential, Φ_s , is then

$$\Phi_s = \Phi(R) = \Phi_0 - \frac{qN_D}{4\epsilon_s} \left(R^2 - r_d^2 - 2r_d^2 \ln\left[\frac{R}{r_d}\right] \right)$$
(F.9)

This equation is transcendental and cannot be solved analytically. However, through the charge neutrality condition, there is a simple numerical way to find the depletion width, r_d .



Figure F-4: Schematic diagram of a hypothetical silicon nanowire current limiter with interface charge density, Q_{it} . The interface charge density has the effect of depleting the outer part of the silicon nanowire, even with no voltage applied. If Q_{it} is too high relative to the doping density and the tip radius, the entire nanowire can be depleted with no potential applied.



Figure F-5: The nanowire in cross section showing the depletion (shaded) and neutral regions due to interface states.

First, make the simplifying approximation that D_{it} is approximately constant over the region of interest, equation F.1 simplifies to:

$$Q_{it} = -q^2 D_{it} \Phi_s \tag{F.10}$$

In equilibrium, the system must remain charge neutral. This implies that the charge in the area of the depletion region must equal the charge on the surface for each radial slice of the semiconductor. That is

$$\pi q N_D (R^2 - r_d^2) + 2\pi R (Q_{it} + Q_f) = 0$$
(F.11)

This model can be simply extended to handle cases where D_{it} is not constant in energy. To do this equation F.11 can be extended to use the general $Q_{it} = \int D_{it}(\Phi_s)$ and solved over the range $0 < r_d < R$. The rest of the discussion, however, will focus on the case of uniform D_{it} . Assuming uniform D_{it} and no fixed charge Q_f , solving eq. F.11 for Φ_s yields:

$$\Phi_{s} = \frac{N_{D}(R^{2} - r_{d}^{2})}{2qRD_{i}}$$
(F.12)

Equations F.9 and F.12 must be consistent, so they uniquely find the depletion width, r_d . In the case where the nanowire is fully depleted, i.e. $r_d \rightarrow 0$, they cannot be solved consistently and equation F.9 is greater than F.12 over the entire domain $0 \le r_d \le R$. In this case with the nanowire fully depleted, the interface charge is fixed at the depletion charge because charge neutrality must hold. A consequence of this is that the surface potential is now fixed to:

$$\Phi_s = \frac{N_D R}{2q D_{it}} \tag{F.13}$$

The potential inside the nanowire must be calculated from this surface potential, as the potential in the center of the nanowire will change from Φ_0 .

$$\Phi(r) = \Phi_s + \frac{qN_D}{4\epsilon_s}r^2 \tag{F.14}$$

Once the potential at each point in the nanowire is calculated, the electron concentration at each point along the nanowire radius is easily calculated using eq. F.3. To find the effective electron concentration, we need to find the average electron concentration by integrating along the radius:

$$n_{eff} = \frac{1}{\pi R^2} \int_0^R n(\Phi) 2\pi r dr$$

$$= \frac{2}{R^2} \int_0^R n_0 \exp\left(\frac{q\Phi(r)}{kT}\right) r dr$$

$$= \begin{cases} n_0 \frac{4kT\epsilon_s e^{q\Phi_s/kT}}{N_D q^2 R^2} \left(1 - \exp\left(-\frac{R^2 N_D q^2}{4kT\epsilon_s}\right)\right) & r_d = 0 \\ n_0 \exp\left(\frac{q\Phi_0}{kT}\right) \left[\frac{r_d^2}{R^2} + \frac{4\epsilon_s kT}{q^2 N_D R^2} \left(1 - \exp\left(\frac{q^2 N_D (r_d^2 - R^2)}{4\epsilon_s kT}\right)\right)\right] & \text{otherwise} \end{cases}$$
(F15)

With an analytical expression for n_{eff} , we can begin to explore how the interface trap density effects the silicon nanowire and make some interesting estimates about the state of silicon nanowires for a variety of different doping densities, diameters, and interface trap density. Figure F-6(A) shows how n_{eff} varies with doping density for a 100-nm diameter NW for a variety of different interface trap densities. A characteristic of these plots is that there is a clear inflection point where the nanowire becomes fully depleted. If the doping is lower than this critical doping, the effective carrier concentration decreases exponentially, approaching n_0 .

This model indicates that for a 100-nm diameter nanowire with a moderate interface trap density of 1×10^{11} cm⁻²eV⁻¹, the doping density must be at least $\approx 10^{16}$ to prevent the nanowire

from being fully depleted while quasi-neutral. It is likely that the actual value of D_{it} of our SiNWs is at least 1×10^{11} cm⁻²eV⁻¹. While high quality planar MOSFETs fabricated on Si with (100) orientation can achieve $D_{it} < 10^{10}$ cm⁻²eV⁻¹ [93]. Due to nano-scale surface roughness induced from the deep-reactive ion etching of the nanowire, and because of the circular cross-section of the nanowire, a superposition of all possible crystal orientations are oxidized. The data shown in Figure F-7(A) indicates that the configuration of the surface (Fig F-7(B)) can change the D_{it} by about an order of magnitude for the same processing conditions.

Figure F-6(B) demonstrates the potential effect of increasing the doping density of the SiNWs to $\approx 5 \times 10^{16}$ cm⁻³, and shows the dependence on the nanowire diameter. The higher



Figure F-6: Simulations of effective electron concentration (n_{eff}) for various nanowire geometries and electrical parameters. (A) n_{eff} vs. N_D for various values of D_{it} . The NW radius was fixed at 50 nm. (B) n_{eff} vs R for various values of D_{it} . The doping was fixed at 5 × 10¹⁵ cm⁻³. (C) n_{eff} vs R for various doping concentrations. D_{it} was fixed at 10^{11} cm⁻³. (D) semilog (left) and linear (right) plots of n/N_D as a function of position along the channel for the various doping densities in (A) for $D_{it} = 10^{11}$ cm⁻² eV⁻¹.



Figure F-7: (A) Midgap interface trap density vs dry oxidation temperature for p and n-type silicon, (100) and (111) orientations. Data are for samples with no hydrogen anneal. Figure reprinted with Permission from [108]. (B) Structure and location of traps (Pb centers) on oxidized silicon wafers of the three primary orientations. Figure reprinted with permission from [104].

doping allows a SiNW with a diameter of 100 nm to support a higher interface trap density of 3×10^{11} cm⁻²eV⁻¹ without being fully depleted. Based on the simple analytical model presented earlier, a device with this geometry should reach $\approx 3.5 \ \mu$ A and begin to saturate at a drain-to-source voltage $V_{DS} < 1$ V with a mobility of 850 cm²/Vs. As a consequence of the higher doping, the output conductance will be larger (≈ 10 nS), however, the device should still allow for more uniform emission and prevent tip failure due to arcing and Joule heating.

For sufficiently low doping, increasing the nanowire diameter will be unable to compensate for the doping. Figure F-6(C) shows effective electron concentration vs. nanowire diameter for a moderate $D_{it} = 1 \times 10^{11} \text{ cm}^{-2} \text{eV}^{-1}$. These corroborate previous results obtained by Y. Niu and L.F. Velsáquez-García with 1 μ mdiameter, 100 μ m tall silicon pillars. When fabricated with 50 $\Omega \cdot \text{cm}$ ($N_D \approx 1 \times 10^{14} \text{ cm}^{-3}$) wafers, test devices had much lower current than was predicted based on results from devices fabricated from wafers with higher doping, analytical models without surface states, and numerical simulations.

To test this models ability to predict device performance, we attempted to fit the temperature dependence of linear conductance extracted from I-V characteristics of a single SiNW


Figure F-8: Comparison of extracted linear conductance, g_{lin} with a numerical model of linear conductance for a SiNW current limiter.

current limiter (see Section 3.4) using this semi-analytical model of surface states. With reasonable values of doping density, nanowire radius, and mobility values from literature, we find a good agreement with the measured characteristics. The model diverges from the measured characteristics at high temperature, possibly due to surface recombination.

Appendix G

Solution of Poisson's equation in cylindrical coordinates for a partially depleted Si nanowire

We begin with Poisson's equation in cylindrical coordinates as given in equation F.6:

$$\nabla^2 \Phi(r) = \frac{1}{r} \frac{\partial}{\partial r} \left(r \frac{\partial \Phi(r)}{\partial r} \right) = -\frac{\rho(r)}{\epsilon_s} \tag{G.1}$$

We assume that the nanowire has radius R. Making the full depletion approximation, we assume that the nanowire has a depletion edge r_d . Thus, we find that the space charge:

$$\rho(r) = \begin{cases} 0 & 0 \le r < r_d \\ qN_D & r_d \le r \le R \end{cases}$$
(G.2)

multiplying both sides by r, and integrating:

$$\int_{0}^{r} \frac{\partial}{\partial r} \left(r \frac{\partial \Phi}{\partial r} \right) = - \int_{0}^{r} r \frac{\rho(r)}{\epsilon}$$
(G.3)

yields:

$$\left(r\frac{\partial\Phi}{\partial r}\right) = \rho(r)\frac{r^2}{2\epsilon}\Big|_0^R = \begin{cases} -\frac{qN_D}{2\epsilon}\left(r^2 - r_d^2\right) & r_d < r < R\\ 0 & r \le r_d \end{cases}$$
(G.4)

dividing both sides by r gives an expression for the electric field in the nanowire, E(r):

$$\frac{\partial \Phi}{\partial r} = -E(r) = \begin{cases} -\frac{qN_D}{2\epsilon} \left(r - \frac{r_d^2}{r}\right) & r_d < r < R\\ 0 & 0 \le r \le r_d \end{cases}$$
(G.5)

Integrating both sides gives the potential:

$$\int_{0}^{r} \frac{\partial \Phi}{\partial r} = -\frac{qN_{D}}{2\epsilon} \int_{0}^{r} \left(r - \frac{r_{d}^{2}}{r}\right)$$
(G.6)

Solving the integral with the boundary condition that at r = 0, $\Phi(0) = \Phi_0$, the quasi-neutral potential, gives:

$$\Phi(r) = \begin{cases} \Phi_0 - \frac{qN_D}{4\epsilon} \left(r^2 - r_d^2\right) - 2r_d^2 \ln\left(\frac{r}{r_d}\right) & r_d < r < R\\ \Phi_0 & 0 \le r \le r_d \end{cases}$$
(G.7)

and substituting r = R gives an equation for the surface potential:

$$\Phi_s = \Phi(R) = \Phi_0 - \frac{qN_D}{4\epsilon} \left(R^2 - r_d^2\right) - 2r_d^2 \ln\left(\frac{R}{r_d}\right)$$
(G.8)

Appendix H

Process Flow for ungated field emitter arrays with nanowire current limiters

Sten	Tool	Description	Processing Conditions /
Step			Recipe
0		Starting substrate: 150 mm	
		<100> n-Si prime wafer	
1	RCA-ICL	RCA clean prior to tube run	
2	Tube-5C FieldOx	Grow 2500 Åoxide hardmask	1000 C wet oxidation
2	Coater6	HMDS/Coat wafers with	Resist: SPR700 / T1HMDS
5		1.1µm Resist/Prebake	recipe
4	i-stepper	Expose dots	5 μ m pitch 0.6 μ m features
4			Mask designed by Vivi Jayanty
5	Coater6	PEB / Develop / and PB resist	PUDDLE3
		Verify exposure in optical	
6	Metrology	microscope and image dots in	
		SEM	
		Etch oxide dots in	
7	AME5000	$CF_4/CHF_3/Ar$ Chemistry	Baseline ox new, ~ 120 s
		20% overetch	
Q	TAN/AOOD	Rough etch of Si cones using	Nitride etch av 30 s
0	LAIVI490D	SF_6/O_2 chemistry	
9	STS2	Etch nanowires using DRIE	FARDAD recipe, 14 minutes
		Strip oxide and DRIE	>30 minutes in the barrel
10	Asher-TRL	polymer. Must be done	asher
		immediately after etching	
11	Matrology	Measure pillar diameter and	
11	wietrology	tip diameter in SEM	
12	RCA-TRL	RCA clean prior to tube run	

13	Tube-A2	Tip Sharpening / Diameter Shrinking	950 C wet oxidation. Target diameter of pillar: 100nm Target tip size, <10nm. Use process simulator to guide oxidation time
14	Oxetch-BOE	Strip oxide in BOE	etch rate 90 Å/s

Appendix I

Process Flow for gated field emitter arrays with nanowire current limiters

Step	Tool	Description	Processing Conditions/recipe
1	Alignment Marks		
1.1	RCA-ICL	Standard RCA Clean	10 min SC-1 (5:1:1 DI Water: H_2O_2 :NH ₃ at 80°C 1 min HF dip (50:1 DI water:HF) 15 min SC-2 (6:1:1 DI water: H_2O_2 :HCl at 80°C SRD
1.2	Coater6	Coat with 1.1um resist	60s 90C prebake
1.3	Istepper	Photolithography LVL:0	Recipe: T1HMDS Alignment Marks Maskname FEAREV2 Exposure: 160 ms
1.4	Coater6	Develop	Dynamic puddle developer dispense 120C 30s Postbake Recipe: PUDDLE3
1.5	AME5000	STI Etch of alignment marks	40s in Ch.B recipe:STI:CH.B
1.6	LAM490B	Roughen surface to improve contrast	1:00 Black Si Etch
1.7	P10	MEASUREMENT STEP	Measure step height of alignment marks Using profilometer: Nominal result: 5000 A
1.8	ASHER-ICL	Remove resist	3 minutes
1.9	FLX	MEASUREMENT STEP	Measure Wafer bow Used for later stress compensation
2	Mesa Formation and planarization		
2.1	RCA-ICL	Standard RCA clean prior to oxidation	
2.2	Tube-5C	Hardmask oxidation - 2500 A	19 minutes and 4 seconds 1050 C wet oxidation
2.3	UV1280	MEASUREMENT STEP	Measure thickness in ellipsometer
2.4	Coater6	Coat with 1.1um resist	60s 90C prebake Recipe: T1HMDS

2.5	Istepper	Photolithography LVL:01	LVL Name: MESA
			Maskname FEAREV2
			Exposure: 160 ms
26	Castorfe	Davalan	Dynamic puddle developer
2.0	Coalero	Develop	dispense
			120C 30s Postbake
			Recipe: PUDDLE3
2.7	AME5000	Pattern Oxide etch mask	Recipe: Baseline-ox-new
			90s / Endpoint
28	AME5000	Mesa Etch	Etch continues without
2.0	MMEJ000		breaking vacuum
			Recipe: Guerrera deep Si
			575 sec
			Nominally 2um deep trench
2.9	ASHER-ICL	Remove resist	3 minutes
2.10		MEASURMENT STEPS	
a	P10		measure step heigh of trenches
Ь	SEM		image sidewall of trenches
2 11	RCAJCI	Standard RCA clean prior to	
2.11	REAFICE	oxidation	
2.12	Tube-5C	Liner oxidation - 500 A	1000C dry oxidation
			1 hour
213	DCVD	PECVD deposition of oxide	2 microns to front and
2.13			backside
			backside oxide is deposited to
			compensate stress
2.14	Tube-5D	Anneal oxide	950 C 1 Hr O2 Anneal
2.15	DCVD	PECVD deposition of oxide	3 microns to front and
		L	backside (5um total)
			backside oxide is deposited to
			compensate stress
2.16	Tube-5D	Anneal oxide	
		950 C1 Hr O2 Anneal	
2.17	UV1280	MEASUREMENT STEP	measure oxide thickness and
ļ	1		uniformity
2.18	Outside	CMP of oxide	Sent waters to vendor:
			Entrepix
			Instructions to polish oxide
			and stop on Si
			Use 2 water as pilot waters
2.19	premetal-piranha	post-cmp clean	2x10minutes in 3:1
			H2504:H202

		required to return to	15s din in 50:1 DI H2O:HF
		processing	
2.20		MEASUREMENT STEPS	
a	P10		Measure step height of
1	17/1200		oxide/Si to see overetch
D			Image in SEM for particles
			inage in SEW for particles
3		Emitter and 51 Nanowire I	formation
3.1	RCA-ICL	Standard RCA Clean	
3.2	Tube-5B	Grow 500 ALiner oxide	1000C dry oxidation 1 hr
3.3	DCVD	Deposit Hardmask	250nm DCVD oxide
3.4	Tube-5D	Hardmask densification	1 hour 950 C under O2
3.5	Coater6	Coat front of wafer with photoresist	Hardbake 120C 60s
			Recipe: Protect
3.6	FLX	MEASUREMENT STEP	Measure wafer bow
		Timed etch of backside oxide	Iterate between 3.6 and 3.7
3.7	OxEtch-BOE	to compensate stress from	until wafer bow is 0
		oxide removed during CMP	
3.8	Premetal-piranha	Strip resist	10minutes in 3:1
	F	r	H2SO4:H2O2
			15s dip in 50:1 DI H2O:HF
3.9	Coater6	Coat with I.lum resist	60s 90C prebake
2.10	T		Recipe: THMD5
3.10	Istepper	Photolithography LVL:02	LVL Name: 0.6um dots
			Francisco 160 mg 0 50m
			Exposure: 100 ms 0.9um
			Dynamic puddle developer
3.11	Coater6	Develop	dispense
			120C 30s Postbake
			Recipe: PUDDLE3
3.12		MEASUREMENT STEPS	
			Optically verify vernier marks
a	Microscope		are aligned
h	SEM		Fine alignment and PR dot
U	SEIVI		size
3.13	AME5000	Etch hardmask	Recipe: Baseline-ox-new
			110s with 10s Descum
3.14	SEM	MEASUREMENT STEP	verify oxide disk shape and size
3.15	LAM490B	Rough emitter cone etch	25s Standard Nitride on Si
			Recipe + 10s overetch
3.16	SEM	MEASUREMENT STEP	Measure neck of cone

3.17	STS2	Si Nanowire etch	Etch pillars using recipe GUERRER1 Etch conditions: $6.7s C_4F_8$ passivation cycle @ 45 sccm, 5s SF ₆ etch cycle @ 90 sccm, 600 Watts RF (passivation), 800 Watts RF (etch), 120 Watts bias power, passivate first. Constant pressure 24 mTorr. Total etch time was 8 minutes. End on etch step to minimize polymer contamination.
3.18	Asher-TRL	Remove photoresist and Polymer	Ash in barrel asher minimum 1 hr to remove excess polymer and strip PR
3.19	OxEtch-BOE	Strip oxide hardmask in BOE	
3.20	SEM	MEASUREMENT STEP	Measure Emitter tip cone and pillar diameter and uniformity.
3.21	RCA-TRL	Standard RCA clean prior to oxidation	
3.22	Tube A2	Reduce emitter and pillar diameters using oxidation	Oxidize emitters and nanowires at 950 C using wet oxidation for time based on Silvaco simulations to attain sharp emitters and pillars of the correct diameter. Temperature is limited to 950 C because of viscoelastic flow of oxide at T>950 C.
4	Dielectric Fill-in and Planarization		
4.1	RCA-TRL	Standard RCA Clean	May be omitted if transferring right from diffusion furnace (recommended)
4.2	Tube-B4 Poly	Polysilicon deposition	LPCVD of Polysilicon 800 A at 590 C
4.3	RCA-TRL	Standard RCA clean prior to oxidation	May be omitted if transferring right from PolyTube (recommended)
4.4	Tube-A2	Oxidize polysilicon	40 minutes at 950 C wet oxidation. Enough to oxidize polysilicon + 50% overoxidation to ensure poly is completely consumed

			run 3-5 monitor wafers with	
			characterization	
4.5	UV1280	MEASUREMENT STEP	measure thickness and uniformity of poly-Si	
4.6	RCA-TRL	Standard RCA Clean	May be omitted if transferring right from diffusion furnace (recommended)	
4.7	VTR	Nitride fill in	Deposit 1um LPCVD Si-Rich Nitride Using standard recipe run 3-5 monitor wafers (blank Si) for characterization	
4.8		MEASUREMENT STEPS		
a	UV1280		Measure Nitride thickness and uniformity	
Ь	SEM		image sample ensuring regular pattern and no agglomeration. Optional: cleave and xsem of	
с	AFM		a water image surface roughness of the nitride	
4.9	GnP	CMP Planarization of nitride film	Standard auto recipe polishing rate 4 A/s. Use monitor wafers to characterize polishing rate prior to running real wafers. Don't run for more than 1800s continuously. Enusure wafer is seated well in polishing head prior to starting.	
4.10	premetal-piranha	post-cmp clean	2x10minutes in 3:1 H2SO4:H2O2	
		required to return to processing	15s dip in 50:1 DI H2O:HF	
4.11	UV1280	MEASUREMENT STEP	Measure thickness of nitride and oxide in various spots over wafer	
5	Ga	Gate Definition, Deposition and Patterning		
5.1	NitrEtch-HotPhos	Hot Phosphoric Etch	Etch nitride to expose 200-300 nm of oxide domes. Etch rate 4nm/min	
5.2		MEASUREMENT STEPS		
a	SEM		Image sample to ensure oxide domes are exposed	

with
with
with
with
<u>y</u>
con
·НЕ
,1 11
)er
Der
)er
per
)er
Der
per
per PM,
per PM, le
per PM, le
PM, le inutes
PM, le inutes
per PM, le inutes
1

		· · · · · · · · · · · · · · · · · · ·	×
6.4	Coater6	Image reversal bake	Recipe: PEB95 (95C 90s)
6.5	Istepper	Flood expsosure	500 ms, mask removed
66	Wetbench	Develop Resist	AZ922 Developer until clear
6.6			(45 s)
6.7	Microscope	MEASUREMENT STEP	Verify features
6.8	EbeamFP	Deposit Metal Stack	Ni/Ti/Au 10nm/50nm/100nm
6.9	PhotoWetAu	Liftoff	Overnight in Acetone and
6.9			then sonicate
6.10	Microscope	MEASUREMENT STEP	Verify good liftoff
611	HMDS-TRI	HMDS oven prior to coating	Becipe 5
0.11	TIME/0-TICE	with photoresist	
612	Conter	Spin resist to protect front side	SPR700 1000 RPM 7s
0.12	Coater	during backside film etching	dispense, 30s spin cycle
6.13	Postbake	Hardbake resist	1 hour 120C
614	STS1	Etch backside Poly and	NITRIDE RECIPE. Variable
0.14	5151	Nitride. Watch for clear	time
6.15	LAM590-AU	Etch backside oxide	Standard oxide recipe
			Microstrip 2001 - 30 mins
6.16	PhotoWetAu	Strip resist	70C in waterbath on hot plate
			x2 / SRD
617	Tube B1 Au	-B1 Au Sinter metal	30 mins at 350C under
0.17	Tube-DI Au		forming gas
618	GreenFlo	enFlo Release Emitter tips	Silox Vapox III pad etch - 5
0.10	Gicentio		mins

Bibliography

- [1] SOP for UV1280.
- [2] K. Ajito and Y. Ueno. THz Chemical Imaging for Biological Applications. *IEEE Transactions on Terahertz Science and Technology*, 1(1):293–300, Sept. 2011.
- [3] A. Akinwande, B. Johnson, J. Holmen, D. Murphy, and D. Arch. Thin-film-edge emitter vacuum microelectronics devices for lamp/backlight applications. In *Vacuum Microelec*tronics Conference, 1995. IVMC., 1995 International, pages 418–422, July 1995.
- [4] M. Allan. Measurement of differential cross sections for excitation of helium by electron impact within the first 4 eV above threshold. *Journal of Physics B: Atomic, Molecular and Optical Physics*, 25(7):1559–1575, Apr. 1992.
- [5] R. A. Anderson and J. P. Brainard. Mechanism of pulsed surface flashover involving electron-stimulated desorption. *Journal of Applied Physics*, 51(3):1414, 1980.
- [6] S. Bae, H. Kim, Y. Lee, X. Xu, J.-S. Park, Y. Zheng, J. Balakrishnan, T. Lei, H. Ri Kim, Y. I. Song, Y.-J. Kim, K. S. Kim, B. Özyilmaz, J.-H. Ahn, B. H. Hong, and S. Iijima. Roll-to-roll production of 30-inch graphene films for transparent electrodes. *Nature Nanotechnology*, 5(8):574–578, Aug. 2010.
- [7] J. Baek, M. Shur, K. Lee, and T. Vu. Current #8212;Voltage characteristics of ungated GaAs FET's. *IEEE Transactions on Electron Devices*, 32(11):2426–2430, Nov. 1985.
- [8] M. Basten, J. Tucek, D. Gallagher, K. Kreischer, and R. Mihailovich. A 0.85 THz vacuum-based power amplifier. In *Vacuum Electronics Conference (IVEC)*, 2012 IEEE *Thirteenth International*, pages 39–40, Apr. 2012.
- [9] H. Boll, J. Iwersen, and E. Perry. High-speed current limiters. *IEEE Transactions on Electron Devices*, ED-13(12):904–907, Dec. 1966.
- [10] W. Braun. Applied RHEED: Reflection High-energy Electron Diffraction During Crystal Growth /Wolfgang Braun. Springer Science & Business Media, Jan. 1999.
- [11] I. Brodie and C. A. Spindt. Vacuum Microelectronics. In P. W. Hawkes, editor, Advances in Electronics and Electron Physics, volume 83 of Microelectronics and Microscopy, pages 1–106. Academic Press, 1992.

- [12] J. Browning, N. McGruer, S. Meassick, C. Chan, W. Bintz, and M. Gilmore. Gate field emitter failures: experiment and theory. *IEEE Transactions on Plasma Science*, 20(5):499–506, Oct. 1992.
- [13] J. Browning, N. E. McGruer, W. Bintz, and M. Gilmore. Experimental observations of gated field emitter failures. *IEEE Electron Device Letters*, 13(3):167–169, Mar. 1992.
- [14] J. Browning, C. Watkins, J. Alwan, and J. Hoffman. Scaling of Field Emission Display Technology. Toronto, Canada, 1997.
- [15] J. N. H. Brunt, G. C. King, and F. H. Read. A study of resonance structure in helium using metastable excitation by electron impact with high energy resolution. *Journal of Physics B: Atomic and Molecular Physics*, 10(3):433–448, Feb. 1977.
- [16] S. J. Buckman and B. Lohmann. Low-energy total cross section measurements for electron scattering from helium and argon. *Journal of Physics B: Atomic and Molecular Physics*, 19(16):2547, Aug. 1986.
- [17] J. S. Bunch, S. S. Verbridge, J. S. Alden, A. M. van der Zande, J. M. Parpia, H. G. Craighead, and P. L. McEuen. Impermeable Atomic Membranes from Graphene Sheets. *Nano Letters*, 8(8):2458–2462, Aug. 2008.
- [18] J. Cathey, D.A. Field emission displays. In 1995 International Symposium on VLSI Technology, Systems, and Applications, 1995. Proceedings of Technical Papers, pages 131–136, May 1995.
- [19] P. Chalise, M. Watanabe, A. Okino, E. Hotta, and K. Ko. Study of a low energy pulsed electron beam extraction system. In *Pulsed Power Conference, 2003. Digest of Technical Papers. PPC-2003. 14th IEEE International*, volume 1, pages 275–278 Vol.1, June 2003.
- [20] F. M. Charbonnier, L. A. Southall, and W. A. Mackie. Emission characteristics of NbC/Nb field emitters. *Journal of Vacuum Science & Technology B*, 22(4):1643–1652, July 2004.
- [21] L.-Y. Chen and A. I. Akinwande. Aperture-Collimated Double-Gated Silicon Field Emitter Arrays. *Electron Devices, IEEE Transactions on*, 54(3):601–608, Mar. 2007.
- [22] L.-Y. Chen, L. Velasquez-Garcia, X. Wang, K. Teo, and A. Akinwande. A Microionizer for Portable Mass Spectrometers Using Double-Gated Isolated Vertically Aligned Carbon Nanofiber Arrays. *IEEE Transactions on Electron Devices*, 58(7):2149–2158, July 2011.
- [23] C. D. Child. Discharge From Hot CaO. Physical Review (Series I), 32(5):492–511, May 1911.
- [24] W. Cho, T. Ono, and M. Esashi. Proximity electron lithography using permeable electron windows. *Applied Physics Letters*, 91(4):044104, July 2007.

- [25] B. E. Deal. Standardized Terminology for Oxide Charges Associated with Thermally Oxidized Silicon. *Journal of The Electrochemical Society*, 127(4):979–981, Apr. 1980.
- [26] J. del Alamo. Integrated Microelectronic Devices. In preparation, 2008.
- [27] R. H. Dennard, F. H. Gaensslen, H.-N. Yu, V. L. Rideout, E. Bassous, and A. R. LeBlanc. Design of ion-implanted MOSFET's with very small physical dimensions. *Proceedings of the IEEE*, 87(4):668–678, 1999.
- [28] M. Diarra, C. Delerue, Y.-M. Niquet, and G. Allan. Screening and polaronic effects induced by a metallic gate and a surrounding oxide on donor and acceptor impurities in silicon nanowires. *Journal of Applied Physics*, 103(7):073703, Apr. 2008.
- [29] M. Diarra, Y.-M. Niquet, C. Delerue, and G. Allan. Ionization energy of donor and acceptor impurities in semiconductor nanowires: Importance of dielectric confinement. *Physical Review B*, 75(4):045301, Jan. 2007.
- [30] M. Ding. Field emission from silicon. Thesis, Massachusetts Institute of Technology, 2001.
- [31] M. Ding, H. Kim, and A. Akinwande. Highly uniform and low turn-on voltage Si field emitter arrays fabricated using chemical mechanical polishing. *IEEE Electron Device Letters*, 21(2):66–69, Feb. 2000.
- [32] M. Ding, G. Sha, and A. Akinwande. Silicon field emission arrays with atomically sharp tips: turn-on voltage and the effect of tip radius distribution. *Electron Devices, IEEE Transactions on*, 49(12):2333 2342, Dec. 2002.
- [33] E. S. D. S. a. T. Division. Environmental Issues in the Electronics and Semiconductor Industries: Proceedings of the Third International Symposium. The Electrochemical Society, Jan. 2000.
- [34] T. Doll, J. Vuckovic, M. Hochberg, and A. Scherer. Low-energy electron beam focusing in self-organized porous alumina vacuum windows. *Applied Physics Letters*, 76(24):3635–3637, June 2000.
- [35] V. M. Donnelly and A. Kornblit. Plasma etching: Yesterday, today, and tomorrow. Journal of Vacuum Science & Technology A, 31(5):050825, Sept. 2013.
- [36] Y. Du, H. Cang, W. Wang, F. Han, C. Chen, L. Li, K. Hou, and H. Li. Note: Design and construction of a simple and reliable printed circuit board-substrate Bradbury-Nielsen gate for ion mobility spectrometry. *Review of Scientific Instruments*, 82(8):086103-086103-3, Aug. 2011.
- [37] L. Dvorson, M. Ding, and A. Akinwande. Analytical electrostatic model of silicon conical field emitters. I. *IEEE Transactions on Electron Devices*, 48(1):134–143, Jan. 2001.
- [38] L. Dvorson, I. Kymissis, and A. I. Akinwande. Double-gated silicon field emitters. Papers from the 14th International Vacuum Microelectronics Conference, 21:486–494, 2003.

- [39] E. P. EerNisse. Viscous flow of thermal SiO2. Applied Physics Letters, 30(6):290–293, Mar. 1977.
- [40] J. L. Ellsworth, V. Tang, S. Falabella, B. Naranjo, and S. Putterman. Neutron production using a pyroelectric driven target coupled with a gated field ionization source. In *AIP Conference Proceedings*, volume 1525, pages 128–132. AIP Publishing, Apr. 2013.
- [41] A. Fomani, L. Velasquez-Garcia, and A. Akinwande. Low-Voltage Field Ionization of Gases Up to Torr-Level Pressures Using Massive Arrays of Self-Aligned Gated Nanoscale Tips. *IEEE Transactions on Electron Devices*, 61(5):1520–1528, May 2014.
- [42] R. H. Fowler and L. Nordheim. Electron Emission in Intense Electric Fields. Proceedings of the Royal Society of London. Series A, 119(781):173–181, May 1928.
- [43] E. F. Fullam. A Closed Wet Cell for the Electron Microscope. Review of Scientific Instruments, 43(2):245–247, Feb. 1972.
- [44] M. Ghannam, R. Mertens, and R. De Keersmaecker. Temperature and doping dependence of the fundamental surface recombination velocity. *Physica B+C*, 129(1-3):283-285, Mar. 1985.
- [45] E. Gidcumb, B. Gao, J. Shan, C. Inscoe, J. Lu, and O. Zhou. Carbon nanotube electron field emitters for x-ray imaging of human breast cancer. *Nanotechnology*, 25(24):245704, June 2014.
- [46] R. Gomer. Field Emissions and Field Ionization. American Inst. of Physics, Dec. 1992.
- [47] Y. Gotoh, K. Mukai, Y. Kawamura, H. Tsuji, and J. Ishikawa. Work function of low index crystal facet of tungsten evaluated by the Seppen-Katamuki analysis. *Journal of Vacuum Science & Technology B*, 25(2):508–512, Mar. 2007.
- [48] Y. Gotoh, M. Nagao, D. Nozaki, K. Utsumi, K. Inoue, T. Nakatani, T. Sakashita, K. Betsui, H. Tsuji, and J. Ishikawa. Electron emission properties of Spindt-type platinum field emission cathodes. *Journal of Applied Physics*, 95(3):1537–1549, Feb. 2004.
- [49] Y. Gotoh, H. Tsuji, and J. Ishikawa. Relationship between work function and current fluctuation of field emitters: Use of SK chart for evaluation of work function. *Journal of Vacuum Science & Technology B*, 19(3):992–994, May 2001.
- [50] E. W. Gray. Vacuum surface flashover: A high-pressure phenomenon. Journal of Applied Physics, 58(1):132, 1985.
- [51] H. Gray, G. Campisi, and R. Greene. A vacuum field effect transistor using silicon field emitter arrays. In *Electron Devices Meeting*, 1986 International, volume 32, pages 776–779, 1986.
- [52] E. D. Green and G. S. Kino. Atmospheric scanning electron microscopy using silicon nitride thin film windows. *Journal of Vacuum Science & Technology B*, 9(3):1557–1558, May 1991.

- [53] E. M. Grumstrup, M. M. Gabriel, E. M. Cating, C. W. Pinion, J. D. Christesen, J. R. Kirschbrown, E. L. Vallorz, J. F. Cahoon, and J. M. Papanikolas. Ultrafast Carrier Dynamics in Individual Silicon Nanowires: Characterization of Diameter-Dependent Carrier Lifetime and Surface Recombination with Pump–Probe Microscopy. *The Journal of Physical Chemistry C*, 118(16):8634–8640, Apr. 2014.
- [54] S. Guerrera and A. Akinwande. Self-aligned, gated field emitter arrays with integrated high-aspect-ratio current limiters. In *Vacuum Nanoelectronics Conference (IVNC)*, 2013 26th International, pages 1–2, July 2013.
- [55] S. A. Guerrera, L. F. Velasquez-Garcia, and A. I. Akinwande. Scaling of High-Aspect-Ratio Current Limiters for the Individual Ballasting of Large Arrays of Field Emitters. *Electron Devices, IEEE Transactions on*, 59(9):2524 –2530, Sept. 2012.
- [56] M. Hadsell, G. Cao, J. Zhang, L. Burk, T. Schreiber, E. Schreiber, S. Chang, J. Lu, and O. Zhou. Pilot study for compact microbeam radiation therapy using a carbon nanotube field emission micro-CT scanner. *Medical Physics*, 41(6):061710, June 2014.
- [57] L. Hanlon, M. Greenstein, W. Grossman, and A. Neukermans. Electron window cathode ray tube applications. *Journal of Vacuum Science & Technology B*, 4(1):305–309, Jan. 1986.
- [58] S. S. Hodgman, R. G. Dall, L. J. Byron, K. G. H. Baldwin, S. J. Buckman, and A. G. Truscott. Metastable Helium: A New Determination of the Longest Atomic Excited-State Lifetime. *Physical Review Letters*, 103(5):053002, July 2009.
- [59] B. Hoeneisen and C. A. Mead. Fundamental limitations in microelectronics—I. MOS technology. *Solid-State Electronics*, 15(7):819–829, July 1972.
- [60] C.-y. Hong. Intelligent field emission arrays. Thesis, Massachusetts Institute of Technology, 2003.
- [61] C.-y. Hong and A. Akinwande. Silicon metal #x2013;oxide #x2013;semiconductor field effect transistor/field emission array fabricated using chemical mechanical polishing. Journal of Vacuum Science Technology B: Microelectronics and Nanometer Structures, 21(1):500-505, Jan. 2003.
- [62] C.-Y. Hong and A. Akinwande. Temporal and spatial current stability of smart field emission arrays. *Electron Devices, IEEE Transactions on*, 52(10):2323 2328, Oct. 2005.
- [63] Y. H. Hsien, H. K. Hsu, T. C. Tsai, W. Lin, R. P. Huang, C. H. Chen, C. L. Yang, and J. Y. Wu. Process development of high-k metal gate aluminum CMP at 28 nm technology node. *Microelectronic Engineering*, 92:19–23, Apr. 2012.
- [64] J. C.-M. Huang. Cold cathode field emission display with each microtip having its own ballast resistor, May 1997. U.S. Classification 313/495, 313/336, 313/309, 313/351; International Classification H01J1/304; Cooperative Classification H01J2201/304, H01J2201/30407, H01J2329/00, H01J1/3042, H01J2201/319; European Classification H01J1/304B.

- [65] C. E. Hunt, J. T. Trujillo, and W. Orvis. Structure and electrical characteristics of silicon field emission microelectronic devices. *IEEE Transactions on Electron Devices*, 38(10):2309–2313, Oct. 1991.
- [66] J. Ishikawa, H. Tsuji, Y. Gotoh, T. Sasaki, T. Kaneko, M. Nagao, and K. Inoue. Influence of cathode material on emission characteristics of field emitters for microelectronics devices. *Journal of Vacuum Science & Technology B*, 11(2):403–406, Mar. 1993.
- [67] J. Itoh, T. Hirano, and S. Kanemaru. Ultrastable emission from a metal-oxidesemiconductor fieldeffect transistorstructured Si emitter tip. *Applied Physics Letters*, 69(11):1577-1578, Sept. 1996.
- [68] J. Jackson. Classical electrodynamics, 2nd edition. Wiley, Chichester, Sussex, UK, 1975.
- [69] K. L. Jensen and J. G. Webster. Field Emission Fundamental Theory to Usage. In Wiley Encyclopedia of Electrical and Electronics Engineering. John Wiley & Sons, Inc., 1999.
- [70] K. L. Jensen, E. G. Zaidman, M. A. Kodis, B. Goplen, and D. N. Smithe. Analytical and seminumerical models for gated field emitter arrays. I. Theory. *Journal of Vacuum Science & Technology B*, 14(3):1942–1946, May 1996.
- [71] E. Johnson. Physical limitations on frequency and power parameters of transistors. RCA Review, 26(2):163–177, 1965.
- [72] T. Kamins. Deposition. In *Polycrystalline Silicon for Integrated Circuit Applications*, number 45 in The Kluwer International Series in Engineering and Computer Science, pages 1–40. Springer US, 1988.
- [73] T. I. Kamins, M. M. Mandurah, and K. C. Saraswat. Structure and Stability of Low Pressure Chemically VaporDeposited Silicon Films. *Journal of The Electrochemical Society*, 125(6):927–932, June 1978.
- [74] S. Kanemaru, T. Hirano, H. Tanoue, and J. Itoh. Control of emission currents from silicon field emitter arrays using a built-in MOSFET. *Applied Surface Science*, 111:218–223, Feb. 1997.
- [75] H. Kang, K. Kim, Y. Shin, I. Park, K. Ko, C. Kim, K. Oh, S. Kim, C. Hong, K. Kwon, J. Yoo, Y. Kim, C. Lee, W. Paick, D. Suh, C. Park, S. Lee, S. Ahn, C. Hwang, and M. Lee. Highly manufacturable process technology for reliable 256 Mbit and 1 Gbit DRAMs. In *Electron Devices Meeting*, 1994. IEDM '94. Technical Digest., International, pages 635–638, Dec. 1994.
- [76] R. W. Keyes. The effect of randomness in the distribution of impurity atoms on FET thresholds. *Applied physics*, 8(3):251–259, Nov. 1975.
- [77] J. Kraus, R. Reichelt, S. Gunther, L. Gregoratti, M. Amati, M. Kiskinova, A. Yulaev, I. Vlassiouk, and A. Kolmakov. Ambient-Pressure X-ray Photoelectron Spectroscopy through Electron Transparent Graphene Membranes. arXiv:1405.3611 [condmat, physics:physics], May 2014. arXiv: 1405.3611.

- [78] A. Kumar and C. Huei. Synthesis and Biomedical Applications of Graphene: Present and Future Trends. In M. Aliofkhazraei, editor, *Advances in Graphene Science*. InTech, July 2013.
- [79] A. Kyritsakis, J. P. Xanthakis, and D. Pescia. Scaling properties of a non-Fowler-Nordheim tunnelling junction. *Proceedings of the Royal Society A: Mathematical, Physical and Engineering Sciences*, 470(2166):20130795–20130795, Mar. 2014.
- [80] C. Lee, X. Wei, J. W. Kysar, and J. Hone. Measurement of the Elastic Properties and Intrinsic Strength of Monolayer Graphene. *Science*, 321(5887):385–388, July 2008.
- [81] P. Lenard. Ueber Kathodenstrahlen in Gasen von atmosphärischem Druck und im äussersten Vacuum. *Annalen der Physik*, 287(2):225–267, Jan. 1894.
- [82] C. Li, M. T. Cole, W. Lei, K. Qu, K. Ying, Y. Zhang, A. R. Robertson, J. H. Warner, S. Ding, X. Zhang, B. Wang, and W. I. Milne. Highly Electron Transparent Graphene for Field Emission Triode Gates. *Advanced Functional Materials*, 24(9):1218–1227, Mar. 2014.
- [83] M. Litchy, D. Grant, and G. Van Schooneveld. Effects of Fluid Handling Components on Slurry Health. In *Proceedings of PacRim CMP 2005*, 2005.
- [84] J.-N. Longchamp, T. Latychevskaia, C. Escher, and H.-W. Fink. Low-energy electron transmission imaging of clusters on free-standing graphene. *Applied Physics Letters*, 101(11):113117, Sept. 2012.
- [85] M. Lundstrom. Fundamentals of Carrier Transport. Cambridge University Press, July 2009.
- [86] R. B. Marcus, T. S. Ravi, T. Gmitter, K. Chin, D. Liu, W. J. Orvis, D. R. Ciarlo, C. E. Hunt, and J. Trujillo. Formation of silicon tips with <1 nm radius. *Applied Physics Letters*, 56(3):236–238, Jan. 1990.
- [87] J. McPherson, J. Kim, A. Shanware, H. Mogul, and J. Rodriguez. Trends in the ultimate breakdown strength of high dielectric-constant materials. *IEEE Transactions on Electron Devices*, 50(8):1771–1778, Aug. 2003.
- [88] W. I. Milne, K. B. K. Teo, G. A. J. Amaratunga, P. Legagneux, L. Gangloff, J.-P. Schnell, V. Semet, V. T. Binh, and O. Groening. Carbon nanotubes as field emission sources. *Journal of Materials Chemistry*, 14(6):933–943, Mar. 2004.
- [89] U. K. Mishra, L. Shen, T. Kazior, and Y.-F. Wu. GaN-Based RF Power Devices and Amplifiers. *Proceedings of the IEEE*, 96(2):287–305, Feb. 2008.
- [90] G. Moore. Cramming More Components Onto Integrated Circuits. *Proceedings of the IEEE*, 86(1):82–85, Jan. 1998.
- [91] J. Y. Mutus, L. Livadaru, J. T. Robinson, R. Urban, M. H. Salomons, M. Cloutier, and R. A. Wolkow. Low-energy electron point projection microscopy of suspended graphene, the ultimate 'microscope slide'. *New Journal of Physics*, 13(6):063011, June 2011.

- [92] M. Nagao, T. Yoshida, S. Kanemaru, Y. Neo, and H. Mimura. Field emitter array with a built-in multi-electrode lens. In *Vacuum Nanoelectronics Conference, 2009. IVNC 2009.* 22nd International, pages 39 –40, July 2009.
- [93] H. S. Nalwa. Handbook of Surfaces and Interfaces of Materials: Surface and interface phenomena, volume 1. Academic Press, 2001.
- [94] M. Nandakumar, A. Chatterjee, S. Sridhar, K. Joyner, M. Rodder, and I.-C. Chen. Shallow trench isolation for advanced ULSI CMOS technologies. In *Electron Devices Meeting*, 1998. IEDM '98. Technical Digest., International, pages 133–136, Dec. 1998.
- [95] D. Nicolaescu, M. Nagao, T. Sato, V. Filip, S. Kanemaru, and J. Itoh. Emission statistics for Si and HfC emitter arrays after residual gas exposure. *Journal of Vacuum Science & Technology B*, 23(2):707-717, Mar. 2005.
- [96] L. Nilsson, O. Groening, C. Emmenegger, O. Kuettel, E. Schaller, L. Schlapbach, H. Kind, J.-M. Bonard, and K. Kern. Scanning field emission from patterned carbon nanotube films. *Applied Physics Letters*, 76(15):2071–2073, Apr. 2000.
- [97] L. Nilsson, O. Groening, P. Groening, O. Kuettel, and L. Schlapbach. Characterization of thin film electron emitters by scanning anode field emission microscopy. *Journal of Applied Physics*, 90(2):768–780, July 2001.
- [98] L. W. Nordheim. The Effect of the Image Force on the Emission and Reflexion of Electrons by Metals. *Proceedings of the Royal Society of London. Series A*, 121(788):626–639, Dec. 1928.
- [99] D. Palmer, H. F. Gray, J. Mancusi, D. Temple, C. Ball, J. L. Shaw, and G. E. McGuire. Silicon field emitter arrays with low capacitance and improved transconductance for microwave amplifier applications. *Journal of Vacuum Science & Technology B*, 13(2):576–579, Mar. 1995.
- [100] A. A. Patterson and A. I. Akinwande. Elementary framework for cold field emission: Incorporation of quantum-confinement effects. *Journal of Applied Physics*, 114(23):234303, Dec. 2013.
- [101] A. Persaud. Analysis of slope-intercept plots for arrays of electron field emitters. *Journal of Applied Physics*, 114(15):154301, Oct. 2013.
- [102] D. Pflug, M. Schattenburg, H. Smith, and A. Akinwande. Field emitter arrays for low voltage applications with sub 100 nm apertures and 200 nm period. In *Electron Devices Meeting, 2001. IEDM '01. Technical Digest. International*, pages 8.5.1–8.5.4, Dec. 2001.
- [103] D. G. Pflug. Low voltage field emitter arrays through aperture scaling. Thesis, Massachusetts Institute of Technology, 2000.
- [104] E. H. Poindexter. MOS interface states: overview and physicochemical perspective. Semiconductor Science and Technology, 4(12):961–969, Dec. 1989.

- [105] X.-Z. Qin, W.-L. Wang, N.-S. Xu, Z.-B. Li, and R. G. Forbes. Analytical treatment of cold field electron emission from a nanowall emitter, including quantum confinement effects. *Proceedings of the Royal Society of London A: Mathematical, Physical and Engineering Sciences*, 467(2128):1029–1051, Apr. 2011.
- [106] A. Ranadive. Low Power Row and Column Drivers for Field Emission Displays. PhD thesis, Massachusetts Institute of Technology, May 1998.
- [107] R. Raue, A. T. Vink, and T. Welker. Phosphor screens in cathode-ray tubes for projection television. *Philips technical review*, 44(11-12):335–347, 1989.
- [108] R. R. Razouk and B. E. Deal. Dependence of Interface State Density on Silicon Thermal Oxidation Process Variables. *Journal of The Electrochemical Society*, 126(9):1573–1581, Sept. 1979.
- [109] F. Roozeboom, F. v. d. Bruele, Y. Creyghton, P. Poodt, and W. M. M. Kessels. Cyclic Etch/Passivation-Deposition as an All-Spatial Concept toward High-Rate Room Temperature Atomic Layer Etching. ECS Journal of Solid State Science and Technology, 4(6):N5067–N5076, Jan. 2015.
- [110] E. Rutherford. The scattering of and particles by matter and the structure of the atom. *Philosophical Magazine Series 6*, 21(125):669–688, May 1911.
- [111] V. Schmidt, S. Senz, and U. Gösele. Influence of the Si/SiO2 interface on the charge carrier density of Si nanowires. *Applied Physics A*, 86(2):187–191, Nov. 2006.
- [112] N. Shiono and C. Hashimoto. Threshold-voltage instability of n-channel MOSFET's under bias-temperature aging. *IEEE Transactions on Electron Devices*, 29(3):361–368, Mar. 1982.
- [113] B. S. Simpkins, M. A. Mastro, C. R. E. Jr, and P. E. Pehrsson. Surface depletion effects in semiconducting nanowires. *Journal of Applied Physics*, 103(10):104313, May 2008.
- [114] I. Solano, B. Reichenbach, P. R. Schwoebel, D. L. Chichester, C. E. Holland, K. L. Hertz, and J. P. Brainard. Field desorption ion source development for neutron generators. Nuclear Instruments and Methods in Physics Research Section A: Accelerators, Spectrometers, Detectors and Associated Equipment, 587(1):76–81, Mar. 2008.
- [115] C. Spindt, C. Holland, and P. Schwoebel. 11.1: A reliable improved Spindt cathode design for high currents. In *Vacuum Electronics Conference (IVEC)*, 2010 IEEE International, pages 201–202, May 2010.
- [116] C. A. Spindt, I. Brodie, L. Humphrey, and E. R. Westerberg. Physical properties of thinfilm field emission cathodes with molybdenum cones. *Journal of Applied Physics*, 47(12):5248–5263, Dec. 1976.
- [117] C. A. Spindt, C. E. Holland, A. Rosengreen, and I. Brodie. Fieldemitterarray development for highfrequency operation. *Journal of Vacuum Science & Technology B*, 11(2):468–473, Mar. 1993.

- [118] R. Stratton. Diffusion of Hot and Cold Electrons in Semiconductor Barriers. *Physical Review*, 126(6):2002–2014, June 1962.
- [119] S. M. Sze and K. K. Ng. Physics of Semiconductor Devices. John Wiley & Sons, Oct. 2006.
- [120] H. Takemura, Y. Tomihari, N. Furutake, F. Matsuno, M. Yoshiki, N. Takada, A. Okamoto, and S. Miyano. A novel vertical current limiter fabricated with a deep trench forming technology for highly reliable field emitter arrays. In *Electron Devices Meeting*, 1997. IEDM '97. Technical Digest., International, pages 709–712, Dec. 1997.
- [121] D. Temple, W. D. Palmer, L. N. Yadon, J. E. Mancusi, D. Vellenga, and G. E. McGuire. Silicon field emitter cathodes: Fabrication, performance, and applications. *Journal of Vacuum Science & Technology A*, 16(3):1980–1990, May 1998.
- [122] T. Utsumi. Vacuum microelectronics: what's new and exciting. IEEE Transactions on Electron Devices, 38(10):2276–2283, Oct. 1991.
- [123] G. N. A. van Veen. Space-charge effects in Spindt-type field emission cathodes. Journal of Vacuum Science & Technology B, 12(2):655-661, Mar. 1994.
- [124] L. F. Velasquez-Garcia, S. A. Guerrera, Y. Niu, and A. I. Akinwande. Uniform High-Current Cathodes Using Massive Arrays of Si Field Emitters Individually Controlled by Vertical Si Ungated FETs - Part 1: Device Design and Simulation. *IEEE Transactions on Electron Devices*, 58(6):1775–1782, June 2011.
- [125] L. F. Velasquez-Garcia, S. A. Guerrera, Y. Niu, and A. I. Akinwande. Uniform High-Current Cathodes Using Massive Arrays of Si Field Emitters Individually Controlled by Vertical Si Ungated FETs - Part 2: Device Fabrication and Characterization. *IEEE Transactions on Electron Devices*, 58(6):1783–1791, June 2011.
- [126] D. Whaley, R. Duggal, C. Armstrong, C. Bellew, C. Holland, and C. Spindt. Operation of a low-voltage high-transconductance field emitter array TWT. In *Vacuum Electronics Conference, 2008. IVEC 2008. IEEE International*, pages 78–79, Apr. 2008.
- [127] J. D. Wood, G. P. Doidge, E. A. Carrion, J. C. Koepke, J. A. Kaitz, I. Datye, A. Behnam, J. Hewaparakrama, B. Aruin, Y. Chen, H. Dong, R. T. Haasch, J. W. Lyding, and E. Pop. Annealing free, clean graphene transfer using alternative polymer scaffolds. *Nanotechnology*, 26(5):055302, Feb. 2015.
- [128] P. Wrschka, J. Hernandez, G. S. Oehrlein, and J. King. Chemical mechanical planarization of copper damascene structures. *Journal of the electrochemical society*, 147(2):706–712, 2000.
- [129] J. Yeom, Y. Wu, J. C. Selby, and M. A. Shannon. Maximum achievable aspect ratio in deep reactive ion etching of silicon due to aspect ratio dependent transport and the microloading effect. *Journal of Vacuum Science & Technology B*, 23(6):2319–2329, Nov. 2005.

[130] N. Zouache and A. Lefort. Electrical breakdown of small gaps in vacuum. *IEEE Transactions on Dielectrics and Electrical Insulation*, 4(4):358–364, Aug. 1997.